UNIVERSITÉ CLERMONT AUVERGNE ÉCOLE DOCTORALE SCIENCES POUR L'INGÉNIEUR DE CLERMONT-FERRAND

THÈSE

Présentée par

ZHENG ZE ZHU

Master Génie Mécanique

pour obtenir le grade de

Docteur d'Université

Spécialité : Doctorat Electronique et systèmes

Hierarchical and Hybrid Decisional Control Architecture for Cooperative Navigation of Autonomous Vehicles in Complex Environments/Situations

Soutenue publiquement le 13 Juillet 2022 devant le Jury composé de :

	Rapporteur Rapporteur	Professeur à l'INSA de Lyon
JAKOB F UCHINGEN	napporteur	FIDIESSEUL à UTIVEISILE Faits Saciay
HAIYING ZHOU	Examinateur	Professeur, Dongfeng Motor Corporation Technical Center,
		Associé à Hubei University of Automotive Technology (Chine)
REINE TALJ KFOURY	Examinatrice	Chargée de recherche CNRS, HDR à Heudiasyc/UTC
LEÏLA KLOUL	Examinatrice	Maître de Conférences, HDR à Université de Versailles Saint-
		Quentin-en-Yvelines
YOUCEF MEZOUAR	Examinateur	Professeur, SIGMA Clermont
LOUNIS ADOUANE	Directeur de thèse	Professeur, Université de Technologie de Compiègne,
		Associé à l'Institut Pascal/UCA
ALAIN QUILLIOT	Directeur de thèse	Professeur, Université Clermont Auvergne

ACKNOWLEDGEMENTS

ABSTRACT

Cooperative navigation (CN) is a frequently used technique for ensuring the effective navigation of Intelligent Vehicles (IVs). In this background, efforts to establish a Connected Autonomous Vehicles (CAVs)-based traffic control system via vehicular communications technology have accelerated in recent decades. Safe and flexible multivehicle coordination (MVC) technology, in particular, has attracted considerable interest for its ability to deal with complicated environments/situations. Additionally, a feasible hierarchical architecture is critical for cooperative driving with numerous control goals in autonomous vehicles. Thus, the objective of this PhD thesis is to develop reliable MVC technology (e.g., trajectory planning and decision-making for motion planning) and CAVs-based frameworks for use in complex environments/situations. To achieve this goal, this thesis first presented a safe and flexible cooperative navigation technique with risk assessment for cramped local locations (defined as single intersection/roundabout). The *ɛ*-constraint Probability Collectives (PC) algorithm, which is based on the distributed Collective Intelligence (CI) theory, is developed to offer proper solutions for cooperative driving. More precisely, IVs can compute their optimal/sub-optimal and risk-sensitive (i.e., invasive or conservative) cooperative navigation strategies base on the decentralized ε -PC framework, enabling collision-free trajectories in the decision-making level. Next, it is suggested a global supervisor responsible for scheduling and improving vehicle navigation routes while also proposing well-suited trade-offs between speed and risk to achieve the targeted tasks. To better deal with the inhere complexity of CN system in a transportation network (e.g., intersection/roundabout and the expended intersection network), the second part of the thesis addresses the potentialities of adopting Multilayer Hybrid Control Policy and Motion Planning (MHCP-MP) framework. Given the fluctuating road traffic, it was recommended that local supervisors be in control of the urban network's intersections (tricky regions). Specifically, the local supervisor works as a mediator between the global traffic management level and the CAVs decision level, sending instructions to regulate vehicles' trajectories and improve the mobility and safety of the overall transportation system. To accomplish the aim, a Macroscopic Fundamental Diagram (MFD)-based approach in the proposed MHCP-MP framework is designed with concise urban traffic data (e.g., vehicle position, speed, etc.). Further, the Micro-Macro Flow Control (MiMaFC) strategy is proposed to demonstrate the advantages of establishing a link between the suggested local collective optimization framework and macro traffic model for improving the fluidity of the overall transportation system. Following that, the suggested intersection navigation protocols in a deep relationship with our established intelligent intersection management system are designed to permit an uncertain traffic flow. Finally, the proposed CN management architecture in this thesis has been proven in a dedicated transportation network through intensive simulation.

Keywords: Cooperative navigation, Multi-vehicle coordination, Hierarchical architecture, Risk assessment, Probability collectives, Traffic management.

Résumé

La Navigation Coopérative (NC) est une technique fréquemment utilisée pour assurer la navigation efficace des Véhicules Intelligents (VI). Dans ce contexte, les efforts visant à établir un système de contrôle du trafic basé sur les véhicules autonomes connectés (CAVs) par le biais des technologies de communication entre véhicules se sont accélérés au cours des dernières décennies. La technologie de coordination multi-véhicules (MVC) sûre et flexible a suscité, en particulier, un intérêt considérable grâce à sa capacité à gérer des environnements/situations complexes. En outre, une architecture hiérarchique faisable est essentielle pour la conduite coopérative avec de nombreux objectifs de contrôle des véhicules autonomes. Ainsi, l'objectif de cette thèse est de développer une technologie MVC fiable (e.g., la prise de décision pour la planification) et des cadres basés sur les CAVs pour une utilisation dans des environnements/situations complexes. Pour atteindre cet objectif, cette thèse présente, tout d'abord, une technique de NC sûre et flexible avec une évaluation des risques pour les emplacements locaux encombrés (définis comme une seule intersection et/ou un rond-point). L'algorithme ε-Probabilté Collective (PC) à contrainte, qui est basé sur la théorie de l'Intelligence Collective (IC) distribuée, est développé pour offrir des solutions appropriées pour la conduite coopérative. Plus précisément, les VI peuvent calculer leurs stratégies de navigation coopérative optimales/sous-optimales et sensibles au risque (invasives ou conservatrices) en se basant sur le cadre décentralisé *e*-PC, ce qui garantit des trajectoires sans collision. Ensuite, nous suggérons d'utiliser un superviseur global responsable de l'ordonnancement et de l'amélioration des trajectoires de navigation des véhicules, tout en proposant des compromis adaptés entre la vitesse et le risque de la réalisation des tâches visées. Afin de mieux gérer la complexité inhérente aux systèmes NC dans un réseau de transport (e.g., les intersections/ronds-points et le réseau étendu d'intersections), la deuxième partie de la thèse aborde le potentiel de l'adoption d'une architecture de contrôle hybride multicouches et de planification du mouvement (CHM-PM). Compte tenu de la fluctuation du trafic routier, il a été recommandé que des superviseurs locaux contrôlent les intersections du réseau urbain (régions dangereuses). Plus précisément, le superviseur local joue le rôle intermédiaire entre le niveau de gestion global du trafic et le niveau de décision des CAVs, en envoyant des instructions pour réguler les trajectoires des véhicules et améliorer la mobilité et la sécurité du système de transport globale. Pour atteindre cet objectif, une approche basée sur le Diagramme Fondamental Macroscopique (DFM) dans l'architecture CHM-PM proposée est conçue avec des données de trafic urbain concises (par exemple, la position du véhicule, la vitesse, etc.). En outre, la stratégie de contrôle des flux micro-macro (MiMaFC) est proposée pour démontrer les avantages de l'établissement d'un lien entre l'architecture d'optimisation collective locale proposée et le macro modèle de trafic pour améliorer la fluidité du système de transport global. Ensuite, les protocoles suggérés de navigation aux intersections, en forte relation avec notre système de gestion intelligente des intersections, sont conçus pour permettre un flux de trafic incertain. Enfin, l'architecture de gestion des NC proposée dans cette thèse a été évaluée dans un réseau de transport par un travail de simulation intensive.

Mots-clés : Navigation coopérative, Coordination multi-véhicules, Architecture hiérarchique, Évaluation des risques, Probabilté Collective, Gestion du trafic.

CONTENTS

Ge	enera	l introd	luction		9
I	Con	ntext and state of the art			17
1	Coo	perativ	ve motior	planning for multi-vehicle navigation	19
	1.1	Introdu	uction		19
		1.1.1	From sir	gle-vehicle intelligence to multi-vehicle collaboration	21
		1.1.2	Overview	w of multi-vehicle control architectures	28
			1.1.2.1	Centralized architectures	28
			1.1.2.2	Decentralized architectures	29
			1.1.2.3	Other architectures	29
		1.1.3	Multi-vel	nicle navigation in complex environment/situations	30
			1.1.3.1	Highway entrance/exit ramps	31
			1.1.3.2	Intersection/roundabout	32
	1.2	Intellig	ent vehic	le's motion planning at an intersection	34
		1.2.1	An overview of the intersection management system		
		1.2.2	2.2 Driving strategies at an intersection		
			1.2.2.1	Reactive strategies	38
			1.2.2.2	Cognitive strategies	40
		1.2.3	Main cha	allenges under urban conditions	41
			1.2.3.1	Flexible and reliable navigation under uncertainty	41
			1.2.3.2	Parallel implementation (scalability)	42
	1.3	Conclu	usion		42
2	Dec	ision-n	naking fo	r multi-vehicle navigation in a transportation network	45
	2.1	Introdu	uction		45
		2.1.1	Overview	w of a road network: scenarios and requirements	46
		2.1.2	Multi-vel	nicle navigation management systems/behaviors	49
	2.2	Dynar	Dynamic multi-vehicle navigation in intersection networks		

		2.2.1	Main res	search paradigms and implementation	55
		2.2.2	Decisior	n-making modeling	59
			2.2.2.1	Decision-making: microscopic modeling and control	60
			2.2.2.2	Decision-making: macroscopic strategy	62
			2.2.2.3	Decision-making: hierarchical control	64
		2.2.3	Probabil	istic approach for decision-making	66
		2.2.4	Risk ma	nagement for multi-vehicle navigation	67
	2.3	Conclu	usion		68
II	Co	operat	ive navi	gation scheme in complex environment/situations	69
3	Safe	e and fl	exible co	operative navigation with risk assessment	71
	3.1	Proble	em statem	ent	72
		3.1.1	Related	work and main challenges	73
		3.1.2	Probabil	ity Collectives (PC) algorithm	73
			3.1.2.1	Formulation of searching space	74
			3.1.2.2	Two steps for re-acceleration	75
			3.1.2.3	Objective function	75
			3.1.2.4	A summary of PC model	76
	3.2	The ε -	PC for sa	fe and flexible CN	76
		3.2.1	MVS for	mulation	76
			3.2.1.1	An overview of MVS framework	77
			3.2.1.2	Time-Slot-Based (TSB) mechanism	78
			3.2.1.3	Motion planning for continuous vehicles crossing	80
		3.2.2	Risk me	asurement by 2D TTC	80
		3.2.3	Risk ma	nagement based on $arepsilon$ -PC framework \ldots \ldots \ldots \ldots	82
			3.2.3.1	A selection of ε value \ldots \ldots \ldots \ldots \ldots \ldots	83
			3.2.3.2	arepsilon-PC framework	84
		3.2.4	Simulati	on results	87
			3.2.4.1	Communication characterization for the expected data ex- change between vehicles	87
			3.2.4.2	Parameter setting and results evaluation	88
			3.2.4.3	Scalability properties	92
			3.2.4.4	Experiments using continuous vehicle flow	94
	3.3	Beyon	d local co	operation: speed versus risk from a global perspective	95

		3.3.1	Definitions and general context	95
		3.3.2	Global transit network and SPR model	97
		3.3.3	Local search algorithm for SPR	100
		3.3.4	Comparison experiments	104
	3.4	Concl	usion	106
4	84:84		need bieversbied traffic menowers eventite ture in two and	
4	tion	netwo	rks	109
	4.1	Backg	round and challenges	109
	4.2	MiMal	-C-based Hierarchical architecture	111
		4.2.1	Supervised traffic management scheme.	112
		4.2.2	Hierarchical and hybrid decision-making architecture	112
	4.3	MiMal	-C strategy in Traffic flow model	113
		4.3.1	Formulation of vehicle driving model	114
		4.3.2	Formulation of traffic flow fundamentals	114
		4.3.3	Formulation of MiMaFC strategy	117
	4.4	MiMal	FC strategy for CN	118
		4.4.1	Proposed intersection navigation protocol	118
			4.4.1.1 Preparation area: vehicle sorting	121
			4.4.1.2 Decision area: selecting strategy	121
			4.4.1.3 Implementation area: conflicts processing	122
		4.4.2	Developing and executing CN strategy	123
			4.4.2.1 Velocity-planning-based strategy	123
			4.4.2.2 Spatio-temporal velocity adaptation approach	126
			4.4.2.3 MiMaFC-based CN strategy	127
	4.5	Simula	ation results	128
	4.6	Concl	usion	133
6	nora		usion and future work	195
Ge	enera			135
	_			
	An	nexes		141
A	Stru	icture i	results for Shortest Path under Risk constraint (SPR) problem	143
В	The	stocha	astic space policy in macroscopic flow model	145
Bi	Bibliography 147			

LIST OF FIGURES

1	Multi-vehicle hierarchical and hybrid decisional control architecture	13
2	The outline of the PhD manuscript	14
1.1	2021 Gartner hyper cycle for Artificial Intelligence (AI) [151]: Autonomous Vehicles at the "rock bottom" before entering the plateau of productivity	20
1.2	Operational Design Domain (ODD) relative to driving automation [98]	20
1.3	A brief view of autonomous vehicle development in the last two decades	23
1.4	Examples of common multi-robot systems.	25
1.5	Application of Connected Autonomous/Automated Vehicles	26
1.6	General architecture of a MVS with communication, perception, localiza- tion, planning and control modules [481].	28
1.7	A classification of MVS coordination scenarios from both rural and urban areas (cite from https://www.freepik.com/)	31
1.8	Merging/splitting maneuvers at highway entrance/exit ramps (designed on https://icograms.com/)	32
1.9	Signalized and unsignalized intersection/roundabout (designed on https://icograms.com/)	33
1.10	General intersection management with MVS	35
1.11	The intersection structure with different topologies [485]	36
1.12	Relationship between reactive and cognitive control [8]	39
2.1	Urban arterial corridors and road networks in the literature.	47
2.2	Traditional traffic direct/indirect control	50
2.3	An evolution of Multi-Vehicle System (MVS)-based UTC systems (moti- vated by [266])	50
2.4	Numerous cooperative behaviors by MVS-based UTC system	52
2.5	An illustration of traffic oscillations [545]: between 5:00 and 5:15 p.m. (left image), vehicles experience greater oscillations than between 4:00 and 4:15 p.m. (right image).	53
2.6	The spatial and temporal dependence in traffic forecasting: a strong influence between adjacent roads in (a); the traffic volume changes periodically within one week or one day in (b) [538].	54
2.7	A realistic environment for deploying MVS urban navigation technology	59

2.8	A bird's view of the proposed agent-based traffic management scheme in a bottom-up and hierarchical system.	60
2.9	An illustration of microscopic models	61
2.10	A comparison of various decision-making strategies in relation to the MFD/NFD curves.	63
2.11	In the synchronization-based control model, any vehicle from <i>i</i> to <i>j</i> will pass through (1) the uncontrolled zone between <i>i</i> and <i>j</i> , (2) the adaption phase control zone inside <i>j</i> 's radius of <i>R</i> , and (3) the crossing zone of radius r_0 [446].	65
3.1	IVs system with action probability distribution to predict their behaviors	72
3.2	Example of strategies hypotheses for vehicle actions.	74
3.3	Application scenario and main zones characterizing the addressed MVS	77
3.4	Time-Slot-Based (TSB) communication mechanism for applying PC	79
3.5	A collision of two-vehicles based on circle area.	81
3.6	Flowchart of the proposed ε -PC algorithm: at the initial state, for vehicle i in turn to run the ε -PC algorithm; a combined strategy Y_k^i including its own strategy Π_k^i and other selected strategies is given after each vehicle's iteration. Additionally, MVS keeps track of current favorable actions Y_{cur}^* and compares them to earlier ones. Finally, as the solution converges, the final optimal solution Y_{opt}^* is chosen.	85
3.7	Three vehicles navigation by original PC (simulation video https://bit.ly/ 3kkRJ9v)	90
3.8	Three vehicles navigation by ε -PC (simulation video https://bit.ly/3KmRgxZ).	91
3.9	Effect of strategy number on the execution time.	92
3.10	Effect of vehicle number on the performance of ϵ -PC	93
3.11	MVS navigation in real-time (simulation video https://bit.ly/3KojvfO)	94
3.12	A warehouse like transit network: at time t , both green and black vehicles are scheduled in a risky area within an arc $e(A, B)$ (inspired from [38])	97
3.13	A piecewise function $\Pi^{e}(t)$	98
3.14	Risk versus time for every instance.	106
3.15	Reference CPU computing time.	106
4.1	Urban road network for autonomous vehicle Cooperative Navigation (CN) with local supervisor S_{LocA} at each intersection region.	111
4.2	Basic schematic of the proposed Multi-layer Hybrid Control Policy and Mo- tion Planning (MHCP-MP) framework.	113

4.3	CAVs navigation in transportation network with Multiple unsignalized in- tersections: the origins (destinations) are noted by $O_i(D_i)$. Each 4-way intersection is in charged by a local supervisor S_{LOAi} . The 3-tuple of traf- fic density, velocity and flow rate are defined for each link { K_{L_i} , V_{L_i} , Q_{L_i} }, destination lane { K_{D_i} , V_{D_i} , Q_{D_i} } and intersection area { K_{S_i} , V_{S_i} , Q_{S_i} }	115
4.4	The Macroscopic Fundamental Diagram (MFD) of theoretical correlation between traffic density and traffic flow.	117
4.5	Individual vehicles approach the intersection and make decision from IDMP P_0 to participate in CN. The interactive area supervised by S_{LoA} is divided into three parts (i.e., MiMaFC-DMA, IND-DMA and CA identified by different colors), each of which implements different corresponding navigation protocol for cooperative strategy.	119
4.6	Flow-chart showing the proposed cooperative strategy for traffic network navigation diagram.	120
4.7	An illustration of the possible CAVs trajectories with sampled self- interested speed profiles.	124
4.8	A spatio-temporal velocity adaptation approach for CAVs' developing speed strategy at the supervised intersection.	125
4.9	Unidirectional flow of CAVs navigation in traffic network for both supervised and unsupervised system. The red circles stand for the coved range of a local supervisor S_{LoA} noted from 1 to 9 (simulation videos can be found at https://bit.ly/3zcDwAu)	130
4.10	A Comparison of CAVs velocities between MHCP-MP framework with S_{Loc_A} and baseline method without S_{Loc_A} . An augmented navigation protocol was also implemented in the MHCP-MP framework.	131
4.11	A comparison of inter-vehicle distance in traffic network	131
4.12	Traffic flow-density diagrams for intersections and output flow in whole exits.	132
4.13	The traffic displacement diagram for the whole vehicles	132
4.14	Past and potential evolution towards Cooperative Navigation (CN) [49]	138
B.1	A comparison of different time headway $t\hat{h}_i$ distribution.	146
B.2	The time headway distribution $t\hat{h}_i$ at speeds of $[0m/s, 5m/s]$, $[5m/s, 10m/s]$, $[10m/s, 15m/s]$ and $[15m/s, +\infty m/s]$.	146

LIST OF TABLES

2.1	Summary of decision-making model for MVS navigation in road networks .	66
3.1	Data exchange in "single" mode and "full" mode for three vehicles	88
3.2	Parameters and initial states	89
3.3	Performance comparison of different ε_i	92
3.4	Characteristics of the instances	105
4.1	Parameters adopted in the tackled scenario	129
4.2	Average flow in local area	133
B.1	A test of the log-normal distribution's parameters	145

GLOSSARY

- AGVs: Autonomous Guided Vehicles.
- AHS: Automated Highway Systems.
- AI: Artificial Intelligence.
- AIM: Autonomous Intersection Management.
- CACC: Cooperative Adaptive Cruise Control.
- CAVs: Connected Autonomous/Automated Vehicles.
- CIC: Cooperative Intersection Control.
- CN: Cooperative Navigation.
- CVs: Connected Vehicles.
- FIFO: First In First Out.
- **121:** Infrastructure-to-infrastructure.
- I2V: Infrastructure-to-vehicle.
- ITS: Intelligent Transportation Systems.
- IVs: Intelligent Vehicles.
- MFD: Macroscopic Fundamental Diagram.
- MRS: Multi-Robot System.
- MVC: Multi-Vehicle Coordination.
- MVN: Multi-Vehicle Navigation.
- MVS: Multi-Vehicle System.
- NFD: Network Fundamental Diagram.
- PAVIN: Plate-forme d'Auvergne pour Véhicules INtelligents.
- PC: Probability Collectives.
- RL: Reinforcement learning.
- SPaT: Signal Phase and Time.

- SPR: Shortest Path under Risk constraint.
- **TTC:** Time-To-Collision.
- UGV: Unmanned Ground Vehicle.
- UTC: Urban Traffic Control.
- UVS: Unmanned Vehicle System.
- V2I: Vehicle-to-infrastructure.
- V2V: Vehicle-to-vehicle.
- V2X: Vehicle-to-everything.
- VIPALAB: Véhicule Individuel Public et Autonome.

GENERAL INTRODUCTION

CONTEXT OF THE PHD THESIS

Over the last few years, the development of fully autonomous vehicles for transportation tasks has received even more attention from different laboratories/companies throughout the world [8], [71]. The focus of the proposed PhD subject is passengers' transportation in midtown or in closed/dedicated areas like inside big companies, amusement parks, airports, etc., which need autonomous shuttles between their different areas. Within the context of a complex task in such a constrained region, multi-vehicle navigation and coordination need very precise design and management of the vehicle interaction [8]. The applicative focus of the proposed PhD thesis corresponds to the field of autonomous public transportation. Nevertheless, it is important to mention that several of the targeted scientific developments and targeted experiments can be easily transferred to other economics sectors, such as agriculture or to the general domain of Industry of the future, with for instance the task of autonomous goods transportation in warehouses.

In particular, safe and accurate coordination of Multi-Robot System (MRS) is a field of research of high effervescence. Indeed, this kind of system of large potentialities makes possible to carry out for example tasks which are unfeasible for only one robot (e.g., to move a too heavy or bulky object [199]) or improve certain criteria related to the velocity, the robustness or the flexibility of the task to achieve [368]. Inspired by these scientific challenges, this PhD work deals with cooperative autonomous vehicles management and navigation in complex environments/situations (mainly in terms of cooperative scheduling, planning and control).

The proposed PhD thesis in this manuscript is done and a combined effort in two laboratories: LIMOS (https://limos.fr/) and Institut Pascal (http://www.institutpascal.uca.fr/) in Université Clermont Auvergne (UCA). The collaboration was initiated and supervised by Prof. Alain Quilliot from LIMOS/Université Clermont Auvergne (UCA) and Prof. Lounis Adouane associated with Institut Pascal from Université de Technologie de Compiègne (UTC). Both of them made equal contribution to this PhD manuscript.

LIMOS (le Laboratoire d'Informatique, de Modélisation et d'Optimisation des Systèmes) is a collaborative research unit which is focused on computer science, and more generally in information and communication sciences and technologies. The scientific mission of the author's group which named *Outils Décisionnels pour la Production et les Services* (Decision-making Tools for Production and Services, ODPS) is concerned with management of operations in modes of transport with very different characteristics (road, rail, sea, air). The lab members are also committed to process issues as network design, time planning or calculating best routes in transportation network at different time horizons (strategic, tactical, operational, real-time, etc.) [38, 47, 48, 109, 119, 386].

Institut Pascal is a multi-disciplinary laboratory including several team clusters and addressing various engineer issues relating next generation transports, hospitals and factories. The author is one of the members in the *Modélisation, Autonomie et Contrôle dans les Systèmes Complexes* (Modeling, Autonomy and Control in Complex Systems, MACCS) team. It is worth mention that the robotics application especially for ground vehicles and intelligent passenger transportation systems are gained high research interests in Institut Pascal [51, 470, 473, 474]. The research theme covers a wide variety of perspectives, for instance, robotics for agricultural applications, Lyapunov based navigation of autonomous vehicles, platooning [52, 472, 474], Bayesian reasoning for decision making [217, 218], risk assessment/management for reliable navigation [45, 46] and decentralized multi-vehicle collaboration techniques [373].

It is worth also noting that both of LIMOS and Institut Pascal are highly involved in the LABEX (Laboratory of Excellence) IMobS3 (http://www.imobs3.uca.fr/), dedicated to innovative mobility. The IMobS3 LABEX project has novel features that are of considerable implementation of sustainable mobility solutions through cooperative approaches combining various subjects. The goal of this research is to exploit a hierarchical and hybrid decisional control architecture for improving the mobility/safety of Multi-vehicle cooperative navigation in a cramped, cluttered environment.

MAIN OBJECTIVES AND THEORETICAL APPROACH

Safe, efficient and flexible coordination of a group of autonomous vehicles in dynamical environments requires taking into account both inter-connected aspects: high-level (e.g., supervision for optimal progression of the mission; management of the existing interactions between the multi-vehicle systems) and low-level (e.g., control of the vehicles while taking into account their structural constraints: non-holonomy, maximum torque, avoiding encountered obstacles, etc.). The architecture of control/management that will be able to guaranty simultaneously these two aspects should be elaborated with modular and bottom-up manner (e.g., subsumption architecture like what has been introduced in [65]). The aim of following this methodology is to permit us to break the inherent complexity of the coordination of a high number of vehicles (autonomous agents) which must achieve efficiently the assigned mission. In addition, the application of the MVS is strongly influenced by many potential disturbing factors (sensor measured errors, communication congestion/packet losses, speed oscillations, etc.). The flexible and efficient strategic planning need to be properly designed so that will reduce the uncertainty and run a low navigation cost in such a multi-level configuration.

A MVS is controlled either while adopting a centralized approach (characterized by: global information on the environment, trajectories and/or missions planning, etc.) or while using a distributed approach that only uses local information of the environment. Each of these two approaches presents strong points and shortcomings. The centralized aspect of the control allows having more robust and reliable control; however, it requires quasi-complete information on the environment and on the task to achieve. Nevertheless, this is not always possible in the context of the MVS. Otherwise, it is in this context that distributed approaches take their full interest, because they only require partial information on the environment. Unfortunately, these last approaches have generally no prove on their convergence toward the global optimum that characterizes the cooperative task to achieve. However, it is possible to centralize only a part of the cooperative system to dispose the current state of the system and its short term evolution whilst preserving the decentralized part to perform cooperative navigation strategies [8, 146]. The works of

this PhD thesis consists in proposing suitable control/management architecture to obtain optimal and sub-optimal balance between centralized and distributed functionalities in order to enhance the overall efficiency of the MVS. Moreover, global performance metrics of the MVS were entailed by the developed hierarchical motion planning/control architecture, which allow the improvement of traffic stream mobility for an overall transportation task.

The high-level aspect of the control concerns the management of the inter-vehicles interactions that will permits us the right cohesion and efficiency of the MVS to achieve a desired global task. For this, the wireless communication network between vehicles serves an important role. Indeed, this support of communication will permit, according to the local perception of each vehicle, to synchronize global information on the evolution of the whole MVS [339]. From this information, each vehicle will be able to tack in a distributed manner, the most suitable decision/action according to the current system configuration. Nevertheless, this requires having suitable communication protocols that will be able to adapt to the different configurations of the environment, and to the possibilities of eclipses and failings of the wireless emission-reception. It will be proposed thus, and validated during this thesis appropriate communication protocols/architecture that should be perfectly integrated into the architecture of control/management of the MVS while allowing for rapid interaction (to acknowledge time scales and purposes of the inter-vehicles) [472].

Among the important items that will be addressed in this PhD thesis, one can mention:

- Safe and flexible cooperative navigation strategy for both optimizing progression of the collaboration task and reducing the likely risks between inter-vehicles (the main bibliographic part and contribution regarding to this aspect are given in Chapter 1 and Chapter 3 respectively).
- Appropriate construction of a hierarchical control architecture for the management/control of the MVS in constrained and dynamical environments (the main bibliographic part and contribution regarding to this aspect are given in Chapter 1 and Chapter 4 respectively).
- Global versus Local Planning-re-planning of multi-vehicle tasks and the influence of the local strategies adopted by the entities on the global evolution of the MVS (the main bibliographic part and contribution regarding to this aspect are given in Chapter 2 and Chapter 4 respectively).
- Eligible procedures/protocols in the proposed interaction network (the main contributions regarding this aspect are given both in Chapter 3 and Chapter 4.).

The main topics given above are gathered to belong to the main scientific topics of this PhD thesis. We further highlighted below the main research objectives and theoretical approaches in this manuscript:

 Safe and reliable Multi-Vehicle Coordination (MVC) and decision-making under uncertainties: The challenge consists of guaranteeing safe and reliable navigation of a MVS at critical time to manage in-road risks [45, 218, 470]. This will make it possible to deal with the uncertainties in real-time for such a critical cooperative system according to the context of navigation. One of the aims of this PhD thesis is to develop concepts, which should be enough generic in order to be applied for complex intersection/roundabout coordination. This could be resolved while taking inspiration from the already developed MVC techniques [470] and collective intelligence theory [261, 262, 263, 373], while having always flexible and robust collision avoidance functionalities [9] in order to manage any risky situations. It is also expected that an algorithm can guaranty the flexibility and the safety of the MVS at the time of transitions between the individual controls of each vehicle to the phase of coordinating the MVS when performing the complex maneuvers. The developed concepts/methods in this work should serve also several merged/separated platoons or single-vehicle dynamic w.r.t. the overall fluidity of traffic state. This could be resolved while taking inspiration from the already developed car-following/formation reconfiguration approach and the proposed techniques of traffic flow control [259, 472].

- · Risk management of autonomous vehicles in a transit network: The target in this part is to deal with a group of collaborative vehicles which are required to perform transportation tasks in a transit network (see [53, 137, 269, 477]). As previously said, when autonomous vehicles are engaged, safety is a concern. Further, it is discussed in this PhD manuscript the risk management of autonomous navigation in a high level. It is preferred that the top level agent (also known as a global supervisor) in the PhD thesis can be deployed in such restricted areas for professional purposes in a foreseeable term. More precisely, the global supervisor must compute and schedule routes in such a way that not only tasks are completed quickly, but also risk can be appropriately limited. The issue can be viewed as a Shortest Path under Risk constraint (SPR) problem [137, 401]. An important challenge in the field of SPR is guaranteeing rapid traversal time while ensuring the safety of traversal of any arc at a given time in an overall transit network. As a result, a common requirement for SPR is the need of a time dependent estimation of the risk induced by the traversal strategy even in uncertain environment [401, 533]. Thus, it is proposed a middle agent (called local supervisor) to be in charge of the small tricky areas which are risk sensitive. The local supervisor acts as a mediator, sending instructions to the vehicles in order to regulate their transit and avoid the risk of traffic congestion. Nevertheless, our goal is to compute and schedule the route of MVS, in such a way that its riding time is minimized and that induced risk estimation does not exceed some threshold. Therefore, motivated by vehicle routing problems in [356] and [430], the heuristic local search algorithm was further refined in the PhD manuscript.
- Robust and generic hierarchical and hybrid decisional control architecture: The inherent complexity of the cooperation/coordination of the movements between autonomous entities will be addressed while deeply investigating the potentialities of multi-vehicle hierarchical and hybrid decisional control architecture [8, 469, 476] (e.g., Figure 1 for example of a decentralized hierarchical architecture). Indeed, an autonomous MVS can compute feasible trajectories in a very complex physical network and requires, in addition, accurate and safe coordination between the vehicles (to cross an intersection for instance or to take a roundabout in cramped tricky areas). Thus, to control this complexity, it is planned to implement a hierarchical supervision architecture dedicated to managing such a system. The goal in this PhD thesis is to develop a robust and generic control architecture in 3 levels (see also Figure 1). A group of accurate and reliable strategies (collision avoid-



Figure 1: Multi-vehicle hierarchical and hybrid decisional control architecture.

ance [212, 264, 373], coordination and motion planning in multiple intersections [52, 470, 474], etc.) which link different traffic fundamental information (e.g., flow, density given by sensors in a mesh network) is involved for the interests of the decision-making of a MVS. An important part of the targeted work corresponds to finding the optimal/sub-optimal balance between control demands among unevenly distributed traffic demand and the control of the group of vehicles relevant for intersection crossing or roundabout navigation. Effectiveness in such a hierarchical and hybrid decisional scheme will be evaluated by safety, mobility and scalability of the MVS both in micro tricky areas and macro intersection networks.

MANUSCRIPT ORGANIZATION

According to the above mentioned PhD subjects, the manuscript are comprehensively divided into two parts. The outlines of the PhD manuscript is depicted in Figure 2. More precisely, the first part, which contains two chapters, targets to discuss the state of the art — in particular, the current research gap for autonomous entities cooperative motion planning and decision-making for the MVS in a complex transportation network — as follows.

Chapter 1 - Cooperative motion planning for autonomous vehicles

This chapter clarifies the main developments and control structures for MVS and makes the focus on the development of advanced cooperative motion techniques in cluttered/complex environments. More precisely, much focus has been paid on the effectiveness of the control system coping with single intersection driving strategies. The methods related to round-about and highway entrance/exit ramps are also delivered. The main challenges for multi-vehicle navigation in protected logistic regions and even urban areas are considered as well.

 Chapter 2 - Decision-making for multi-vehicle navigation in a transportation network



Figure 2: The outline of the PhD manuscript.

Through a broad analysis of published literature, this chapter investigates the MVS navigation management systems/tasks in a transportation network. Further, this chapter is dedicated to the related work for decision-making in multiple intersection networks. Particular attention is paid to the link between macroscopic traffic perturbation and ego-vehicle navigation strategy making based on statistic (and/or probabilistic) models. The commonly used techniques/algorithms to the fluidity of traffic in cities and the system limit of different mainstream traffic flow control models are presented. Additionally, an introduction of the promising approaches to resolve risk constraint decision-making issues in road networks is performed.

The latter part of the PhD manuscript is focused on the dissertation's main addressed approaches/proposals which can be categorized into two chapters (as can be seen also in Figure 2).

· Chapter 3 - Safe and flexible cooperative navigation with risk assessment

The state of the art can lead to a conclusion that a safe and flexible cooperative navigation scheme is crucial for a MVS to deal with in-road risks. In this aim, this chapter presents the reliable single intersection crossing strategy for the proposed cooperative navigation system. Particularly, collective intelligence theory applied in the mobile system will be further explained. Next, the ε -constrained Probability Collective (PC) algorithm is used to obtain safe and flexible solutions for determining the proper intersection crossing speeds. The robustness, scalability and flexibility of the proposed algorithm will be demonstrated as well. In addition, we extend the risk management problem from a global perspective. A local search heuristic algorithm is proposed in this chapter to deal with the SPR problem in mobile systems.

Chapter 4 - Proposed hierarchical traffic management architecture based on

Micro-Macro Flow Control (MiMaFC) strategy in traffic network

This chapter considers the problem of multi-vehicle cooperative navigation within the hierarchical control architecture. The main ideas in this chapter are: to extend the addressed algorithm in Chapter 3 while enabling its functionalities (e.g., collision-free, mobility, risk management) in the scaled network; to explore the influence of high traffic density that induces the decrease of the flow, or to show thus the limit of the system, which will validate at the same time the macroscopic flow modeling of the system; further, to demonstrate the advantages of establishing a link between the suggested local collective optimization and the macro model (i.e., the Micro-Macro Flow Control strategy: MiMaFC in micro model). Following that, it is proposed intersection navigation protocols in a deep relationship to our established intersection crossing methods. Finally, based on the proposed MiMaFC technique, the optimization procedures are integrated into the entire traffic navigation scheme for MVS. The suggested cooperative navigation management architecture is illustrated through a multitude and various simulation results in dedicated traffic networks.

General conclusion and main contributions in the PhD thesis are summarized at the end of the manuscript as well as several PhD prospects.

CONTEXT AND STATE OF THE ART

1

COOPERATIVE MOTION PLANNING FOR MULTI-VEHICLE NAVIGATION

This chapter is dedicated to introduce the development and control structures of MVS. The main focus will be paid on the cooperative technologies for a group of autonomous vehicles in cluttered environments. The multi-vehicle navigation scenarios are introduced in detail. The important characteristics/requirements for collaborated vehicles operating in various environments are underlined even further. We also bring up the issue of intersection motion planning. The related reactive/cognitive strategies to crossing an intersection are discussed by comparing to current scientific literature. Finally, the major challenges of multi-vehicle navigation in urban areas are summarized.

1.1/ INTRODUCTION

Over the past few decades, advanced technologies for Intelligent Vehicles (IVs) have gained considerable attention in research communities, industry managers and regulators. IVs provide with an ability to assist humans in achieving a higher level of maneuverability [21], [131]. Nevertheless, determining what the new generation of autonomous/semi-autonomous vehicles can exactly perform remains a critical topic. The idea to validate Artificial Intelligence (AI) techniques in autonomous vehicles system is also one of the most promising directions to develop intelligent driving. However, in order to successfully implement public transportation service with a sophisticated IVs system, other relevant automotive sector fields must also undergo adjustments and development (e.g., automotive design, advanced electronic and electrical products and policy setting). In densely populated metropolitan areas, the public's trust in such vehicles driving without external guidance remains low. Therefore, autonomous vehicles are identified currently at the "rock bottom" of the hype cycle from a perspective of the commercial institutions, as seen in Figure 1.1.

Instead, semi-autonomous or fully-autonomous vehicles are more likely to be restricted in the protected area in the near future, which includes professional purposes and/or dedicated operational conditions. In terms of the taxonomy of vehicles, SAE INTERNA-TIONAL's Standard J3016202104 provides six levels of driving automation [98], ranging from no driving automation (Level 0) to full driving automation (Level 5). Vehicles with Levels 3 (conditional driving automation) and 4 (high driving automation), in particular, still require a human driver in a supervisory capacity who is prepared to retake control in



Figure 1.1: 2021 Gartner hyper cycle for Artificial Intelligence (AI) [151]: Autonomous Vehicles at the "rock bottom" before entering the plateau of productivity.

certain situations. Additionally, each Level 3 and Level 4 vehicle will have its own Operational Design Domain (ODD)¹ [98, 144, 401] (see Figure 1.2). Consequently, autonomous vehicles are favored to travel freely inside some limited locations, such as vast parking lots, and to execute rural or urban logistics in lieu of the overly constricted Autonomous Guided Vehicles (AGVs) (tied to any kind of cable/rail track) deployed in warehouses or industrial structures.



Figure 1.2: Operational Design Domain (ODD) relative to driving automation [98].

¹"Operating conditions under which a given driving automation system or feature thereof is specifically designed to function, including, but not limited to, environmental, geographical and time-of-day restrictions, and/or the requisite presence or absence of certain traffic or roadway characteristics." ([98], Page 17)

Under this circumstance, one of the critical scientific issues for intelligent vehicles will be to govern and maintain a collection of autonomous vehicles that must accomplish navigation duties while safely engaging with collaborators in dynamic environments. The focus of the associated problem will be on using related decision approaches linked to MRS and Operational Research to satisfy the industrial and/or public transportation needs.

In this section, the essential characteristics of intelligent mobile robots and the transition process to build a multi-vehicle system will be discussed. In addition, different cooperative navigation control structures are evaluated in order to determine the most efficient and safe way to handle such a complicated system. The high-level automated vehicles that cooperative driving in common-yet-difficult conditions are then investigated further.

1.1.1/ FROM SINGLE-VEHICLE INTELLIGENCE TO MULTI-VEHICLE COLLABORA-TION

One of the goals of an intelligent unmanned system is to expand human geographical reach while decreasing risk of collision. The majority of robotic mobile systems are required to do jobs that are repetitive, unpleasant or dangerous [139]. In general, autonomous and Unmanned Vehicle System (UVS) may be described and categorized according to their operating environments (land, sea and air) with the most representative being [8]: Autonomous Underwater Vehicles (AUVs), Unmanned Surface Vehicles(USVs), Unmanned Maritime Vehicles (UMV)¹, Unmanned(or Uninhabited) Aerial Vehicles (UAVs) and Unmanned Ground Vehicles (UGVs)². Moreover, UGVs may move using various devices, like legs, wheels or specialized mobile robot wheels for planetary exploration (for more details, see chapter 1 in [8]). In particularly, the problems and primary works in this PhD manuscript are interpreted mainly in terms of UGVs with wheels. Apart from the application scenarios, a sophisticated UVS system must meet the following generic demands/requirements [139, 147, 349]:

- Persistence, low cost, stealth, and ready deploy/retrieve-ability;
- The capacity to detect, locate, track, identify and engage targets autonomously;
- The ability to gather, disseminate and act on several types of information;
- That they are networked together and to the high-value, manned assets;
- That the individual platform and sensor elements can be self-organized;
- That they do not impose significant risk or burden upon the operators.

UGVs are also known as mobile robots in the research community, which differs from traditional robotics (oriented to the control of industrial manipulators [147]). More specifically, mobile robots are preferred to deal with problem related to path planning, obstacle avoidance, and perceptual control. The involved navigation tasks/control issues in the application domain of UGVs can be categorized, for instance, in terms of surveillance

¹Unmanned Maritime Vehicles (UMV) are "Comprising Unmanned Underwater Vehicles (UUVs) and Unmanned Surface Vehicles (USVs)." ([139], Page 2)

²"An Unmanned Ground Vehicle (UGV) is any piece of mechanized equipment that moves across the surface of the ground and serves as a means of carrying or transporting something, but explicitly does not carry a human being." ([147], Page 1)

[63, 350, 528], visual navigation [227] [267], human search and rescue (disaster robotics) [237] [343], solar tracking [15] [185], military [390], precision agriculture/farming [384] [385], service robotics [108, 173, 370] or transportation/automated guided [36, 436, 470]. Nevertheless, it is worth mention that UGVs can also deal with tasks incorporating both mobility and manipulation as seen in NASA's space robotics project: Robonaut [22], Regolith Advanced Surface Systems Operations Robot (RASSOR) Excavator [340], Curiosity (mars rover) [411] and Valkeyrie (robot) [387]. Despite the fact that the generated UGVs control method/concepts might be applied to the various tasks/domains listed above, the transportation domain (both for passengers and for products) remains our primary focus.

Indeed, one of the most important areas of UGVs research is intelligent vehicles (also known as self-driving cars, autonomous vehicles, driver-less car or robotic car) for public/private transit. The deployment of a fully (SAE Level 5) or semi-autonomous system (SAE Level 3/Level 4) in the surface transportation system have drawn much attendance/efforts from government institutions, universities, commercial Big Tech companies and car manufacturers. The early research efforts for developing intelligent vehicles are referred to the numerous interesting publications in the literature [8, 143, 297, 443, 456, 474, 478, 489]. In this PhD manuscript, a short view of autonomous vehicle development is presented in a timeline covering the recent two decades (as seen in Figure 1.3) and thereby follows a brief summary.

In the 2000s, large-scale self-driving activities fueled the development of intelligent vehicle technology. The DARPA (Defense Advanced Research Projects Agency) Grand Challenge [68], which was held in 2004, 2005 and 2007, had a significant impact on future self-driving automobile development. Further, other autonomous vehicle events, such as the European Land Robot Trial (ELROB) [407], Intelligent Vehicle Future Challenge (IVFC) [503], and SparkFun Autonomous Vehicle Competition (AVC) [120], were held on a yearly basis in the last two decades. Moreover, the early development of self-driving testing datasets (for object detection and recognition) was founded mostly for individual naturalistic driving data. See the 100-Car Naturalistic Driving Studying [254], Strategic Highway Research Program 2 (SHRP 2) [60] and European Field Operational Test (EuroFOT) [50] for instance. Google's self-driving project which is the best-known as Waymo later [303] was performed initially in 2009.

From the 2010s till now (2021), single intelligent vehicles' technologies were in a new age of rapid progress. The startup technology companies supported by/combined with academia aimed to develop their own self-driving vehicles. We don't aim to exhaust all the lists of technology companies developing self-driving cars. One can find the most notable examples in the Figure 1.3 (like: Transdev, Lyft, Cruise, Zoox, NAVYA, Uber, Argo AI, AutoX, Pony.ai, JD, Aurora, Zenuity, Aptiv and Didi among others). Particularly, the aforementioned technology companies, hardware developers and research institutes have released advanced simulation tools (e.g., Gazebo [256], TORCS [500], CARLA [115] and others) as well as self-driving frameworks (e.g., Autoware [242], Nvidia DriveWorks [361], openpilot [97] and Apollo [128], etc.), which are very useful for developing intelligent vehicles. Furthermore, various training datasets have been utilized to improve computer vision technology in self-driving cars, due to the application of machine learning to large-scale data. This study also list the most remarkable ones as: ImageNet [110], KITTI Vision Benchmark [157], MS COCO [294], Oxford RobotCar [313], DAVIS Driving Dataset (DDD17) [59], CommonRoad [20], LiDAR-Video Driving Dataset (LiVi-Set) [89], ApolloScape [211], UC Berkeley DeepDrive (BDD) [523], nuScenes datasets [73] and



Figure 1.3: A brief view of autonomous vehicle development in the last two decades.

others. Traditional vehicle manufacturers, on the other hand, further unveiled their newgeneration private sedans claiming to enable SAE Level 2 to 4. However, because of the ambiguous policy obligations for self-driving car accidents, automobile manufacturers have had to abandon or postpone their plans to provide customers with full autonomous driving technology on roads. More information on Audi, Renault, Toyota, Mercedes Benz, Honda, BMW, and General Motors, among other companies, can be also found in Figure 1.3. For further information on the development of single-vehicle intelligence, interested readers are referred to [37, 170, 371, 409, 512, 526].

A natural extension of a single intelligent vehicle is to deal with Multiple intelligent Vehicle System (still referred MVS). Recall MRS as a reference. Due to the enhancement of the dynamic interaction between robots, controlling MRS rather than a single robot significantly increases the control complexity: for instance, the amount of control variables and sub-objectives to investigate/achieve; the uncertainty associated with observing/localizing/communicating with a collection of robots [8]. The primary benefit of MRS is that it enables the completion of complicated tasks that are either too demanding for a single robot or are intrinsically distributed [26, 75, 369, 452]. Additionally, constructing several resource-constrained robots is less complicated than developing a single powerful robot [265, 325, 400]. A common categorization strategy for cooperative robotic systems is based on information exchange, which is separated into collective swarm systems and intentionally cooperative systems [420] (see Figure 1.4 for instance). In collective swarm systems, cooperative robots only interact locally to complete self-assigned tasks. Typical applications include cooperative exploration robots [276, 357] (Figure 1.4a), entertainment robots [17], localization & mapping robots [72, 279, 402] (Figure 1.4c) and warehouse robots [311, 378] (Figure 1.4f). In intentionally cooperative systems, all system agents have complete knowledge about the same aim to co-manipulate or co-transport objects [8, 134] (Figure 1.4d,1.4e). Some examples can be found in co-manipulation robots [201], load transport robots [154] and construction robots [298, 487]. Since its fast growth in the 1970s and 1980s, MRS in the form of mobile robots has been extensively utilized and tested by a variety of research goals, tasks and projects. MRS technology was encouraged in physical cooperative vehicles in the 1990s by information technology and vision-based control. A brief history of the MRS may be found in [474].

MVS is supposed to engage in cooperative navigation tasks [481], sharing certain characteristics and benefits such as MRS. As indicated before, this PhD thesis focuses on cooperative navigation for public transit. Thus, Connected Autonomous/Automated Vehicles (CAVs), which have shown immense promise as a component of future transportation systems [78, 145, 424], will be further discussed in the latter of the section. Noting that the proposed MVS makes the assumption that connection is always available, either locally or globally. As a result, we made no clear distinction between MVS and CAVs.

Similar to MRS, CAVs provide appealing benefits in terms of operating in resourceconstrained environments, resolving problems concurrently, considerably reducing accidents and related costs [363, 475]. Vehicle connectivity is a new technology that enables Vehicle-to-infrastructure (V2I)/Infrastructure-to-vehicle (I2V), Vehicle-to-vehicle (V2V) and Vehicle-to-everything (V2X) communications [61, 312]. Vehicles equipped with interactive Advanced Driver Assistance Systems (ADASs) and cooperative Intelligent Transportation Systems (C-ITSs) are typically considered to be "connected" [455]. Car connection may give both the regular vehicle and the autonomous vehicle with new information and services. CAVs are capable of receiving information from V2X outside the field of vision and negotiating with other road users. Thus, CAVs may outperform the single intelligent ve-
hicle not just in terms of safety and performance, but also in terms of traffic throughput and fuel efficiency via global route planning and cooperative driving [124]. In addition, CAVs cooperating with other road users can be classified into two types: informationbased and maneuver-based cooperation [70]. In information-based cooperation, such as Cooperative Adaptive Cruise Control (CACC) [250, 286, 296] and cooperative perception/prediction [252, 253], agents communicate their own knowledge (e.g., system states, sensor data and intention) with one another, and they use the received information to maximize their own utility [62]. In maneuver-based cooperation, vehicles get sensor data, intentions or intended trajectory from other CAVs and use planning to maximize an estimated/negotiated total utility [70]. The following are the empirical applications and research works for using CAVs in both motorways and urban environments (as seen in Figure 1.5): telematics applications/devices plugged in CAVs (including Vehicleto-broadband, telemetry services, etc.) [421]; fleet/convoy management [392, 474] (Figure 1.5a); Electronic Toll Collection (ETC) system with CAVs [95, 274] (Figure 1.5b); HD map data collection by CAVs [40, 258] (Figure 1.5c). It is difficult to test most CAVs technology on public highways, and there is a lack of experimental validation in real-world conditions [181]. Therefore, several programs based on CAVs have been established by governments or national organizations. The following highlight several interesting examples of CAVs projects around the world:

- PATH: Caltrans and the University of California jointly created the PATH program in more than two decades to guide the implementation of CAVs in California [417]. The PATH initiative focused on difficulties associated with navigation, automation and electrification in increasingly sophisticated traffic management systems. In 1997, the PATH project showed platoon driving and safe operations such as lane changes using numerous autos (Honda-PATH vehicles) [389, 437]. In 2011, as part of the PATH project, a platoon of autonomous trucks was driven [55, 220]. Additionally, PATH has developed a CACC system for platooning heavy trucks in partnership with the Volvo company since 2015 [221] (Figure 1.5d).
- · Google's self-driving car: Google's CAVs have tested more than 3.5 million au-



(a) Mars exploration robotics [357].



(d) Co-manipulate robots [201] .



(b) A swarm of 1024 Kilobot [400].



(e) Load and co-transport robots [154].



(c) Localization robots [431].



(f) Warehouse robots [320].



tonomous miles (5.63 million kilometers) since 2009 [25]. In 2012, Google demonstrated a fleet of seven automated Toyota Priuses that traveled more than 140 thousand miles (225 thousand kilometers) in California [29, 492]. The vehicle uses data from Google Street View in combination with perceptive information to determine its location and ensure safe operation in urban environments [29, 492]. In 2017, Google invested 1 billion USD in Lyft to support Waymo's Robo-taxi fleet [257] (Figure 1.5e).

- Energy-ITS: Since 2008, the Japanese Ministry of Economy, Trade, and Industry has been administering the Energy-ITS program [450]. The project's objective is to save energy and avert global warming via autonomous driving. On a test vehicle, a platoon of three completely automated heavy trucks and one fully automated light truck traveled at 80 kilometers per hour with a distance of up to 4.7 meters [450] (Figure 1.5f). The lateral control was based on computer vision-based lane marker identification, while the longitudinal control was based on gap measurement using radar and LIDAR with Dedicated Short Range Communication (DSRC) and infrared inter-vehicle communications. Additionally, this research included testing on four big vehicles equipped with Cooperative Adaptive Cruise Control (CACC) [450].
- Demo 2000: In November 2000, "Demo 2000 Cooperative Driving" was performed with five autonomous cars to demonstrate the technology's viability and promise [243]. The vehicle uses RTK-GPS for localisation and is equipped with a laser radar, an inter-vehicle communication unit, and onboard displays to allow passengers to locate nearby cars and obstacles. On a test track, the five autonomous vehicles demonstrated scenarios such as stop-and-go, platooning, lane changing, passing and merging at speeds of 40 ~ 60 kilometers per hour [451].
- **SARTRE:** The SARTRE program (SAfe Road TRains for the Environment) [79] evaluated platooning (Volvo vehicles and trucks) in public traffic conditions by using CACC, which has the potential to enhance traffic flow and safety. From September 2009 to September 2012, the European Commission supported this project [80, 105].



Figure 1.5: Application of Connected Autonomous/Automated Vehicles.

- GCDC: The Grand Cooperative Driving Challenge (GCDC) was an international competition in which teams from universities and industry competed to maily test CACC (as seen in Netherlands' GCDC in 2011) [376]. CACC performance was evaluated using a number of parameters, including platoon gap, travel time, platoon merging behavior and damping behavior during rapid acceleration/deceleration [156]. In 2016, the Netherlands hosted another GCDC challenge aimed at achieving cooperative merging and intersection management [375].
- SAE: SAE International has established many standards for Dedicated Short Range Communication (DSRC) application layers, including SAE J2735 [222], SAE J2945 [223] and SAE J3067 [224]. The FCC reallocated all of DSRC's spectrum to other purposes (i.e., Wi-Fi and cellular V2X) in November 2020 [491]. In the same year, SAE introduced a revised standard for V2X communication [225].

Although decades of research and development have gone into CAVs systems, a great deal of technology has the potential to benefit road users. CAVs/MVS are still confronted with a slew of challenges as follows [124, 181, 317, 474]:

- Accurate perception/location data: Perception errors may be exacerbated in CAVs. A data association mechanism that is efficient for a variety of vehicle and sensor designs is required. The performance of cooperative perception is highly dependent on the accuracy of relative localisation. However, relative localisation accuracy may be limited in particular circumstances.
- Real-time coordination: In multi-agent situations characterized by dynamic environments, cooperative motion planning for a new maneuver frequently becomes a matter of system safety. This type of planning is highly dependent on on-board algorithms and is quite sensitive to nearby traffic. As a result, real-time coordination is required for the on-board controller.
- **Communication:** Unified communication topologies and protocols for deploying cooperative CAVs systems are lacking. A reliable control approach is required to handle communication delays, packet loss, error messages and uncooperative agents. Additionally, transmission latency and bandwidth constraints may decrease cooperative perception's efficiency. The trade-off between communication bandwidth and closed-loop performance has received less attention.
- Accurate forecasts: Due to the absence of an integrated CAVs control architecture that enables access to more precise forecasts, the majority of existing approaches rely only on static route data (such as road grade and speed limits). Nevertheless, CAVs prospects are often associated with improved forecasts for data and remote computations, which are highlighted as the primary technological difficulties.
- Real-world verification The majority of CAVs technology (e.g., CACC) has been undertaken in controlled environments, difficult to know its efficacy. Current verification issues for CAVs' advanced algorithms include real-world testing and application to non-highway scenarios.

Indeed, CAVs can eliminate (or significantly reduce) unexpected human factors and gain awareness of their surroundings via perception sensors and communication. There are

numerous scenarios where V2X communication provide a significant advantages. However, CAVs running on public roads that take advantage of this benefit remain need to be fully explored and exploited. Control of CAVs system should also take into consideration the partial or total system state, rather than relying just on the perception or aims of a single intelligent vehicle [8]. One of the primary issues for CAVs cooperative navigation is developing an effective control strategy that enables all vehicles in the CAVs system to establish coherent and efficient configurations to accomplish the desired tasks. In fact, with vehicle communication, CAVs may exchange their perceptive data and current condition with other road users (for example, intended acceleration, actual acceleration or velocity) [181]. As a result, control structures for processing system information have become critical for implementing such networking technologies efficiently. An overview of CAVs/MVS control architecture will be discussed in the next section.

1.1.2/ OVERVIEW OF MULTI-VEHICLE CONTROL ARCHITECTURES

As seen in Figure 1.6, the MVS is comprised of multiple modules deployed in each connected vehicle. More specifically, multi-vehicle control architectures addressed here refer to the system architectures that operate the communication, perception, localization and planning modules (see also in Figure 1.6). Numerous control architectures for MVS are described in the literature [10, 124, 200, 307, 471, 481]. Clearly, exhaustively classifying the control structures of MVS would not be the primary objective in this PhD thesis. Nevertheless, one of the critical concerns to address before developing the multi-vehicle control architectures is whether to use centralized or decentralized control over vehicles [8, 396]. This section will further evaluate the centralized vs. decentralized management of MVS.

1.1.2.1/ CENTRALIZED ARCHITECTURES

The MVS architecture is said to be centralized when a part of all the sensory and/or decisional loops of each vehicle entity is delocalized from its physical structure and governed by a central unit (referred to as a supervisor or central planner) [236, 359]. The primary benefit of this architecture is that a central unit can make decisions based on global knowledge, which is often superior than an agent making decisions based on local information



Figure 1.6: General architecture of a MVS with communication, perception, localization, planning and control modules [481].

[249]. Thus, centralized systems often need a large amount of computing power to analyze massive data through extensive communication [249, 527]. Such architecture lacks robustness as a result of its deep reliance on the central unit.

In terms of implementation, the targeted communication modules, perception/localization modules and motion modules may all be handled by centralized architecture in the fields of information flow topology design [543], sensor fusion approach [180, 490], and cooperative planning [396]. A centralized control method for MVS is explored in [86], with the purpose of minimizing a cost function that incorporates MVS safety, efficiency and ride comfort. Noting that centralized planning, based on V2V and V2I communication, may give a dependable solution that is fully aware of the states and intents of other cars. The sophisticated 5G technology standard is gaining popularity as a centralized method of V2V and V2I communication. However, its longer end-to-end latency (which may be much longer during handovers) prohibits its use in vehicle safety [312]. Numerous MVS and mathematical optimization techniques are used to address the problem of centralized planning, as described in V2V and V2I [85, 113, 126, 358]. However, V2V and V2I are not yet widely used. As a consequence, centralized planning will be difficult to implement in the near future.

1.1.2.2/ DECENTRALIZED ARCHITECTURES

In contrast, in the decentralized (distributed) control architecture, each on-board module (see Figure 1.6) has its own perception, localization and decision-making (planning) process among a group of vehicles [8]. Due to the fact that MVS can always conduct control on their own without receiving extra commands from outside, the benefits of this decentralized design allow robustness to system defects and failures. Additionally, decentralized control enables the parallel solutions, hence ensuring the implementation's dependability and scalability [75, 130, 138, 393]. The disadvantages of such a decentralized management need a high degree of coordination, since the given tasks of each vehicle are incorporated in the local control, and if the assigned work changes, global reconfiguration of the MVS tasks without a supervisor may be problematic.

Decentralized (distributed) control may also be used to MVS in terms of communication technology [5, 289], perception technique (decentralized fusion [180, 490]), and cooperative motion planning [396]. The research in [318] established a decentralized theoretical framework for MVS coordination. Thought has been paid to rear-end, speed-dependent safety constraints. Additionally, as compared to cellular networks such as 5G, DSRC is preferable for road safety driving because to its decentralized nature and low latency for end-to-end transmission [418]. Since maximizing the flexibility and autonomy of controlled MVS is often preferred. Some of the above-mentioned literature extensively researches decentralized multi-vehicle control systems in complex environments or situations (mainly in terms of cooperative scheduling, planning and control in industry, warehouses, hospital and urban backgrounds, etc.).

1.1.2.3/ OTHER ARCHITECTURES

Other studies with similar discussion in various control approach/architecture might also be discovered in [84, 162, 231, 379, 515, 546]. We highlight several kinds of notable control architectures as follows:

- Hierarchical architecture: In a global sense, this architecture is decentralized control, but it is centralized at the local level [326, 354]. More specifically, the vehicle conducts its assigned responsibilities independently while update a central planner on the status of those activities. The great majority of MVS are structured hierarchically [164, 241]. The idea is that every large-scale system's control is structured in a dispersed hierarchy [465]. With this approach, a large design challenge is partitioned into a number of smaller, more manageable sub-problems that are handled in distinct layers. Since MVS tasks become more and more complicated, it is preferable to have hierarchical architecture that can capture the complicated interactions.
- Hybrid Centralized/Decentralized architecture: This architecture integrates a high-level controller (centralized planner) with the local control (decentralized executor) in individual vehicles [359, 420, 474]. As a consequence, the primary advantages of this designs include: the centralized planner which can act as a high-level control over the autonomous vehicles; the decentralized executor's fault tolerance; the flexibility to reschedule the global task/control based on both (centralized/decentralized) control. As shown in [422], researchers use this architecture to conduct a high-precision docking task. In addition to vehicle navigation in formation, centralized/decentralized architecture was successfully used in [474].

Roughly, the possibility also exists to centralize only a part of the control and let the other part be decentralized (hybrid centralized/decentralized control) [146]. For example, hybrid fusion is common in CAVs systems. The Global Navigation Satellite System (GNSS) information from centralized planner is unreliable while the ego-CAV is under tunnels or bridges. So, in addition to the on-board Inertial Navigation System (INS), the CAVs location may be enhanced by incorporating perception sensor data coupled with a reference map [481]. Noting that the addressed multi-vehicle control structure must manage data from several devices, the primary challenge is to use data from the communication/perception module to complete navigation tasks in complicated environments/situations through information-based or maneuver-based cooperation (e.g., vehicle classification, vehicle tracking, lane detection [549] and cooperative navigation [124]). The primary focus of this PhD thesis is to develop a control architecture for maneuverbased collaboration that is compatible with the individual planning module of the targeted MVS. While decentralized control is always favored in our proposed control architecture for MVS, global knowledge obtained by a central planner can be used to enhance MVS control in certain scenarios (or tasks). We will further specify the research problem in the following section on exploring MVS navigation environments and situations. Notably, in the rest of the manuscript, when we mention MVS, we also refer to CAVs, since both of them perform cooperative navigation tasks and make the assumption that connectivity is always available, either locally or globally.

1.1.3/ MULTI-VEHICLE NAVIGATION IN COMPLEX ENVIRONMENT/SITUATIONS

According to literature reviews, MVS (or CAVs) coordination issues that emerge in rural and urban areas include motorway/highway speed coordination, merging at on-ramp and coordination at signalized/autonomous intersections [56, 181, 481]. Figure 1.7 depicts a visual categorization of these MVS application scenarios. Additionally, MVS platooning technology, which has a significant impact on road capacity [23, 539], has garnered considerable attention during the last decades. Moreover, the majority of the fundamental



(c) Principle/minor arterial.

(d) Urban core/intersection area.

Figure 1.7: A classification of MVS coordination scenarios from both rural and urban areas (cite from https://www.freepik.com/).

research efforts have been devoted at MVS highway speed harmonization [197, 365], which has the potential to lower overall trip time and average energy consumption [479]. Readers interested in MVS freeway longitudinal motion control (mainly in terms of platooning and speed harmonization) may read [181, 204, 248, 481]. It should be noted that lateral maneuvers (mostly seen on principle/minor arterial, see Figure 1.7c) such as maintaining or changing lanes are not considered in this manuscript. This section will examine multi-vehicle navigation in complex environment/situations such as highway entrance/exit ramps and intersections/roundabouts.

1.1.3.1/ HIGHWAY ENTRANCE/EXIT RAMPS

Indeed, to avoid lateral collision, highway entrance/exit ramps (as seen in Figure 1.8) create issues about safety and mobility for MVS coordination. An overview of studies on highway on-ramp can be found in [396, 404, 541]. Ramp metering, for example, is a common use for highway traffic management that is mostly done using traffic lights positioned at highway entrances [366]. In contrast, the challenge of seamlessly merging or crossing two streams of vehicles without causing stop-and-go driving is similar to the problem of speed harmonization. Under this situation, vehicles on the mainline may decide to accelerate or decelerate in order to enable the merging of vehicles on the other lane, altering overall traffic flow [332]. Similarly, it has been shown that saturated exit ramp (also known as off-ramp) attract a lot of attention to creating effective traffic management to prevent the off-ramp queue spill back into the freeway [230, 342, 427]. The primary idea of generating gaps by deceleration and lane changes may also be used nearby off-ramp (at a diverging bottleneck area). However, when implemented on off-ramp, the MVS cooperative technique is primarily developed by automation and connection technology that use lane change tactics [544]. As a result, in the following text, the multi-vehicle navigation approach will be analyzed in more details in terms of entrance ramp for merging.



(a) Highway entrance ramp (on-ramp).



(b) Highway exit ramp (off-ramp).

Figure 1.8: Merging/splitting maneuvers at highway entrance/exit ramps (designed on https://icograms.com/).

Several algorithms for assisting on-ramp merging have been suggested, including controlling mainline vehicle deceleration to create gaps for MVS [367, 380, 403]. Additionally, cooperative control algorithms were developed between ramp and mainline vehicles to facilitate merging operations [403, 502]. In [458], a virtual vehicle is created by mapping a car on one lane to another. This enables longitudinal control between a mainline vehicle and one on a ramp, allowing for seamless merging. A distributed cooperative highway on-ramp merging system is presented in [482], which utilizes both V2V and I2V communication and involves the formation of two vehicle strings on the main line and on the on-ramp. Particularly, [169] proposes a protocol for merging vehicles from the neighboring lane, with a special emphasis on the communications exchanged between vehicles. The authors in [395] establish an optimization framework and a closed-form analytical solution for online vehicle coordination in the merging zone. Although there has been a lot of published work on autonomous highway entry ramp, there has been far less practical implementation of cooperative merging throughout the world [481]. According to a study from the PATH program, testing on cooperative automatic merging systems was done at the Richmond Field Station, U.C. Berkeley, and Crows Landing, California in the US [309]. The University of Minnesota Duluth [214], as well as East Tennessee State University [11, 12], conducted comparable experimental implementations on highway on-ramp merging situations using V2V communication [481].

1.1.3.2/ INTERSECTION/ROUNDABOUT

In urban core areas (see Figure 1.7d), multi-vehicle cooperative navigation becomes highly difficult under a variety of urban situations involving numerous road users. Several MVS applications have been explored to benefit safety, mobility and the environment with connected vehicles [460]. Among them, MVS cooperation technology for intersection or roundabout applications is particularly representative and significant effort has been done to improve traffic safety/mobility in such densely urban regions. More precisely, research on MVS cooperative driving is conducted at both signalized and unsignalized intersection/roundabout (see Figure 1.9 for instance). Roughly, V2I communication is often preferred at signalized intersections in order to get Signal Phase and Time (SPaT) information and prevent unnecessary speed changes or complete stops [7]. Within an intelligent transportation system, V2V and I2Vcommunication are often used at unsignalized intersections [181, 481]. Thus, following the planning and scheduling algorithm, MVS can be assigned specific sequences to cross the intersection/roundabout. Theoretical studies and applications are given below.



Figure 1.9: Signalized and unsignalized intersection/roundabout (designed on https://icograms.com/).

In fact, the research work related to MVS navigation in intersection/roundabout has gained considerable attention during the last decades. Interesting and comprehensive surveys about this problem are analyzed in [85, 121, 183, 345, 396]. Several signalbased control systems assured the efficient and economic management of intersections and aided in the alleviation of traffic congestion [111, 524]. The authors in [513] propose an Eco-CACC algorithm that computes the most fuel-efficient vehicle trajectory through a signalized intersection by guaranteeing that the vehicle arrives at the intersection stop bar as soon as the last connected vehicle in the queue is discharged. The work in [540] considers a real-time cooperative eco-driving approach for a group of vehicles with a mix of autonomous vehicles (AVs) and human-driven vehicles (HVs) crossing a signalized intersection. As a result of newly developed vehicle communication technology, a large range of unsignalized intersection management approaches are also introduced in recent literature [88, 508]. It provides promise for Cooperative Intersection Control (CIC), particularly in terms of adapting to the use of MVS. Additionally, those CIC methods can be classified into: cooperative resource reservation techniques, trajectory planning approaches and virtual traffic lights solutions [85]. In an autonomous intersection study from [117], the reservation-based technique based on a communication protocol is shown to outperform existing intersections with either traffic lights or stop signs. To develop a CIC strategy, researchers in [327] propose the notion of "virtual platooning", which enables cars in separate lanes of the intersection with different destinations to form platoons. Similar platoonbased intersection management methods can also be found in [235, 505]. Nevertheless, MVS collaboration studies at intersections/roundabouts confront similar challenges in terms of implementation in today's traffic/infrastructure settings. Thus, the United States Department of Transportation initiatives the Multi-Modal Intelligent Traffic Signal System (MMITSS) project to offer a series of advanced intersection control applications for MVS [459]. In Europe, INRIA (Institut national de recherche en sciences et technologies du numérique) in France has done many experimental implementations of automatic coordination at unsignalized intersection, including the Cybercars and Cybercars-2 projects [64, 107].

Recent research has explored these MVS cooperative navigation challenges on highway

ramps and intersections/roundabouts [181, 317], assuming all cars are MVS. This issue has sparked considerable interest in the control community, with plenty of literature [396]. The challenges of merging at highway entrance/exit ramps and intersection/roundabout may be tackled similarly using comparable assumptions and ideas [360, 395, 537]. Apparently, these assumptions may be altered; for example, [534] discusses a MVS application that allows both left and right turns at intersections. In [93], the authors investigate V2V communication limitation situations for MVS cooperative merging control. Although those studies' findings are attractive, we emphasize three critical issues where application concerns need to be addressed further, as follows.

- The described MVS coordination issue mainly relies on the on-board algorithm. For deployment on public roadways, planning must be updated in real-time to respond to changing traffic circumstances.
- Automated Highway System (AHS) research has proven significant promise in freeway environment (see Figure 1.7a). Simulations and trials of varying complexity have been conducted on motorway/highway [428], further study is required to implement these MVS collaboration on urban areas (as seen in Figure 1.7d).
- MVS with intersection management has emerged as a prominent research field for the use of cooperative technology inside a specific control framework. However, the system lacks a unified communication protocol and requires a more risk-sensitive design to ensure reliability.

As a result, this PhD thesis will concentrate on the multi-vehicle navigation challenge particular to intersection/roundabouts. It is worth noting that the intersection control and highway on-ramp merging control are quite similar in nature, and that the majority of methodologies given for intersection control may simply be modified for merging coordination for an on-ramp, and vice versa [396]. Further, MVS can both apply information-based cooperation (optimized their own utility) [62] and maneuver-based cooperation (optimize the total utility in a system) [70] for intersection management. To avoid any misunderstanding, the issue we discussed in this work primarily refers to maneuver-based collaboration for the motion planning of MVS. Thus, the intersection management method and navigation strategy of MVS will be discussed in more detail in the next section.

1.2/ INTELLIGENT VEHICLE'S MOTION PLANNING AT AN INTERSEC-TION

This section will highlight the primary intersection model, typologies, assumptions and driving strategies enabling the application of MVS/CAVs. The purpose of this part is to explain related terms and to classify the main methodologies created for intersection/roundabout motion planning. Following that, an overview of the major challenges is given in the summary.

1.2.1/ AN OVERVIEW OF THE INTERSECTION MANAGEMENT SYSTEM

The situation shown in Figure 1.10 demonstrates how MVS may be used to cross intersections. There is a gray *Buffer area* [292] with a length of S_0 for vehicles to approach



Figure 1.10: General intersection management with MVS.

the intersection. By the length of S_1 , connected vehicles in the green region referred to as the *Decision-making area* may interact with other agents (vehicles or coordinators). The red *Core area* (alternatively referred to as the conflict area or intersection zone [396]) in length S_2 is located at the merging point of multiple roads. The MVS longitudinal kinematics model can be illustrated by a discrete model like the following:

$$\begin{cases} a_i(t) = u_i(t) \\ v_i(t + \Delta t) = v_i(t) + a_i(t) \times \Delta t \\ x_i(t + \Delta t) = x_i(t) + v_i(t) \times \Delta t + \frac{1}{2}a_i(t) \times \Delta t^2 \end{cases}$$
(1.1)

Where $x_i(t)$ and $v_i(t)$ denote respectively the displacement and velocity of the vehicle *i* at time instant *t*. $a_i(t)$ is the corresponding acceleration for a time interval Δt . Besides, $a_i(t)$ is addressed by the control input $u_i(t)$. The purpose here is likely to compute $u_i(t)$ for each vehicle to safely cross the intersection, i.e., without rear-end and/or lateral collisions.

Clearly, there are more conflicts at intersections [278]. Left-turning of MVS raises the chance of crashing much more than straightening or turning right [90]. From an intersection management perspective, the present study will concentrate on the isolated intersection driving strategy (In Chapter 2, we will examine more complex instances involving multiple intersections). Additionally, the intersection structure, other than crossroads and roundabouts, is rather sophisticated in terms of topology. A diagram generated by [485] (i.e., Figure 1.11) reveals further information regarding the intersection topology. However, the nature of these intersection management issues is extremely similar, and they may be simply transformed by referring to the crossroad (i.e., Figure 1.11a). As a consequence, the focus dedicated to the technique part of this PhD thesis will not include the intersection topology. All of the intersection navigation scenarios covered in the manuscript have been handled using the topology shown in Figure 1.10.



Figure 1.11: The intersection structure with different topologies [485].

According to previous research, intelligent intersection management tactics may be categorized into two categories: a) traffic management at the high level; b) MVS driving behaviors at the low level [485, 536]. In fact, there is currently no complete intersection management system capable of integrating autonomous vehicles' safety driving with these two levels at isolated intersections. Significant progress has been made in the first study topic (i.e., traffic management level) through optimizing traffic light control algorithms. Due to the fact that traffic lights are more realistic solutions for today's traffic conditions, they are considered one of the most effective method of ensuring regular traffic flows at intersections [13, 396]. Particularly, in [117], a multi-agent strategy was used to offer a novel Autonomous Intersection Management (AIM) policy termed First Come First Serve (FCFS) Light. To manage mixed traffic, it employs a reservation-based scheme for autonomous vehicles and traffic signals for human-driven vehicles. By expanding the model described in [117], the authors introduced a novel protocol named Hybrid Autonomous Intersection Management (H-AIM) in order to enhance intersection performance under mixed traffic situations in [414]. The authors in [290] suggested two adaptive signal control algorithms that are successful in alleviating traffic congestion and achieving adaptive signal control goals using real-time traffic data. Another adaptive traffic signal controller based on reinforcement learning was presented in [412]. Additionally, mathematics model [34, 129, 295], fuzzy logic [355, 388, 440], Petri Net-based control [76, 148, 300], queuing theory and agent-based learning approaches [42, 335, 547] are all critical areas of traffic light study. Several reviews and surveys on signalized intersection control may be found for further summary [85, 396]. However, it has been shown earlier that standard traffic signals are inefficient at large traffic volumes [415]. Meanwhile, as predicted by IEEE, traffic signals may be obsolete by 2040 [219]. Thus, as MVS take over the major roads, the need for traffic signals at intersections reduces. As a result, more efforts have been made for unsignalized intersection management with MVS motion planning (i.e., low-level link to autonomous vehicle's behaviors). Due of the uncertainty in human-driven vehicles' capacity to communicate and collaborate with other road users [345], mixed traffic at unsignalized intersections will not be addressed in the context of this PhD dissertation.

The works cited above give a brief overview of intersection management systems. The following section discusses the primary techniques for non-signalized intersections with MVS. Given that we will use the terms *Autonomous Intersection Management* (*AIM*)/*Decision-making and trajectory planning*/*Multi-agent system* (*MAS*), etc. to de-

fine the proposals and deal with key scenarios in this PhD thesis, we provide a brief description to help clarify the concepts.

- Autonomous Intersection Management (AIM): the notion of AIM was first introduced by Dresner and Stone [117] as an alternative intersection manager (compared to current traffic signals and stop signs). AIM is expected to increase the efficiency of the present intersection by using vehicular communication to regulate traffic [499]. In [117], an early effort is made to specify the requirements of AIM. It may be considered a subset of Intelligent Transportation Systems (ITS), which strive to enhance the system by suggesting that individuals about traffic conditions and making mobility coordination safer and more intelligent [419].
- Decision-making and trajectory planning: MVS are commonly assumed to be primarily concerned with perception, planning and control activities. The planning layer firstly calculates the optimal global route using data from the global map and world information [124]. The planning layer then determines a local ideal trajectory through Decision-making and trajectory planning based on the vehicle's present perception messages [170]. Trajectory planning is frequently employed in vehicle collision avoidance [447] and robotic motion planning [268] to avoid conflicting paths.
- Multi-agent system (MAS): MAS (also referred as self-organized system or MRS addressed in Section 1.1.1) in which agents communicate and coordinate their behaviors to solve problems difficult for an individual agent [206, 393]. Transportation is an ideal environment for applying an Agent-based model (ABM) since transportation systems are very dynamic and all associated stakeholders, such as road users and infrastructure, must cooperate with one another, either actively via negotiation or passively by simply obeying rules [85].
- Cooperative driving: in compared to the conventional urban transportation system, the driving plans of vehicles were also supposed to be adjustable in order to further increase driving efficiency [284]. A common strategy in this region is referred to as "cooperative driving". Cooperative driving as a concept was originally presented in the early 1990s [196], mostly as automated platooning. The phrase "cooperative driving" was also used in [283] to refer to establishing collision-free vehicle travel through unsignalized intersections with the use of inter-vehicle communications.

1.2.2/ DRIVING STRATEGIES AT AN INTERSECTION

Despite surveys on intersection management and driving strategies have been substantially examined in the literature, they have not been evaluated concurrently (from managelayer to control-layer) [485]: for example, [416, 468] reports on intersection monitoring and scheduling studies from the perspective of a component-based system. Additionally, [207, 536] contains some assessments of the decision-making process from the perspective of intelligent vehicles. In fact, the driving strategy may also be influenced by the existing traffic context's assumptions. To simplify the driving issue, for example, the related work (summarized in [396]) assumes that vehicles approaching the intersection are serviced First In First Out (FIFO), that only one vehicle is permitted in the intersection zone at a time, that no turns are permitted, and that vehicles pass the intersection at constant velocity, etc. Thus, the suggested strategies for the AIM system that incorporates MVS are classified into several main categories, including rule-based, optimization, hybrid and machine learning [345]. The authors in [85] provide an overview of significant strategies and solutions for cooperative intersection. Further, cooperative approaches are classified as time slots and space reservations, trajectory planning, and virtual traffic signals in [85]. It is proposed in [485] to categorize autonomous driving approaches into two categories: cooperative driving strategies and individual driving strategies. However, as stated in [396], numerous methods to intersection driving strategies are heavily weighted toward centralized and decentralized cooperation. In particular, AIM driving method may also adopt a hybrid approach. Thus, we divide intersection driving techniques into reactive strategies and cognitive strategies for a compromise scheme in this PhD thesis.

The concepts of reactive and cognitive control were created first for mobile robot navigation, and they are frequently utilized in the literature [8, 28, 65, 123]. The term reactive architecture refers to the practice of determining the robot's actions only based on current sensor data (e.g., stimuli-response robot behavior). On the other hand, cognitive architecture makes considerably more extensive use of data about the robot's surroundings and internal state to determine its behaviors. A summary and translation diagram provided in [8] serve to understand these two control principles, which are shown in Figure 1.12. As previously mentioned, the driving strategy used in intersection management can be similarly characterized in terms of reactive and cognitive, which is applied for MVS decision-making and trajectory planning. Clearly, no significant distinction exists between these two notions in Figure 1.12. It is worth mentioning that although the reactive method employed in AIM does not need the vehicle to be totally devoid of sophisticated perception or decision-making to respond online, it would be a pity to not utilize them in the traffic context. Its primary distinction from the cognitive strategy is that the intelligent vehicle does not make use of all available environmental information to select its trajectory across an intersection. A well-known fact is that a preset traffic coordinator may provide more real-time traffic information to individual cars, allowing for partial cooperative maneuvering by reactive strategy. A supervised intersection, on the other hand, may deploy a local supervisor to monitor or train MAS cooperative movements with all essential environmental information. The next two sections will discuss these two methods with more thorough methodologies and examples.

1.2.2.1/ REACTIVE STRATEGIES

This strategy has been used to determine MVS cooperative maneuvers mostly in decentralized architecture. Furthermore, reactive driving tries to respond in real time to local environmental information while maintaining the essential stability and dependability. We highlight the typical approaches as follows.

- Priority-based resolution: using two decision algorithms, the authors in [16] argued that an autonomous vehicle might make a decision regarding the most acceptable crossing schedule in order to avoid colliding with other humanly operated cars on the road. It is feasible to coordinate the vehicles crossing an intersection with information acquired from local sensors. In [497], the authors proposed a novel intersection crossing strategy based on V2V or V2I communications in which vehicles compete for the right to pass through the intersection.
- Decentralized Model Predict Control (MPC): the authors offer a decentralized



Figure 1.12: Relationship between reactive and cognitive control [8].

MPC solution for autonomous vehicle coordination at intersections in [316]. Each vehicle with a preset path is given a linear quadratic optimal controller in order to save energy while passing through an intersection smoothly. By using local information from other cars within the communication range, each vehicle can define its own constraints. Similarly, [251] details an MPC-based method.

- **Cooperative resource reservation:** cooperative resource reservation has been studied in the form of decentralized intersection management, which does not need infrastructure assistance. Previously in [347, 348], a distributed reservation strategy based on collision areas was presented, in which only one vehicle is permitted to remain in a conflict zone. A distributed reservation protocol was suggested in [464], and comparable works are also available in [32, 33].
- Cooperative game theory: a suggested interaction modeling based on game theory including multiple leaders and followers is created in [285]. It also takes into account common traffic regulations at uncontrolled intersections. In [517], the authors propose a driving model based on cooperative game theory. This model's characteristic functions are based mostly on the purpose of each vehicle's safety, speed and comfort. A cooperative decision-making framework for MVS is proposed utilizing a coalitional game theory to solve vehicle safety and efficiency challenges at multi-lane merging zones [190].
- Ant Colony System (ACS): the work given in [495] presents a system for evacuating automobiles as quickly as possible for each series of vehicle arrivals. Further, the research offers an AIM strategy based on an ACS and a discrete optimization technique to solve the control issue for a large number of cars and lanes in real-time. To investigate the viability of AIM, a prototype based on NXT robots is created.
- **Parallelizable algorithm:** a distributed and parallelizable algorithm is given in study [233] for solving the coordination issue at intersection using a method called

Augmented Lagrangian-based Alternating Direction Inexact Newton technique (AL-ADIN). In this case, each vehicle solves its own optimum control problem and shares information about arrival and departure timings at the intersection with its neighbors in order to prevent accidents.

1.2.2.2/ COGNITIVE STRATEGIES

Cognitive strategies make use of all available information about the environment in order to maximize global efficiency, and it is feasible to show the optimality of the selected decision or action using such a strategy. The most commonly used methodologies are given below.

- Reservation-based system: the reservation-based approach was initially presented in [116], in which automobiles request and get time slots from the intersection during which they may pass. The reservation-based multi-agent control strategy outperforms the traffic light policy in simulation and approaches the theoretical optimum. This technique was then expanded to significantly enhance some limitations [117], such as enabling turning maneuvers and ensuring the appropriate acceleration profile inside the intersection zone. Additionally, two enhancements are described that enable the system to regulate human-driven cars and prioritize emergency vehicles. Next, the authors in [210] continue developing a new reservation-based intelligent intersection method. Along with receiving and validating car reservation requests, the new algorithms prescribe a speed profile for vehicles to follow until they cross the intersection. Additionally, the new reservation algorithms use a dynamic hierarchical reservation protocol that enables distinct reservation requests to be assigned varying priority. The reserve mechanism has also been examined by [31, 107, 278, 532].
- Platoon-based MAS: the authors in [235] suggest a platoon-based MAS in which vehicle agents may tactically organize platoons via communication technologies. Only the platoon's Leader Vehicle Agent (LVA) interacts with the Intersection Management Agent (IMA) by transmitting the platoon's predicted Earliest Arrival Time (EAT) and Earliest Clearance Time (ECT) and getting reservation confirmation (either approval or rejection). All follower vehicle agents (FVAs) just communicate with LVA and follow its trajectory. In [299], the authors provide an evaluation of the possible mobility advantages of a platooning-based strategy at intersections.
- **Centralized optimization:** this approach is primarily concerned with the formulation of an optimization problem whose objectives are, for instance, the travel time [234, 548, 551] and/or the overlap of vehicles in the core area [271, 273]. In [81], the optimal speed profiles of two vehicles approaching an intersection under a specified safety restriction have been deduced using closed-form solutions. The study reported in [552] created a novel algorithm for optimizing vehicle movement at intersections inside a CACC framework. The proposed framework uses game theory to guarantee that no collisions occur while minimizing intersection delays.
- Receding Horizon Control (RHC): RHC, also known as MPC, is a general-purpose control approach that entails repeatedly solving a restricted optimization problem and choosing the control action based on forecasts of future costs, disturbances,

40

and restrictions over a changing time horizon [321]. Just like centralized optimization, RHC is more particularly employed for multi-objective optimization, in which the objective function is minimized across a sequence of equal-length time horizons. The RHC/MPC paradigm is commonly utilized in AIM, as well as several related studies [103, 106, 238, 239]. The authors in [238, 239] proposed a coordination strategy for autonomous vehicles' safe and speedy crossing of an unsignalized intersection by considering their states collectively in an MPC framework. The optimal vehicle trajectories are determined by avoiding cross-collision risks in the vicinity of the intersection.

- Centralized machine scheduling: in this strategy, a control center is allocated to minimize the evacuation time of incoming cars. The authors in [496] propose a novel scheduling model that allows vehicles to negotiate their arrival time using an intelligent embedded device at the intersection. Control's objective is to evacuate all cars as quickly as possible while taking into account the possibility of the appearance of emergency vehicles. Similarly, in [509], the control center is intended to optimize vehicle passing sequence. To address this issue, a Dynamic Programming (DP) method has been developed.
- Coordination space approach: [177] provided a mathematical paradigm based on path-velocity decomposition. The authors provided a coordination space method for coordinating intersection vehicles. The paths of each vehicle are first determined and fixed, and then the velocity is adjusted to allow for safe and efficient intersection passage. The authors claim that it is a constructive locally optimal algorithm. Additionally, the authors explored priority-based strategies in the suggested coordination space approach [178] and the presence of legacy vehicles [382].

1.2.3/ MAIN CHALLENGES UNDER URBAN CONDITIONS

Much work has been published in the literature about MVS driving strategy at intersections in urban environments. Although the suggested cooperative intersection motion planning for MVS has been shown to be efficient in several simulations and field testing, the high computing burden or deadlock issue still restrict real-time implementation. Additionally, given the uncertainty, how much can we enhance traffic efficiency using the suggested intersection management model? How we may maximize cooperative MVS navigation at a broader scale using optimization approaches with extra information outside the ego-vehicle. A potential solution to this problem would be to expand the AIM system to incorporate connected and interdependent transportation segments like intersection and merging lanes, as well as to allow for further network-level traffic monitoring and improvement. The following are the main challenges associated with uncertainty and scalability, especially in urban settings.

1.2.3.1/ FLEXIBLE AND RELIABLE NAVIGATION UNDER UNCERTAINTY

Uncertainty should be treated as an inherent aspect of AIM in order to provide flexible and reliable MVS cooperative navigation. It may arise from a variety of sources, including control, mechanical, perception and communication systems. Vehicles (e.g., traditional vehicles with human drivers in the loop) may not obey orders, as the reservation-based system does. Generally, control and human uncertainties are not taken into account unless the task is expressly designed to cope with uncertainties (such as works presented in [178, 382]). Additionally, some research attempts to develop methods for coping with imperfect data from control systems. For instance, [66] examined a Maximal Controlled Invariant Set that allows for any admissible disturbance. Similarly, the authors addressed the collision avoidance issue in [186]. The model and the state estimation technique explicitly account for sensor uncertainty and transmission delays. These countermeasures are utilized to create a more robust controller, which makes the system more practical. In other researchers' works [280, 281], probabilistic trajectory prediction is also used to account for road users' uncertainties for autonomous vehicles, however, they either emphasize individual driving strategies or continue to rely on interacting systems. A viable option for dealing with uncertainty is to use a fusion-based approach [485]. For instance, the centralized and the distributed driving technique may be coupled to provide a complementary set of capabilities.

1.2.3.2/ PARALLEL IMPLEMENTATION (SCALABILITY)

The majority of existing MVS driving strategy is concentrated on isolated intersections. A natural question is how to apply the suggested strategy simultaneously in an urban road network with multiple intersections, i.e., parallel implementation. The rule-based approaches (e.g., priority-based resolution in [16, 497]) were developed to enable real-time AIM with either autonomous or mixed traffic. Numerous rule-based approaches have been verified in the actual world due to their computational simplicity and explainable models [308]. However, the complexity of the rule-based technique rises dramatically as the model's objectives and constrains expand, and its performance may vary according to traffic circumstances [345]. In comparison, centralized optimization or machine scheduling using intersection controllers has been developed to handle single- or multi-objective problems under a variety of circumstances, ensuring optimal performance (with varying constrains). Due to the computational cost of centralized optimization techniques, only a few of the publicly known optimization-based methods can be used in real-time in parallel [19, 245, 336, 397]. Indeed, when the number of cars and roadways rises, ensuring the algorithm's real-time and reliability becomes a significant difficulty. Even in the situation of full autonomous diving. Intelligent cars may sometimes deadlock or collision in complex situations/locations. The former is often the result of an algorithm that is too conservative, while the latter is frequently the result of an algorithm that is too aggressive [485]. To avoid deadlock and save computational cost, some researchers use a hierarchical design for cooperative intersection management [372] (cf. Section 1.1.2.3). Thus, hierarchical architecture might be seen as a potential option for managing many intersections at the level of traffic management.

1.3/ CONCLUSION

This chapter discusses single-vehicle intelligence systems in comparison to multi-vehicle cooperation systems. The prevalent MVS control architecture is classified primarily into centralized and decentralized designs. Moreover, the circumstances in which MVS may be used in complicated environments and situations are discussed. The relevant works for both high-way entrance/exit ramps and intersections/roundabouts are summarized.

The mechanisms used by the intelligent vehicle's decision-making and trajectory planning at isolated intersections are discussed in depth. The reactive techniques for MVS that have been presented attempt to respond in real-time and need just local environmental information. In contrast, cognitive techniques use all available knowledge about the environment to choose the optimal maneuvers. The advantages and disadvantages of introduced systems for MVS navigation are examined in further detail with an emphasis on actual implementation. Finally, the major issues confronting cooperative motion planning in urban conditions for MVS are outlined, with an emphasis on uncertainty and scalability.

DECISION-MAKING FOR MULTI-VEHICLE NAVIGATION IN A TRANSPORTATION NETWORK

As summarized in the preceding chapter, the implementation of MVS coordination is highly dependent on the planning algorithm's response to traffic conditions. Additional research is necessary in this PhD thesis to concentrate on more challenging circumstances, such as transportation networks with multiple intersections and roundabouts. However, there is currently a lack of safe and effective MVS control systems capable of parallel application (i.e., scalability) of AIM, given the uncertainty associated with traffic (cf. section 1.2.3). Based on the previous chapter's concepts and ideas, an introduction of multiple vehicle navigation in an urban context (e.g., in intersection network structure) is provided in this chapter 2. Additionally, we address the MVS system/tasks in such a complicated intersecting network. On the basis of the anticipated road network, many research paradigms and decision-making models are explored. Due to the uncertainty inherent in real-time traffic management, we emphasize the need of using a probabilistic approach and risk management while making MVS decisions on a transportation network. Finally, the conclusion addresses the present multiple intersection management system constraints as well as the fundamental restraints on the system's operation.

2.1/ INTRODUCTION

Many research works have been devoted to coordinate MVS on highways/arterial roads, as described in the preceding chapter (cf. section 1.1.3). Urban roads with intersections, on the other hand, are more complex landscapes. It has been discovered that 50% of urban crashes and 30% of rural crashes occur near intersections [118]. Additionally, traffic congestion is a typical problem in metropolitan areas, resulting in significant economic and environmental costs, as documented by various countries. For example, drivers contribute an additional \$121 billion loss and 56 billion pounds of CO2 in the U.S. [408] owing to urban traffic congestion. Intersections have been identified as the primary source of urban congestion. Advanced vehicular communication and coordination technologies are expected to result in a more intelligent traffic system. Thus, coordinating strategy is frequently discussed to improve many aspects of the transportation system (e.g., travel time reduction and CO2 emission reduction) within an urban network.

Coordination of groups of vehicles of varied sizes is one of the challenges in MVS trajectory planning and decision-making. In this respect, these applications blur the line between traffic management and MVS cooperation. Essentially, it addresses comparable issues at the road network level rather than at the vehicle level. Further, Urban Traffic Control (UTC) systems have also been created to accommodate the growing demand for traffic. MVS, being one of the most developed approaches, are regarded to offer significant promise for upgrading the present ITS [314], as indicated in the preceding chapter. For concerned readers, [141, 284, 362] provides an overview of traffic control, and [41, 183, 284] discusses the link between traffic control and vehicle connectivity. Indeed, the technology for UTC, ITS, MVS and IVs often overlap. Whereas the majority of published research in recent decades has focused exclusively on ITS for transportation or IVs for vehicle control, individual vehicle characteristics are nearly never addressed, particularly in traditional traffic signal management systems [183]. The integration of MVS with traffic management systems in the urban context introduces new issues that should be properly studied. Here, we primarily offer two questions to guide the PhD thesis through two challenges:

- How can the traffic system be defined and modeled based on knowledge about individual vehicle movements?
- What alternative to passive feedback may be used to regulate the traffic system by MVS-based traffic control?

Before addressing these issues in further details, we shall outline the scenarios, requirements, and associated systems, tasks, for multi-vehicle road-network navigation. This is done in the following section.

2.1.1/ OVERVIEW OF A ROAD NETWORK: SCENARIOS AND REQUIREMENTS

A road network can be visualized as a directed, weighted graph in which each node has a transport demand [441]. Urban road networks have a significant impact on a country's economic growth and are frequently congested, resulting in increased travel times, inconvenience for drivers/passengers, and increased air pollution [439]. It is critical to propose a systematic and sustained extension of urban networks, as well as proper maintenance, in order to ensure high-quality connections between the various areas of a geographical territory [341]. The addition of road infrastructure (e.g., I2V devices) and MVS enhances the UTC system's effectiveness and efficiency. Three forms of MVS-based navigation scenarios exist in this thesis: single intersection control, coordinated multi-intersection control of a traffic corridor and network traffic control. Section 1.2 discusses isolated intersection control. Thus, this section provides an overview of MVS navigation in traffic corridors and networks.

Numerous two-dimensional arterial corridors and urban road networks may be established in the literature using real-world data. To investigate the Cooperative Eco-Driving (CED) system based on MVS, researchers in [483] assume the presence of V2V/V2Ienabled cars traveling in a corridor with two traditional four-way intersections. The simulation is based on real-world flow data for the six-mile (9.66-kilometer) University Avenue corridor in Riverside, California (CA), USA. As seen in Figure 2.1a, two signalized intersections (i.e., University & Cranford, University & Iowa in Figure 2.1a) with defined lengths



(a) Road segments in Riverside, CA, USA: a corridor with two crossroads [483].



(c) Heterogeneous large-scale networks [6].



(b) Modeled urban road networks [247].



(d) Intersection network for simulation [446].

Figure 2.1: Urban arterial corridors and road networks in the literature.

are created in the simulation scenario. Additionally, the simulation network's inputs (e.g., signal timing and traffic count data for a certain period) can be calibrated. Thus, the traffic mobility of the proposed CED system can be tested in a partially MVS environment by using microscopic traffic simulation software. Moreover, the research [247] depicts the Chania urban road network's Central Business District (CBD) in simulation scenarios, as seen in Figure 2.1b. Within the red boundary of Figure 2.1b, a protected network (PN) is established. The eight intersections shown by a black arrow are classified as gates that enable vehicles to enter. Additionally, numerous origins and destinations (O-D denotes in/out traffic flows) are inserted at network borders or positioned inside the internal network encapsulated by a circle. Thus, by altering the O-D flow rate, one may imitate real-world traffic flow in such a 2D road network. The key traffic control challenges also include tackling the management of heterogeneous large-scale networks [6, 213, 306]. A general way to solve this issue is to divide a heterogeneous network into homogeneous sections, assuming distinct control strategies for the network's unevenly distributed vehicles. In [6], the authors analyze a case of heterogeneous urban transportation networks. As seen in Figure 2.1c, a 6.47-square-kilometer region of Downtown San Francisco has 100 intersections and 400 links (120 - 390 meters) with multiple lanes. To implement the suggested perimeter and boundary control [6], the test site is partitioned into three homogeneous reservoirs (highlighted in yellow, red, and green), clustered by small variances in link densities [229] shown by blue lines (see Figure 2.1c). Additionally, specific road networks are developed in the literature, particularly with the need to validate intelligent transportation technology. The authors in [446] present a method for optimizing the flow of a traffic network that avoids having vehicles stop at intersections, which was developed as part of the InTraDE European project [122]. Figure 2.1d illustrates the general road that operates in [446] with several intersections. 6 roads and 12 intersections comprise the traffic network. The intersection range is around 125 meters in radius, while the distance between two neighboring intersections is approximately 450 meters. The synchronized traffic lights may be utilized in such a network to generate "green waves".

Next, it is vital to evaluate the coordination requirements for the future generation of intelligent transportation network management systems that enable each traffic participant to perform better. In fact, the current urban traffic system is a complex multi-agent system governed by elaborate traffic laws and protocols that ensure the behaviors of many conventional road users. Traffic lights and stop signs used in such a system are actually meant to promote safe driving for human-driven cars, leaving a small margin for more accurate management by IVs or MVS [381]. To guarantee the implementation of MVS-based traffic control, one of the most significant challenges is establishing feasible agreements or explicit requirements for a variety of on-board equipment and road presetting devices. The following summarizes the safety, efficiency, and information security requirements discussed in the literature [117, 183, 455] for MVS navigation in urban corridors or road networks.

- Reliable communication (traffic safety): the regularly updated traffic situation is crucial for MVS to ensure a safe trajectory. For road safety applications, there are rigorous requirements for both bounded delay and high reliability [455]. Current standards permit a reliability of 90% - 99% and a latency of up to 100ms, but autonomous driving needs an ultra-high reliability of 99.999% and a ultra-low latency of up to 3ms [529]. Indeed, delayed and/or missing vehicle status (e.g., position mistakes) may result in an inability to forecast or warning of an appropriate action (for lane changes, emergency vehicles approaching, stationary vehicles and road conditions, etc.) [329, 455]. The navigation of MVS built on a centralized architecture is much more dependent on advanced communication technologies for the acquisition of new traffic status messages. Low communication complexity, on the other hand, might increase the system's scalability. Thus, apart from reliable measurement devices, robust planning [521] and stochastic programming [448] reported in the publication may contribute to the relaxation of the strict communication constraints. Additionally, recent research indicates that MVS may be better served by long-distance communication technologies [183].
- Protocol standardization (traffic efficiency): wireless communication applications are utilized to improve traffic efficiency (e.g., regulating road flow and minimizing emissions). An AIM protocol mechanism is early discussed in [117], which emphasizes that each agent needs a simple, standardized protocol for communication. IVs using standardized protocols are not required to comprehend the internal workings of the others. Additionally, new agents may simply be added to the system. Further, there are significant research and standardization initiatives underway as a result of the development of MVS-based technologies in the past two decades [455]. The communication protocol standardization began in 1999 when the U.S. Federal Communications Commission (FCC) authorized a 75-MHz band at 5.9 GHz especially for ITS [30]. In November 2004, the IEEE task group p was formed, which extended protocol stack to Wireless Access in Vehicular Environments (WAVE) including multiple layers. In 2010, the 802.11p vehicular "profile" (also denoted as Dedicated Short-Range Communication, DSRC) was certified [455]. Other protocol standardization efforts, such as ISO's CALM/CALM-non-IP and the European Commission's Mandata M/453, are also being developed prior to the 2010s [140]. Furthermore, Cellular Vehicle-to-Everything (C-V2X or LTE-V), which enables direct communication between connected vehicles using the PC5 interface, is considered

an alternative to IEEE 802.11p [171]. Since 2017, 3GPP has included LTE support for V2X services in Release 14 [529]. A special communication mode (Mode 3 and Mode 4) are dedicated to V2V communication. Indeed, IEEE 802.11p and C-V2X have been claimed to be included in automakers unveiled cars. Further development of new standards may allow the commercialization of fully developed MVS.

• Vehicle cybersecurity (information security): as the on-board information technology of the vehicle becomes more integrated into our daily lives, the resulting information systems evolve into automotive Human Machine Interface (HMI) devices that enable passengers, drivers, and even other road users to interact with the targeted vehicle in a far more natural manner. As a result, ego-vehicles develop increased vulnerabilities and possible attacks, which can be mitigated by the vehicle's anti-cybercrime system. As an autonomous agent, each vehicle may have privacy concerns that must be respected. For MVS cooperative navigation, cybersecurity is especially more critical in defending car systems and components from malicious attacks, illegal access, and damage while on the road [352]. The authors in [133] examine the vulnerability of traffic control systems in a connected environment, with a particular emphasis on transmitting erroneous data to cause maximum system delay. According to trial data, cybersecurity threats have a significant influence on critical intersections in transportation network. Additionally, [87] examined the vulnerability of Connected Vehicles (CVs) in the transportation system with a simulated attacks targeted at causing traffic congestion. The authors caution that the existing CVs signal management system is very vulnerable to data spoofing attacks conducted even by a single attack vehicle. Clearly, vehicle cybersecurity will be in high demand in the near future.

One might add other desired properties or requirements to a specific UTC system, such as deadlock or starvation avoidance; for example, vehicles at intersections could reserve a scheme to handle the stuck situation or stop low-priority vehicles from waiting too long [117]. Additionally, the suggested traffic management system should possess the capability of continuous deployment in chosen circumstances (e.g., add more intersections in a transportation network). Finally, while both safety and efficiency have been extensively studied in the past for advanced urban traffic systems, it is believed that they can be balanced using a variety of criteria (e.g., robust versus efficient, deliberative versus reactive, centralized versus distributed [381], etc.) in specific situations. This issue cannot be fully explored in this PhD thesis. In the following sections, we will further explore these criteria by combining them with other decision-making models (cf. section 2.2.2).

2.1.2/ MULTI-VEHICLE NAVIGATION MANAGEMENT SYSTEMS/BEHAVIORS

Traditional UTC in terms of traffic flow management is implemented by direct control measures (e.g., ramp metering in Figure 2.2a, traffic lights, dynamic route information panels in Figure 2.2b, and speed limits) and indirect road sensor-based operations (e.g., traffic control centers in Figure 2.2c, variable message signs, and highway advisory radio) to help maintain the congestion-free network [41, 374]. Indeed, traffic flow is now dependent on traffic signal-based management systems in large-scale metropolitan networks [247]. The critical issue remaining in conventional traffic management for multi-vehicle navigation is how to optimize network flow. This entails optimizing the phases (offsets) of the various intersections using a possibly high-dimensional variable space (for instance, as many



(a) An application of ramp metering [425].

(b) Dynamic route information (c) Traffic operations center [337]. [125].

Figure 2.2: Traditional traffic direct/indirect control.

as the network's nodes/intersections) [446]. This problem has also been investigated in the fields of operations research and queuing theory, concerning the human driver. With the emergence of IVs, studies are currently concentrating on techniques of autonomous and real-time management. Additionally, the term "Intelligent Transportation Systems (ITS)" or "Intelligent Vehicle Highway Systems (IVHS)" has been used to refer to the new generation of traffic management system [135, 433]. As previously stated, IVs/MVS are critical components of ITS/IVHS since they are capable of sensing the surroundings using on-board sensors (e.g., radar, lidar scanner, and computer vision devices) and controlling the vehicle autonomously or semi-autonomously supported by V2V/V2X communication [41]. Therefore, the term "MVS-based traffic control" will be used throughout this thesis to refer to an intelligent traffic system that utilizes connected vehicle technologies and is distinguished from conventional UTC by its signal-based direct/indirect approaches.

Similarly to ITS/IVHS, this chapter discusses MVS-based traffic control systems in road networks (see Figure 2.3). These systems incorporate traffic management, driver information and vehicle control. Naturally, the addressed system transfers the driver's tasks (steering, braking, and decision-making) from manual operation to MVS cooperation. Consequently, MVS enable real-time control decisions through the use of an appropri-



Figure 2.3: An evolution of MVS-based UTC systems (motivated by [266]).

ate computer algorithm, thereby overcoming or mitigating the negative effects of human driver limitations (e.g., slow reaction time, limited information processing capability), heterogeneity (e.g., different reactions among human drivers), and selfishness (e.g., non-cooperativeness) [383]. An example related to MVS-based traffic management issues is achieving a global optimum in traffic flow (i.e., homogeneous flow, maximum throughput, jam avoidance) during a certain traffic peak period [394]. In other words, traffic flow management may be transformed from a reactive and non-cooperative paradigm to a proactive and cooperative one based on MVS [183, 284, 383]. We provide more clarification on the V2I/I2V forms of MVS-based UTC systems by referencing the Automated Highway Systems (AHS) discussed in [41] and *Smart Roads Classification* Proposed by PIARC [150] (as seen in Figure 2.3):

- **Multi-vehicle cooperative systems:** without requiring V2I/I2V assistance, MVS coordinate their maneuvers with other vehicles using sensors and wireless communication mechanisms. Some conventional infrastructure may provide static digital data that is accessible to MVS, but it is not updated in real time.
- V2I/I2V-assist systems: MVS interact with one another, and roadside infrastructure provides guidelines for the vehicle's decision-making. There is digital information accessible regarding the road segment (such as HD maps). Dynamic information (such as traffic signs, weather, etc.) is updated regularly.
- V2I/I2V-manage systems: Vehicles indicate intended actions like as lane changes, exits, and entries intersections. Following that, the roadside system sends instructions/rules for inter-vehicle coordination of these maneuvers. Additionally, the road section enables cooperative perception: vehicles may communicate infrastructure about microscopic traffic and road conditions through V2I.
- V2I/I2V-control system: the roadside infrastructure takes complete management of vehicle operations, analyzes traffic, and optimizes vehicle operations, etc. In particular, the road section allows cooperative driving by perception information and providing commands to vehicles, hence improving safety and traffic operation.

In fact, there is a need for IVs to behave cooperatively as a form of social-AI [304] or social benefit [480] in urban environments. Although conventional research has provided car-following models, intelligent driver models, and cellular automata models for developing the vehicular cooperation mechanism from a physical perspective, these models mostly focus on straight line movement [444]. Thus, numerous cooperative behaviors for more complex urban scenarios have been seen in studies [304, 394] using the MVSbased UTC system described above. For example, the dynamic traffic flow is discussed in [304]. As seen in Figure 2.4a, traffic might be heavier in one direction than the other on a highway with equal lane width. When traffic is heavy in one direction, MVS system might alert other vehicles that there are now more lanes in one direction and fewer in the other. Consequentially, the MVS-based UTC could re-balance the traffic lane. Figure 2.4b shows more flexible collective vehicle behavior. The concept of a "Danger map" is also described in [394]. MVS might communicate braking data in a specific region. The risky driving road portion is highlighted in red in the published brake pressure diagram (see Figure 2.4c). Furthermore, multiple lanes merging is a common occurrence in urban areas. It is natural to predict that the negotiation/adaptation system between a group of vehicles could adopt optimum cooperative behaviors in order to maximize throughput



Figure 2.4: Numerous cooperative behaviors by MVS-based UTC system.

or traffic flow (see Figure 2.4d). Thus, the PhD thesis mainly focuses on the collective behaviors in this situation, which we'll discuss more below.

The existing MVS-based control algorithm is mainly concerned with single intersection (cf. section 1.2). Cooperative driving at neighbored intersections has also been studied by researchers [483, 510, 535]. However, multi-vehicle navigation has gotten less attention for urban traffic management with network-wide traffic control issues [183, 284]. The primary tasks or difficulties that distinguish the control technique for isolated intersections from corridors or network-level control can be divided into three major issues: *coordination of multiple intersections* using MVS, the associated *increase in traffic efficiency* in road networks, and *accurate traffic prediction/estimation* techniques.

Coordination of multiple intersections: the extension of the aim from isolated to multiple intersections coordination has been studied using both centralized and distributed approaches. The optimization goals for centralized techniques may be defined by aggregating the objectives of all the intersections or by defining a common target such as throughput (equivalent to the travel time of vehicles) within a radius of a single intersection. The authors in [240] suggested a multi-objective (e.g., desired velocity, fuel economy, and safe headway) centralized model predictive control (MPC) system. A control system called the "host" processes system information (such as previous vehicles, road gradients, and traffic signals) to determine the optimal vehicle control inputs. To reduce the computational load, the distributed technique is utilized to model the optimization targets by receiving infor-

mation only from neighboring intersections. The authors of [535] offer a decentralized optimum control framework based on MVS for two neighboring intersections. Individual vehicles could provide the optimal acceleration/deceleration (for minimizing fuel consumption) to pass through an intersection without crowding the connecting route between two coordinated intersections. As is the case with the MVS control architecture (see section 1.1.2), the distributed technique may be ineffective in achieving the global traffic control optimization. Additionally, the coordination (or prediction) of traffic speeds for urban corridors with multiple intersections is a prominent issue. Because drivers like to be aware of the expected speed of their route through multiple intersections when they are on the road [518]. To estimate the corridor speed between multiple intersections, the historical (and real-time) traffic information-based model [520], the extended time-series-based method [334], and the hybrid model [203] have been presented. Authors in [118] developed a consensus-based coordination technique for two unsignalized intersections in order to improve MVS navigation at the macroscopic level using the estimated corridor speed. In real-world, the work in [112] describes a field implementation based on store-and-forward modeling that includes two coordinated intersections in Chania, Greece.

Increase in traffic efficiency: one of the most critical aspects of increasing traffic efficiency is dealing with traffic oscillations. Traffic oscillation is a term that refers to the stop-and-go driving situations that occur in heavy traffic and often result in bottlenecks in transportation networks [545]. For instance, the trajectory data from NGSIM (traffic microscopic dataset [461]) is studied in [545] using distance-time diagrams: vehicles traveling between 4h00pm and 4h15pm suffer minor oscillations (Figure 2.5b), while vehicles moving between 5h00pm and 5h15pm experience significant traffic oscillations (Figure 2.5a). To dampen/eliminate traffic oscillations, it has been suggested to build a sufficient time buffer [429], minimize the reaction time [100], or use a trajectory planning model [310]. Roughly, these model-based approaches aim to optimize the movement of MVS through the use of communication technology by acquiring accurate vehicle status. Further, the aggregate status of vehicles may be synthesized into traffic criteria to represent the current situation,



Figure 2.5: An illustration of traffic oscillations [545]: between 5:00 and 5:15 p.m. (left image), vehicles experience greater oscillations than between 4:00 and 4:15 p.m. (right image).

hence assisting macroscopic traffic management. In this understanding, traffic flow and density, vehicle travel time, and aggregated speed are the primary indicators that can be measured to evaluate the effectiveness of UTC. Additionally, queue length is also a frequently used metric for assessing traffic efficiency. MVS-based traffic control systems are capable of calculating the development of queues exactly in real time [39]. However, since the majority of research has focused on off-line traffic disturbances [383], modeling MVS to increase travel efficiency is expected to be conducted within more real-world settings in the future.

• Accurate traffic prediction and estimation: Estimating the traffic state is a critical component of a traffic control system. Traffic estimation or forecasting is the process of assessing the flow, speed, and density of urban traffic in order to deduce feasible traffic patterns [538]. Additionally, using this predicted data, traffic trends on the road can be forecasted. It's worth mentioning that traffic forecasting is intrinsically spatial and temporal dependent in nature [538]. Changes in traffic volume will have mutual effects (i.e., the transfer effect and feedback effect [114]) on traffic upstream and downstream roadways in different geographical network structures (see Figure 2.6a). From the view of temporal dependence, the traffic volume changes periodically over a day or a week as shown in Figure 2.6b. In particular, key traffic information can even have a direct impact on the entire control approach's performance. Without reliable vehicle position information, neither isolated intersections nor network-scale urban traffic management can implement the suggested method. For instance, the queue length approach [501] or intelligent signalized intersections [91] used to establish a congestion-free road network need previous knowledge of traffic flow in order to infer or forecast the requisite flow parameters (e.g., flow rate, densities, aggregated speed, etc.). Rather than relying heavily on parsimonious control rules to develop a traffic feedback control system [284], the traffic estimating approach must be well-designed for a feedforward (or hybrid feedback/feedforward) traffic control system. Existing traffic estimation/forecasting strategies may be categorized into two types: model-driven and data-driven approaches. Readers interested in a model-driven approach may consult [486, 507], while those interested in a data-driven approach could reference [413, 463]. Cooperative prediction using MVS is a potential method for traffic prediction, hence increasing ego-planning utility [124]. For real world application, Telecom Italia has launched a real-time monitoring



Figure 2.6: The spatial and temporal dependence in traffic forecasting: a strong influence between adjacent roads in (a); the traffic volume changes periodically within one week or one day in (b) [538].

system for Rome that enables traffic estimation [74].

To summarize, as the number of connected vehicles is growing, it is vital to consider adapting the road service level in order to ensure autonomous driving properly. A road segment in the urban transportation network may be more or less ready for vehicles to deploy automation. Thus, it is assumed that the roads of the intersecting network mentioned in this PhD thesis are able to support the V2I/I2V-assistant or V2I/I2V-mange systems that are expected to arise in the short- or mid-term. Also, it is expected to result in the development of a safer, more comfortable, and more efficient multiple vehicle navigation system in this thesis. In the next section, we will go into further details about the research on MVS network-wide navigation with multiple intersections.

2.2/ Dynamic multi-vehicle navigation in intersection networks

This section discusses dynamic multi-vehicle navigation and how it may be used to control traffic flow in intersection networks. Although traffic management and control on highways/principle arterial are not the primary focus of this thesis. Certain critical applications and control systems, including the CACC, may continue to function inside the planned intersection networks (e.g., automated platoon in urban arterial roads). It is worth noting that the following MVS-based UTC are mainly referred to as intersections management in traffic networks. The main research paradigms and implementation on this subject will be explored in more detail, both with and without a transitional traffic signal system (cf. section 2.2.1). Section 2.2.2 focuses on the decision-making models, including various traffic management layers. Finally, we provide a vision for MVN with a probabilistic method (cf. section 2.2.3) and risk management (cf. section 2.2.4).

2.2.1/ MAIN RESEARCH PARADIGMS AND IMPLEMENTATION

Traffic lights have been used to regulate vehicle movement for more than a century¹. In an early review given in [364], road traffic management systems with traffic lights were classified into two general categories: fixed-time strategies and traffic-responsive strategies. The fixed-time strategies are generated off-line using historical data. While traffic-responsive strategies take into account current traffic circumstances. Coordinated strategies (including fixed-time or traffic-responsive control) are particularly effective when applied across a whole urban network. The most applicable work within these categories includes: MAXBAND (fixed-time) [301], TRANSYT (fixed-time) [399], SCOOT (traffic-responsive) [213], and OPAC (traffic-responsive) [152], etc. With the development of IVs, several surveys [183, 284] have addressed MVS-based urban traffic signal management systems in traffic networks. These papers discussed the situation for signal control systems optimizing Signal Phase and Time (SPaT) using Connected Vehicles (CVs) data and the Signal-Vehicle Coupled Control (SVCC) system. More specifically, using mathematical optimization paradigms, the authors in [183] classify three signal-based manage scheme as follows: *actuated traffic signal control, platoon-based traffic signal control* and

¹On August 5, 1914, in Cleveland, America, the world's first electric traffic signal was deployed [488].

planning-based traffic signal control. Additionally, *reinforcement learning (RL)-based traffic signal control* is also a promising technique for controlling traffic lights in complicated metropolitan traffic networks [94]. The following are some notable works in these categories:

- Actuated/adaptive traffic signal control: actuated traffic signal control, as well as adaptive traffic signal control², have the ability to adjust the SPaT depending on historical traffic patterns [530]. The authors in [174] propose an adaptive traffic light system based on wireless communication between vehicles and fixed controller nodes placed at intersections. Despite the effectiveness of traffic signal timing optimization programs, most current research seldom considers coordinated actuation signal management based on V2X communication. A stochastic-optimization approach for coordinated actuated traffic signal systems is described in [525], however its data was gathered from real-world collected by government and fixed test locations.
- Platoon-based traffic signal control: the platoon-based signal control tries to schedule signal timing plans such that platoons may pass through crossings without being interrupted. Following the launch of V2X communication, it has become a popular issue [194, 232, 291]. A platoon-based self-scheduling strategy was presented in [501] to addressing the traffic signal management issue in a road network. Each intersection, in particular, is managed by a self-interested agent with a defined horizon of incoming vehicles into clusters. Indeed, platoon-based signal control is capable of aggregating vehicles in a queue, which is equivalent to handling mid-level traffic flows. The calculation burden is reduced, making it feasible to implement. However, it is difficult to quickly, efficiently, and robustly identify the approaching platoon [155]. Further, aggregated patterns (i.e., anticipated queues and platoons) of traffic flow may have an impact on overall traffic flow management performance.
- Planning-based traffic signal control: planning-based methods take into account the information about each individual vehicle, which allows for more detailed traffic flow management. This technique has been extensively investigated in the literature [172, 272]. Planning-based approaches evaluate the optimal trajectory for an individual vehicle with a predicted horizon. By addressing a two-level optimization problem using CVs data, the suggested approach in [132] optimizes the phase sequence and duration. Nonetheless, when dealing with large-scale networks with the prediction horizon, the computing demands of the planning-based solution becomes very high. A system for optimizing intersections and corridors that consists of two layers of optimization is described in [43]. At the intersection level, Dynamic Programming (DP) is used to change the optimum green time in consideration of coordination requirements. At the corridor level, a Mixed-Integer Linear Program (MILP) is designed to optimize offset along the corridor. The simulation indicates the model has the potential to reduce average delay and stop for coordinated routes as well as the entire network.
- Reinforcement Learning (RL)-based traffic signal control: the Markov Decision Process (MDP) framework is used to formulate RL, which is an alternative method

²Actuated/adaptive traffic signal control uses inductive loop detectors that are installed tens of meters upstream of the stop lines for retrieving vehicle information [302, 364].

for adaptive traffic signal management [434]. Unlike model-based techniques, RL fits its parametric model directly in order to learn the optimal control based on its historical interacting data. The most frequently utilized approach for training a centralized RL agent is Deep Neural Networks (DNNs) [338], which can improve the scalability of RL. Numerous RL algorithms (e.g., the deep Q-learning approach) are offered to control traffic lights in various network configurations [149, 462]. In [514], a modified Proximal Policy Optimization (PPO) algorithm is proposed to be applied in a traffic scheme including multiple intersections and dozens of vehicles. However, in reality, centralized state processing will result in excessive latency, a high failure probability, and an exponentially increasing joint action space. Therefore, adaptive traffic signal management is formulated in [94] as a cooperative Multi-agent RL (MARL) issue, in which each intersection is managed by a local RL agent with restricted communication. Its objective is to develop a completely scalable and decentralized MARL algorithm. Nevertheless, the difficulty in MARL has evolved into a trade-off between optimality and scalability [182]. Thus, researchers establish Independent Q-Learning (IQL) [438] and the Independent Advantage Actor Critic (IA2C) [94] as local agent-based solutions to the partially observed dilemma. In the simulated traffic environments, these MARL algorithms proved their scalable and robust performance in intelligent signalized control systems.

Although traffic lights are still widely used in the real world and are thought to improve overall traffic efficiency in MVS-based urban traffic signal management systems, "signal-free" control systems are preferred to reduce stop-and-go events (which can result in increased fuel emissions [142]) by allowing vehicles to pass through the intersection without fully stopping. In fact, more proactive behavior may be incorporated in the scheme of intersection management. For example, acceleration behavior is effective in coordinating certain traffic flows but cannot be governed by a traffic signal system. Additionally, it should be highlighted that the majority of developed signal-free navigation methods (cf. section 1.2) are limited to isolated intersections. Early in 2006, a greedy search strategy was proposed in [165] for network-wide scheduling problem without a traffic signal. Later in 2015, authors in [548] attempted to establish a connection between autonomous intersection management and dynamic network assignment by using a network-level linear programming control. Further, numerous significant studies on the management of unsignalized intersecting networks include the following:

• AIM-based traffic control: the previously described tile-based reservation intersection control (cf. section 1.2.2), which reserves conflict-free trajectory for an AIM system, is expanded to control a network of linked intersections in [191, 277]. Indeed, AIM-based traffic control is naturally implemented with multiple intersections by using the typical multi-agent system (driver agent, intersection manager) like [116, 117]. The traffic light is actually replaced by an infrastructure (i.e., an intersection manager), who then has the authority to approve or deny a vehicle's traversal request. By taking into account the interactions of intersection managers through a network, AIM-based traffic control aims to offer new, fine-grained, and dynamic regulation for individual vehicles, even allowing for minute-by-minute reaction to traffic circumstances [191]. Additional study conducted in [191, 277] is aimed to investigate the potential of incorporating dynamic traffic assignment (macroscopic level) into the proposed AIM-based intersection control system (microscopic level). Further, [210] developed a simulation test bed (with a network simulator) for evaluating the section of the proposed astimulation test bed (with a network simulator) for evaluating the section of the proposed astimulation test bed (with a network simulator).

AIM-based traffic management. The primary distinctions between different AIMbased traffic control methods are the controlling mechanism or navigation policies [498]. Another study in [466] conducts an empirical evaluation of several policies for managing a reservation-based intersection. The majority of AIM-based research work can provide successful simulation results. However, the coordination of multiple intersection managers has not been thoroughly investigated.

- Vehicle negotiation system: in the field of single-intersection navigation, [347, 348] presented a decentralized solution which can be categorized as vehicle negotiation systems. In the system, the intersection manager is removed. The agents of the vehicles interact with one another to decide the order of passage and departure from the intersection. Specifically, vehicles in the collective negotiating system are intended to solve an optimization problem using sensed vehicle data (location, speed, and inter-distance, etc.). The negotiated results may include information on when and how fast to merge [394]. However, the technique is limited by the number of vehicles attempting to negotiate. Thus, in a negotiation-based approach (sometimes called auction-based approach [405]), just a few vehicles near the intersection are considered to make decisions within a certain time interval. Negotiation-based control often performs less well than AIM-based control, but the vehicle negotiation system in the valuation-aware mechanism [405] has a lower overhead for making decisions and generating agile responses to any unexpected changes [284]. Similar to the negotiating system, [273] proposes a cooperative vehicle intersection control (CVIC) algorithm that operates without the need of typical traffic signals and is expanded to cover a corridor with several intersections. Another study [467] proposes a distributed market-inspired approach for managing an urban road network. In particular, the strategy derives from the reservation-based intersection control (same as AIM-based traffic control). However, the adaptive management system enables MVS to bid on an intersection crossing in accordance with the auction-based traffic control strategy. Additionally, the competitive traffic assignment approach is intended to ensure a free-flowing movement for the desired MVS. Because of unresolved liability concerns and inter-vehicle communication challenges, evaluating negotiation-based techniques in actual traffic conditions is difficult.
- Other work: [293] investigates the MVS routing issue using an iterative A* algorithm in a 9 × 9 grid of intersections. A novel routing algorithm is described that minimizes average delay and optimizes vehicle throughput in a network of linked intersections. The technique is capable of forecasting future traffic flows. The authors in [480] have proposed a cooperative autonomous traffic organization strategy for MVS operating on multi-intersection road networks. Notably, a three-tier decision-making framework is provided, which includes an autonomous crossing strategy at intersections, improved road segment optimization, and a composite routing strategy. In general, the traffic management system based on MVS acquire fine-grained information in this work. In comparison to traditional flow management policies, autonomous intersection control may benefit more from a combination of status information (position, velocity, route, etc.) and flow information (from the macroscopic level). However, a large computation demand may be necessary, which must be considered in the envisioned system.

To the best of our knowledge, real-world validation of MVS-based UTC approaches has just recently begun (see Figure 2.7). Although certain Advanced Driver Assistance Sys-



 (a) Multiple intelligent traffic lights are deployed at the intersections of Jinan, China [209].

(b) Mcity at the University of Michigan, United States of America [454].

(c) Intelligent vehicle testing base in Changshu, China [457].

Figure 2.7: A realistic environment for deploying MVS urban navigation technology.

tems (ADAS) include a variety of real-world applications [18, 270], research on MVSbased traffic management systems has been restricted. The authors in [542] provided a comprehensive platform for forecasting traffic demand and optimizing traffic signals in Jinan, China, using vehicle trajectory data from a commercial company (Figure 2.7a), therefore enhancing traffic signal performance and lowering delay by up to 20%. Meanwhile, reports of MVS-based test track have been made in Mcity-University of Michigan, United States of America [454] (cf. Figure 2.7b) and Changshu, China [282] (cf. Figure 2.7c). With the abovementioned research paradigms and implementation for MVSbased UTC, this latter will concentrate on the decision-making model in the context of various traffic management layers.

2.2.2/ DECISION-MAKING MODELING

As indicated before, traditional attempts have been made to adopt adaptive, proactive control mechanisms to enhance traffic management using real-time communication technology. However, the decision-making method, which involves building a system of rules and deducing the most suitable maneuver, cannot be accomplished by merely connecting individual intersection controllers without regard to their relationships [191]. Agent-based traffic management, in particular, has gained popularity over the past two decades as a multi-layer problem solving pattern [94, 116, 117, 168, 234, 353]. Like the majority of the hierarchical research community, in the context of our research it is adopted a bottom-up approach to describe an agent-based traffic management system in three layers, as seen in Figure 2.8. A more precise definition of the three-level agent-based traffic management architecture is as follows:

- *Macro-level:* a global supervisor (or network management agent) is in charge of the whole transportation network and is capable of recognizing real-time traffic demands.
- *Management-level:* a local supervisor (or Intersection Manager agent) is in charge of the particular intersection and is responsible for regulating local traffic changes.
- *Micro-level:* an intelligent vehicle (or driving agent) may adjust its behavior on an individual level via microscopic control or by contacting a local supervisor for navigation regulations.



Figure 2.8: A bird's view of the proposed agent-based traffic management scheme in a bottom-up and hierarchical system.

As illustrated in Figure 2.8, each level provides a platform distinguished by different communications paradigms. At the micro-level, the IVs are expected to ensure sustainable, low-latency cooperative movements under the supervision of a local intersection manager, which performs flexible mechanism for managing traffic resources according to macro-level policies. The majority of proposed hierarchical agent-based management systems in the literature aim to achieve a balance of local (intersection) and global (network) optimums [118, 391]. As summarized in [226]:

"These layers are expected to make IoAV (Internet of Autonomous Vehicles) the admixture of both durability and interoperability, diversity and profundity, and most importantly, reliability and flexibility."

Particularly, the purpose of this PhD thesis is to investigate decision-making and trajectory planning (or short for "decision-making" cf. section 1.2.1) for MVS. In other words, we focus on the intelligent vehicle (or driving agent)-based decision-making system for the micro-level, where the control objective is accomplished by independently altering the behavior of each vehicle to satisfy the demands of the macro-level or managementlevel. Though a fully automated transportation network is unlikely in the near future, it is worthwhile to investigate the microscopic decision-making technology embedded in the promising hierarchical architecture in order to enable novel efficient vehicle navigation that benefits the entire road network. In the following, we will discuss the decision-making problem in further detail at different levels.

2.2.2.1/ DECISION-MAKING: MICROSCOPIC MODELING AND CONTROL

The majority of the techniques to intersection navigation presented in section 1.2 are microscopic control strategy in nature. These models, on the other hand, are regarded isolated inside their immediate surroundings. In contrast, the microscopic decision-making model of MVS mentioned in this section is designed to be adaptive to the different types of traffic networks. More precisely, *microscopic modeling* techniques (such as cellular automaton models) can provide fundamental principles for individual vehicles and their network activity. One can distinguish these main approaches as follows:
- Cellular automaton model: a one-dimensional array L or a directed graph G is used to define this computational model [165, 246, 344]. A site in L or a vertex $v \in \mathcal{V}$ can be used to represent every cell. Each cell may be filled (by a single-vehicle) or unoccupied in the setup, implying the assignment of time slots. As a result, the cellular automaton model is built into a Discrete-Time and Discrete-Space system. Additionally, researchers in the literature create cellular automaton rules for system updates, which are carried out in parallel for all cars. The authors in [344] make the assumption that each vehicle travels at an integer velocity along a boundary. Four rules are used to update the proposed cellular system: vehicle-free acceleration in safe situations, slowing down for ahead vehicles on site, speed randomization to account for natural velocity fluctuation, and motion planning based on current speed. An interesting "start-stop-waves" was observed by the integer-valued probabilistic cellular automation rules. Additionally, the authors in [165] consider the vehicle's length and the minimum safe distance as the cell length, which is related to time slots. Notably, the intersection conflict points are not regarded as cell points, whereas the cell are directly linked between the neighboring two lanes at the border, as seen in Figure 2.9a. In such a road network, the system state with n vehicles is donated by $X(t) = [x_1(t), x_2(t), \dots, x_n(t)]$ where vertex $x_i(t) \in \mathcal{V}$ representing vehicle *i* at time t. Thus, the essential proposition is to calculate the new state X(t+1) using the present state X(t), and the feasible shift subset is defined as one that satisfies all constraint conditions. In the later part of the study, a feasible system clearing schedule solution is presented in the model, emphasizing the existence of such schedules. The feasible scheduling issue, on the other hand, remains NP-hard. In the cellular automaton model, the heuristic algorithm is commonly used.
- **Trajectory optimization model:** another method for model microscopic trajectory planning is to construct an optimization control system based on the kinematic function. Taking [480] as a reference, the kinematic model is as follows:

$$\dot{\boldsymbol{x}} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & 0 & 0 \end{bmatrix} \begin{bmatrix} x \\ v \\ a \end{bmatrix} + \begin{bmatrix} 0 \\ 0 \\ 1 \end{bmatrix} u$$
(2.1)

where $x = \{x, v, a\}$ are respectively the displacement, speed, and acceleration, *u* is input jerk. Additionally, an objective function *J* is defined with regard to journey



Figure 2.9: An illustration of microscopic models.

time, energy consumption, and other comfortable characteristics during the planning phase, which is subject to kinematic constraints:

$$min \ J = f(u)$$
Subject to: $\dot{x} = Ax + Bu$
(2.2)

One can find the similar research in [480, 535]. Additionally, [480] employs the optimization control model to optimize the road segment's trajectory between two intersections. As seen in Figure 2.9b, a pieced motion process with three phases is introduced: deceleration, stop, and acceleration. Thus, when a vehicle departs the former intersection (at t^0) and enters the latter (at t^f), a smooth trajectory with minimal jerks and acceleration may be ensured. Another work [522] provides a Mixed-integer Linear Programming (MILP) model for cooperatively optimizing vehicular trajectories over a traffic corridor with unsignalized intersections in a completely MVS environment. In the micro-level, the vehicle model is simplified to a first-order dynamic model while maintaining the linear and convex properties of the proposed optimization model. The vehicles' trajectories in the corridor are coordinated in a unified framework to achieve system-optimal in terms of overall vehicle delay. The centralized planning framework, in particular, has the potential to deliver safe and effective car following and lane change behaviors for each vehicle driving through unsignalized intersections. However, a significant computational problem exists.

2.2.2.2/ DECISION-MAKING: MACROSCOPIC STRATEGY

Once the microscopic modeling is assigned, the aggregated vehicle's performance can be measured to value the overall traffic efficiency. On the other hand, decision-making strategies based on *macroscopic strategy* have resulted in the development of traffic flow models with input characteristics such as throughput, velocity, and density [351, 426]. Indeed, early decision-making strategies for traffic flow such as SCOOT[®] (centralized) [213], SCATS (centralized) [306] and Utopia (decentralized) [322] in the UTC system remain applicable to large-scale networks. They are, however, inefficient in light of rising traffic demands, resulting in a frequently crowded traffic situation [247]. On the other hand, an optimization-based method (e.g., RHODES [152]) proposes certain sophisticated traffic responsive techniques, albeit at the cost of a variety of computing resources. Thus, interdisciplinary solutions are provided in these publications, such as genetic algorithms [77], fuzzy logic [168] and artificial immune networks [104]. This section will outline many of the most important decision-making processes from a macro view perspective, like the following.

Fundamental diagram (FD)-based strategy: a fundamental diagram (FD) in the shape of a flow-density curve is employed in highways and two-dimensional urban road networks [247]. The concept of FD has been presented since 1969 [167]. It may be thought of as one of the Aggregated Traffic Relationships (ATR) between average traffic variables (e.g., speed, density, or flow) for a particular network [160]. The form of FD has been established by researchers via simulation-based tests [153], theoretical analysis (using a utilization-based approach) [198], and systematic empirical research [158]. A coordinated signal-based application described in [112] for real urban network has also been reported to have a fundamental diagram-like traffic flow. In addition, the notion of a fundamental diagram is also called

a Macroscopic Fundamental Diagram (MFD) or a Network Fundamental Diagram (NFD) [247]. In fact, MFD/NFD is determined by origin-destination demand, which is very stable when traffic loads are distributed uniformly across networks [160]. Generally, the paradigm for an MFD is based on the assumption that congestion is distributed evenly [158, 160]. Nonetheless, heterogeneity in road density contributes significantly to the shape (i.e., unimodal curve) and scatter level (i.e., low scatter relationship) of MFD/NFD. The influence of heterogeneity on MFD/NFD has been investigated experimentally using real-world data from a medium-sized French city [69]. The authors in [391] offer a two-level aggregated model for the purpose of examining the dynamics of heterogeneity and its effect on the MFD/NFD model. To the best of our knowledge, microscopic decision-making based on MFD/NFD is very rare in research within the intersection networks. A recent publication [531] examined unsignalized intersection passing orders using various driving strategies (i.e., FIFO-based cooperative driving, Monte Carlo tree search based cooperative driving, and a fixed-time traffic signal strategy). The studies demonstrate that passing order (from microscopic decision-making) has a substantial effect on the efficiency of network traffic (Figure 2.10a), and that a better order may greatly increase the curve of the MFD/NFD. As seen in Figure 2.10b, the suggested Monte Carlo tree search (MCTS)-based cooperative driving method [506] outperforms the traditional traffic signal approach or the FIFO-based strategy as measured by the MFD/NFD curve. It implies that the MFD/NFD results might be more effectively used to validate and design the microscopic decision-making techniques.

Route planning-based strategy: route planning is studied in particular in the context of dynamic traffic assignment, which may be accomplished via the use of simulation-based methodologies [432] or through optimization-based approaches [188]. The emerging MVS technology enables vehicles to plan their routes cooperatively in order to take the most efficient trajectory over road networks. More specifically, the authors of [480] suggest a composite strategy for route planning that incorporates cooperative decision-making of MVS: vehicles can select the route to reduce the average delay time by cooperative decision-making before they get into the road networks. To address all MVS requests, an information center (similar to global supervisor) is deployed. Additionally, a greedy algorithm is utilized to op-



Figure 2.10: A comparison of various decision-making strategies in relation to the MFD/NFD curves.

timize the sequence for route planning, taking the expected maximum delay into account. From a macro perspective, vehicles that have a high probability of causing a traffic delay are given priority when routing. Another publication in [293] describes a routing mechanism based on an iterative A^* algorithm. Each iteration consists of three stages: batch processing to store all vehicle data, routing to execute time-based A^* searches consecutively, and congestion checking to encourage vehicles traveling through congested regions to take a detour. By forecasting future traffic flows, the suggested route planning strategy aims to decrease average delay and maximize vehicle throughput. The efficiency of iterative A^* is shown by comparison to time-based and distance-based A^* using the SUMO simulator. Route planning-based strategies, on the other hand, need centralized knowledge for coordination and V2I/I2V data exchange, when in practice, information biases are inevitable.

2.2.2.3/ DECISION-MAKING: HIERARCHICAL CONTROL

Clearly, the overall traffic conditions may have an effect on the behavior of individual vehicles. Low-level vehicle control, such as speed changes, gear changes, steering, and so on, could be translated from high-level reasoning to cooperative behavior [304]. In addition, it is worth mentioning that, as previously said, there is a choice between AIM-based and vehicle negotiation-based decision-making strategies (cf. section 2.2.1). It is still seen as an open issue between centralized and decentralized control. Additionally, a *hierarchical control* system based on management-level and micro-level decision-making techniques is feasible (e.g., synchronization-based intersection control). Consequently, the *hierarchical control* often contains both an intersection manager and a driving agent (see Figure 2.8). To demonstrate how the hierarchical decision-making model regulates the vehicle's behaviors, some common instances are as follows.

• Synchronization-based control: to allow vehicles from different directions passing through intersections without stopping. The authors in [445, 446] proposed a two-level decentralized multi-agent system to improve traffic mobility via a stop-free strategy. The intersections network is illustrated by Figure 2.1d. The network is then modeled as a graph (v, ε) (i.e., it is made up of a collection of labeled vertices ν and edges ε) with lane segments $i \rightarrow j$ where i and j are neighboring intersections (see Figure 2.11). The authors in [445] have proposed a local synchronization strategy for passing through the vehicle alternatively without stopping. Thus, each intersection is managed by a control agent (identical to the intersection manager), which communicates with the vehicle agent in order to manage their passing. However, the vehicle agent should limit their speed to ensure that they arrive at the crossing zone within the period T specified by the intersection control agent. T is derived as the minimal period required to pass two vehicles (one from each flow), which can be shared in the road network. Further, in the management-level, an intersection *i* directly interacts with its neighbors with its phase ϕ_i of periodic signal for synchronization-based control. On the other hand, at the microscopic level, an adaptation phase is used to achieve the desired crossing speed V_{crossing}. As seen in Figure 2.11, vehicles in such a decentralized system depart at a constant speed V_{MAX} from some intersection i. It will continue through the following intersection j at the velocity $V_{Crossing}$ specified by the intersection control agent *j*. Self adaptation phase (equivalent to motion planning) occurs in the ego-vehicle. As a result, the



Adaptation Phase

Figure 2.11: In the synchronization-based control model, any vehicle from *i* to *j* will pass through (1) the uncontrolled zone between *i* and *j*, (2) the adaption phase control zone inside *j*'s radius of *R*, and (3) the crossing zone of radius r_0 [446].

desired performance (to create "green waves") can be obtained by such a stop-free strategy.

 Consensus coordination: consensus coordination is used to balance traffic density throughout a transportation network. Numerous consensus algorithms are explored in the literature [118, 498, 499]. The concept of coordinated multiple intersections is based on a completely AIM system and the Greenshield's traffic model [189]. Consensus coordination is specifically applied in two or three layers via a hierarchical control technique [118, 499]. The infrastructure (AIM) network is the top layer. However, each AIM-like agent is expected to coordinate only local traffic information within its neighborhood. Thus, the paradigm of decentralized control is achieved. Additionally, a single autonomous intersection can be modeled in the lower layer. To handle a vehicle crossing an intersection, the centralized control approach is used. As a result, traffic information is considered between connected neighbourhoods. The main difference is the strategy of consensus coordinating for traffic flow. To ensure uncongested traffic flow, the traffic estimation technology is addressed in [499]. Additionally, it is critical that the network topology model and I2I/I2V communication protocol are well-designed. Simulation results demonstrate that consensus coordination is capable of coping with rising mobility while still requiring a sophisticated communication technology.

Table 2.1 provides the associated research on MVS decision-making in road networks that was highlighted before.

To summarize, with vehicular communication technology, there is a trend from passive feedback control to mixed feedforward/response control in urban traffic management [183, 284]. For example, when the MVS' current states are known or partially known, the future state of traffic flow with desired control actions may be anticipated by recursively solving their dynamical differential equations [284]. Generally, distributed and hierarchical strategies may be attractive network control methods for decomposing large-scale optimization issues into manageable sub-optimization problems. In fact, the sub-problem may be handled more effectively by collective information sharing [14]. Remarkably, the majority of strategies for decomposing centralized network-level management into distributed control are mathematical programming or heuristics [14, 161, 324, 328, 495, 495]. Thus, ensuring the global optimum remains a difficulty for distributed control, so does the decomposition process's stability. Following that, hierarchical techniques make decisions at both the macro (higher) and micro (lower) levels. The main barrier is defining precise macroscopic and microscopic models. At a higher level, traffic flow models are often described as MFD/NFD including concise traffic data that encapsulate the es-

Approach	Main techniques Objectives		References	
Cellular automaton	stochastic discrete au- tomaton	simulate traffic flow	[246, 344]	
	heuristic	traffic scheduling	[165]	
Trajectory optimization	optimal control	safety, fuel consump- tion, comfort	[480, 535]	
	Mixed-integer Linear Program (MILP)	delay	[522]	
	theoretical research	form verification	[153, 167, 198]	
FD-based strategy	empirical research	FD existence	[69, 112, 158, 160]	
	aggregated model	heterogeneity	[69, 391]	
	gating control	protected network	[247]	
	Monte Carlo Tree passing ord		[531]	
Pouting based strategy	greedy algorithm	routing sequence	[480]	
Houting based strategy	iterative A*	delay and vehicle throughput	[293]	
Synchronization-based control	Hill-Climbing iteration	green waves	[445, 446]	
Consensus coordination	Greenshield model	traffic density and de- lay	[498, 499]	
	Fast Model Predictive Control (F-MPC)	traffic velocity	[118]	

Table 2.1: Summary of decision-making model for MVS navigation in road networks

sential feature of network-wide vehicle flow and densities (i.e., the collective behaviors of MVS) [101, 102, 158, 159, 516]. MFD-based network control [159, 184] may outline the complete upper-level network issue and direct the lower-level (for particular regions/intersections) to generate optimization targets. However, the topic of how to merge MFD-based network control with precise intersection/corridor management for MVS remains unanswered. Another approach to resolving challenges associated with heavy traffic management is to implement a cooperative driving strategy that involves the formulation of short-term driving plans via bilateral or multilateral negotiations (i.e., Vehicle negotiation system) [16, 35, 273, 518]. Noting that this technique considers just a few vehicles passing through the intersection within a certain time period. It is seen as a lower layer of control than planning-based control in certain publication [284]. In the meantime, the inherent trade off between the control scheme quality and its computational demands is therefore a crucial issue that should be explored thoughtfully.

2.2.3/ PROBABILISTIC APPROACH FOR DECISION-MAKING

Clearly, distributed/decentralized control architecture is favored in the suggested intersecting networks with moderate computation requirement. Further, probabilistic technique is widely employed for a robust decision-making strategy. For instance, the Probability Collectives (PC) algorithm is an efficient optimization searching framework for distributed systems, which was first proposed by Wolpert and Bieniawski in [58, 494]. It is a COllective INtelligence (COIN) framework that emerged from game theory, statistical physics, and optimization theory [262, 513]. A comparative study has shown that the PCbased approach is superior to traditional Genetic Algorithm (GA) in both rate of decent and avoiding local minima [208]. Kulkarni et al. in [262] designed a shrink-sampling interval method to improve the algorithm performances via benchmark functions. After that, a PC-based approach successfully solved various discrete optimization problem like Multiple Traveling Salesmen Problem (MTSP) and Vehicle Routing Problems (VRP) [261, 262]. To handle MVS/MVS intersection navigation, the PC algorithm in [373] (presented by our team) has two important qualities, namely probabilistic nature and decentralized nature: its probabilistic nature allows a probability distribution over a vehicle behavior set, guaranteeing a risk averse decision strategy. Additionally, the probabilistic method enables the management of uncertainty without causing the shared decision process to deadlock. Further, its decentralized nature allows it to be used without a specific road infrastructure. Besides, vehicles can significantly benefit from an acceptable computing time (around $0.2s \sim 0.8s$ in the full optimization cycle). Thus, it is an interesting and promising method to process the aforementioned Multi-Vehicle Navigation (MVN) problem in restricted areas. Although the formulation of PC in [373] is intended for an isolated intersection, it offers a compelling potential to adapt the probabilistic method (for microscopic-level management) to more complex traffic situations/environments.

2.2.4/ RISK MANAGEMENT FOR MULTI-VEHICLE NAVIGATION

As indicated in section 2.2.2.2, route planning is explored in particular in the context of dynamic traffic assignment. However, when it comes to dynamic route assignment, safety is seldom addressed at the traffic management level. In addition, the majority of MVSbased traffic management techniques are optimized for throughput, travel duration and fuel consumption (see Table 2.1), etc. Clearly, risk management is critical in traffic control, and it can be expressed as a group of goals that increase overall control performance [504, 540]. The primary problem for risk management in the road network is determining the appropriate quantitative and qualitative methods for assessing driving risk, as well as how to establish constrains for various purposes. It is worth noting that one of the PhD thesis aims is to route and schedule vehicles on the road network (macro-level) in a manner that strikes a balance between speed and risk. In micro-/management level control, safe and smooth autonomous navigation technology have been widely considered in the intelligent Advanced Driver Assistance Systems (ADASs) [346]. Lane Keeping Assistance (LKA) and Adaptive Cruise Control (ACC), for instance, are effective tools for obstacle avoidance driving in single vehicle control [346]. However, in the view of multivehicle cooperative navigation, the road accidents are more likely to be regarded as the failures of the multi-agent system rather than failures of any single vehicle [333]. The historical data approach is used to identify particular traffic accidents and apply safety countermeasures [305]. Due to the sparse nature of traffic accidents, the use of such an approach is limited to perform safety analyses based on proper accident database records [27]. A more qualified form of risk management method is proximal safety indicator, which occurs more frequently for safety assessment and requires a short time for data collection [27]. Furthermore, the generally used proximal safety indicators are time measured metrics with a form of Time-To-X (e.g., Time-To-Accident, Time-To-Collision, Time-To-Break, etc.) [202, 205, 484]. Indeed, safety indicators provide an active approach assessing traffic conflicts to road users with reliable results. But these safe concerns lack of consistent definition or a robust theoretical foundation [92]. Among those methods, Time-To-Collision (TTC) is usually viewed as a more objective tool for predicting traffic accidents [27], [45]. A TTC-based traffic event can be always recorded during the entire interactive process. Controllers can decide whether to adopt evasive maneuver in advance (with regard to the intention and purpose) rather than emergency braking at the last resort [215]. The threshold for TTC is generally a definition that implies the riskmargin for drivers to react in a possible accident [92]. Arguably, TTC-concept is widely used as an important part of traffic conflict technique. However, detecting critical traffic events with TTC in varied spatial structures is more challenging. As a result, continued development of a TTC-like indicator as a risk-sensitive road proximate safety indicator is required.

2.3/ CONCLUSION

This chapter examines Multi-Vehicle Navigation (MVN) on road networks and the associated decision-making models. Numerous vehicle network navigation scenarios and requirements are presented in an urban setting (e.g., in the intersection network layout). Furthermore, we discussed the MVS systems/behaviors in such a complex intersecting network. Multiple research methodologies and decision-making models are examined in further depth using the anticipated road network, both with and without a traffic light system. We extend our investigation of the primary decision-making model/strategy and control mechanism in the context of several traffic management layers. Additionally, this chapter emphasizes the current limits on the operation of the multiple intersection management system, as well as the basic constraints on the system's operation. Finally, due to the inherent uncertainty in real-time traffic management, we highlight the need of using a probabilistic approach and managing risk while using MVS in a transportation network.

COOPERATIVE NAVIGATION SCHEME IN COMPLEX ENVIRONMENT/SITUATIONS

SAFE AND FLEXIBLE COOPERATIVE NAVIGATION WITH RISK ASSESSMENT

Cooperative Navigation (CN) is a widespread technique to ensure the efficient navigation of intelligent vehicles in complex/cluttered environments. As discussed in section 1.1.3, coordination issues that arise in rural and urban locations include motorway/highway speed coordination, merging at on-ramps, and coordination at signalized/autonomous intersection. Nevertheless, the main challenge exists in estimating and managing in-road risks while applying flexible CN strategies (cf. section 1.2.3). Thus, this PhD thesis is concerned with designing the navigation strategies used by MVS in signal-free intersections that provide reasonable communication conditions. Remarkably, the decentralized architecture has been chosen for determining the cooperative maneuvers of vehicles. It is intended to address a manner of bilateral or multilateral negotiation in our formulated system. Meanwhile, the suggested method takes into account the inherent trade-off between the quality of the control system and its computational requirements.

More precisely, this chapter firstly outlines a flexible CN scheme for MVS system to deal with a cooperative system (cf. section 3.1). With its relative low execution time (as indicated in section 2.2.3), the Probability Collectives (PC) algorithm has succeeded at generating fast and feasible solutions to cross intersections and roundabouts [373]. Moreover, IVs can run PC in a decentralized manner. However, the PC is still sensitive to uncertainty in the navigation process, which highlights the need to adopt several safety margins. This chapter focuses on balancing between the high-quality cooperative optimization and acceptable computational speed. Thus, a reliable risk management strategy is proposed by introducing a novel ε -constraint PC method (cf. section 3.2). A real-time communication mechanism is suggested for a distributed system to avoid invalid behavior due to inconsistency. The novel ε -PC based navigation strategy allows the vehicles to adapt their dynamics and react to unexpected events while respecting real-time constraints. The typical common-yet-difficult circumstances in a single unsignalized intersection appear to verify our findings: the ε -PC method can ensure collision-free behaviors and reserve reasonable reaction time for vehicles' safety insurance.

Furthermore, this chapter explores risk management issues beyond local cooperation and try to formulate an Shortest Path under Risk constraint (SPR) problem from the perspective of a global supervisor (cf. section 3.3). The idea here is to limit risk while allowing the vehicle to move as quickly as feasible. In contrast to the local CN approach, A discrete decision model is proposed that incorporates a heuristic local search algorithm for rapid run in routing strategies. Finally, the statistical instance analysis provides a thorough

understanding of our algorithm.

3.1/ PROBLEM STATEMENT

Similar to urban transportation systems, specific territories like large hospitals, university campus, commercial and industrial sites have taken steps to improve their navigation services in their shipment/transit areas [216]. MVS system in such restricted areas may help to provide more efficient transport services for passengers [8, 99, 216]. In the meantime, numerous simultaneous requests from multiple delivery locations may invoke cross-linked planning routes for MVS system. However, the inherent trade off between the control scheme quality and its computational demands is therefore a crucial issue that should be explored for this kind of cooperative navigation at intersection points.

In this chapter, a typical graph of two connected IVs cooperative navigation at an intersection is illustrated in Figure. 3.1. Similar to Multi-Robot Systems (MRS) (cf. section 1.1.1), multiple levels of coordination between the different agents take place depending on the overall navigation system feed-backs.

Main assumptions: MVS are provided with an enhanced autonomy. They may manage the assignment of the navigation tasks by themselves through embedded decisional devices and inter-vehicle communication tools. Details about other important autonomous vehicle navigation technical issues, such as cooperative perception and localization, planning and re-planning, control architectures may be found in [8]. Here in Figure. 3.1, two connected IVs are which exchanging their predicted future motion trajectories. Because IVs can better understand the behavior of each other, we consider IVs more likely to have prosocial (or altruistic) behaviors rather than too conservative (or egoistic) behaviors [410]. Thus, the two IVs may perform a collaborative search of coordinated actions based on a utility-maximizing decision model. In presence of a non-collaborative agent (but this agent broadcast its estimated behaviors at first), then the IVs can only achieve a sub-goal of the navigation system by optimizing its own behavior. Let suppose that a pri-



Figure 3.1: IVs system with action probability distribution to predict their behaviors.

ory anticipatory probability set is already specified to predict a potential action of the other vehicle. Nonetheless, the probability distribution of actions need to be updated while the IVs perform their collaborative searching or local optimization process. In that respect, the data processed is stored and shared as files in a distributed system. Accordingly, a distributed approach is applied in a natural way to find out coordinated actions of the MVS. Therefore, this chapter aims to validate a decentralized approach to handle this distributed multi-agent optimization problem.

3.1.1/ RELATED WORK AND MAIN CHALLENGES

In the field of intersection coordination, the direct vehicle control approach has been applied to change the traffic lights pattern [319, 435]. Slots assignment to vehicles [212] is also a popular technique which is used in the same context. However, traditional traffic signal control methods in urban city usually cannot be applied directly in above mentioned areas, because traffic light is subjected to redundant cost in such an inappropriate formed crossing-road and in certain situations increase the level of traffic jam [183]. Automation and communication have turned the cooperative intersection management into a more active research field [85]. Roughly, distributed and decentralized control are becoming a promising way to deal efficiently with this multi-scale navigation problem in complex traffic scenarios. Studies reported in [85, 179] provide more details about such cases. Additionally, a non-signal management of vehicles from a shared space is studied in [373]. A distributed and decentralized optimization algorithm, based on Probability Collectives (PC) [262] (cf. section 2.2.3), is applied to solve vehicle cooperation problem.

For a group of homogeneous MVS, researchers have tended to focus on efficient and effective controls to cut off with customer waiting time or energy consumption [54]. The existing MVS dispatch study rarely discussed how to simultaneously maintain the optimal performance and also avoid risks at intersections. As a matter of fact, risk minimization has been shown to be considered as a priority in such a case [85]. Since sudden changes in the dynamics of ground vehicles in a short time are not realistic [83], [217]. There is still considerable ambiguity with regard to a risk assessment approach for safe and flexible navigation of MVS.

To summarize, vehicles' collaboration with risk management capabilities is a promising way to solve above mentioned problem. Additionally, the consideration of the real-time concerns and the management of several simultaneous actions are of utmost importance for such a distributed navigation system. Thereby, distributed real-time cooperative systems with a safety constraint (collision avoidance) can be generated in our case. Based on the PC theory, the previous research in [260] has investigated several off-line PC optimizations with soft constraints (e.g., tension/compression spring design problem). Thus, next section focuses on the analysis of real-time MVS intersection coordination by integrating the PC theory.

3.1.2/ PROBABILITY COLLECTIVES (PC) ALGORITHM

In order that the proposed method can be simply read, let us sum-up in what follows the already proposed PC formulation to deal with the coordination of MVS in intersections and roundabouts [373].

3.1.2.1/ FORMULATION OF SEARCHING SPACE

Several vehicles are considered crossing through the intersection with fixed known path. Then, the only control degree of freedom of the MVS are the speed of navigation. PC treats the vehicles in a coordination problem as individual self-interested players iteratively [494]. Thus, these agents, in our case of study several vehicles (as shown in Figure. 3.2), should select their actions (velocities in our problem) over a particular predefined interval time to coordinate their navigation motions. An illustration of the possible actions in fixed time windows (T = 10s) which is long enough leaving an intersection as depicted in Figure. 3.2a.

Apparently, in Figure. 3.2a, there are considerable options N_i for each vehicle *i* depend-



(b) Uniform distribution of all the agents' behaviors.

Figure 3.2: Example of strategies hypotheses for vehicle actions.

ing on the initial speed $v_i(0)$. By both considering the safety and comfortable requests, intersection has a speed limit below 10m/s and vehicles tend to restrict acceleration in $[-2m/s^2, 2m/s^2]$ according to a restrained speed profile as illustrated in Figure. 3.2a. A further taken hypothesis is that all the vehicles will get a fixed speed $v_i(T)$ after a predefined action time t_{act} (such as $t_{act} = 3s$ in Figure. 3.2a). At last, the searching space for vehicle *i* can be summarized as a tuple $\mathbf{\Pi}^i$ i.e., $\mathbf{\Pi}^i \sim \{v_i(T), t_{act}, N_i\}, t \in [0, T]$. Then, the admissible member of actions set for the ego-vehicle can be presented as $\mathbf{\Pi}^i \in \mathbf{\Pi}^i = \{\mathbf{\Pi}^i_1, \ldots, \mathbf{\Pi}^i_{N_i}\}$. Here, $\mathbf{\Pi}^i$ can be visualized as the velocity profile in Figure. 3.2a.

As mentioned before, in the PC theory the expected utility of a given action can be calculated by each vehicle. But to do so, it must get (or estimate) the possible actions of the other vehicles. It has been used probability distribution to model relative actions like $q(\Pi_k^i) \in q(\Pi^i) = \{q(\Pi_1^i), \dots, q(\Pi_{N_i}^i)\}$. Obviously, the preferred actions (or strategy) have a high probability of being cost-effective. The driver model used to improve the precision of the predictive control with probability distribution is a hot topic, but not the main research topic in this section. Indeed, it is considered in the proposed work that this probability distribution is given by dedicated algorithm and according to that it is proposed an appropriate strategy to take the most appropriate decision making under this initial probability distribution. Readers are recommended to read [410] to get clearer idea about the estimation of the probability distribution of other ground vehicles. The hypotheses of prosocial (or altruistic) IVs in this section make us formulating a uniform distribution (as shown in Figure. 3.2b) of all the agents' behaviors when $q(\Pi^i)$ is initially loaded for computation.

3.1.2.2/ Two steps for re-acceleration

For various collaborative navigation behaviors, vehicles choose a speed profile that allows them to safely cross an intersection based on an utility function (see section 3.1.2.3). However, for the vehicles which have to choose the arbitrary low speed (or a complete stop), the proposed algorithm allows them to re-accelerate. The re-acceleration permits the vehicles to clear the intersection as fast as possible while ensuring free collisions. An important point that needs to be addressed is that re-acceleration should ensure continuity constraints of the speed. An algorithm that enables a continuous speed profile after the action time have been designed in the previous work [373].

3.1.2.3/ OBJECTIVE FUNCTION

In its initial shape, the original PC approach considers only an unconstrained minimization problem. Such a research case generally involves *n* vehicles, and each vehicle $i \in n$ possesses a strategies/actions set of $\Pi^i = {\Pi^i_1, ..., \Pi^i_N}(i = 1, ..., n)$ including an equal amount of *N* options (cf. section 3.1.2.1). After performing a local motion planning through their on-board embedded devices, each vehicle applies a strategy $\Pi^i_k \in \Pi^i(k = 1, ..., N)$ during time interval [0, T]. Here, *T* refers to the prediction time horizon. During the period [0, T], a particular set of combined strategies $Y = [\Pi^1_k, \Pi^2_k, ..., \Pi^n_k]$ is selected (randomly fixed to initialize the process) to reach at least a minimum system utility level $J([\Pi^1_k, \Pi^2_k, ..., \Pi^n_k])$. The proposed objective function [373] can be formulated as given in equation (3.1):

$$J(Y) = W_{sep} \sum_{i_v \neq i_{self}} \sum_{t_k=1}^{max} \frac{1}{d_k (i_v, i_{self})^2} + W_{cross} (v_{max} - v_{avg})^2$$
(3.1)

where $d_k(i_v, i_{self})$ is the distance between the ego vehicle i_{self} and the vehicle i_v (i.e., all collaborative vehicles) at time step t_k (a discretization of $t \in [0, T]$). v_{max} refers to the maximum speed legally allowed on the road. In addition, v_{avg} is the average recorded speed of all the vehicles during $t \in [0, T]$. W_{sep} and W_{cross} are the weights to balance between the different criterion characterizing (3.1): low separation and slow average speed. It should be noted that the proposed J(Y) value is updated iteratively during the PC algorithm execution by the agents taking part in the coordination process. Thus, the delicate designed searching space approach must ensure a sampling of "good" quality during the first action time. Readers are encouraged to read [262] and [373] for further information.

3.1.2.4/ A SUMMARY OF PC MODEL

As mentioned before, equation (3.1) is utilized without absolute safety constraints. For several cases, a very high weight W_{sep} may be admitted to penalize low separation distance to ensure more safe navigation. This can lead vehicles to preferably choose arbitrary low speeds (or a complete stop). Such behaviors may be regarded as very conservative. In real-time traffic navigation, IVs must have appropriate control architecture with reliable and real-time Risk Assessment and Management Strategies (RAMS). These targeted RAMS must reduce drastically the navigation risk in order to face sudden road hazards and risky situations. Unfortunately, the proposed previous work does not provide a fully nil risk of collision [373] and explicit risk-sensitive strategy. Further, PC running time is inconsistent depending on the number of collaborative involved entities. Theoretically, MVS should have a certain time interval to start executing self-satisfied strategy targeting lower navigation risk. Thus, this chapter main aims correspond to fill this gap and provide cycle-accurate description of these mechanisms in a systemic way. More flexible multi-criteria decision-aids techniques and time consistency in distributed systems will be further discussed in the following sections. More precisely, a method of limiting the time spent for optimizing is suggested at a common-yet-difficult scenario (cf. section 3.2.1), which can calculate the consistent action execution time before entering a conflict zone. Furthermore, in the latter case, a constrained PC algorithm is proposed (cf. section 3.2.2) to compute the corresponding multi-criteria risk management strategy to guarantee 100% collision-free navigation in an appropriate prediction horizon.

3.2/ The ε -PC for safe and flexible CN

3.2.1/ MVS FORMULATION

Our specific objective is to explore PC theory enabling flexible and safe coordination to improve service performance of MVS in restrained and complex areas. Therefore, we can cast our case in a customer pickup-and-delivery scenario. After routes are scheduled, MVS have to decide the actions at an intersection with an on-board autonomous control system as shown in Figure. 3.3. While considering the real life application, the PC method consists in planning motions. These motions (referenced as speed profiles) can be used to control a vehicle or to warn/assist the driver to avoid dangerous situations. Accordingly, a study of motion planning that satisfies real application of MVS will be formulated within optimal control in section 3.2.1.1. However, these planning is restricted by system intrinsic dynamic limitations and surrounding communication environments. As it will be clarified later, the Time-Slot-Based (TSB) communication approach is considered for better predicting how the vehicles will interact and collaborate with each others (cf. section 3.2.1.2). More precisely, based only on every vehicle own-observations, the "single mode" addresses each vehicle individual motion planning without any further cooperation with other road participants. Contrarily, the "full mode" manages the reactions between all existing vehicles while ensuring the motion planning task as seen in section 3.2.1.3.

3.2.1.1/ AN OVERVIEW OF MVS FRAMEWORK

Let suppose that our experimental scenario is as the following: vehicles track a desired path \vec{P} , while searching the most appropriate velocity. If the chassis of an actual car is defined in an x - y reference frame, we can denote the vehicle's position as (x, y). The driving routes are identified by a series of way-points $(x_i, y_i) \in \vec{P}_i$ as illustrated in Figure. 3.3. Here, $(x_0, y_0)_i$ correspond to the the initial position at the time when the computation time is launched for vehicle *i*, where $(x_f, y_f)_i$ is its final position. It comes that what we want to compute is the correspondence between time $t \in [0(\text{initial registered time}), T(\text{vehicle reaches its final position})]$. In Figure. 3.3, three positions of vehicles are indicated (at time t = 0) and the control is related to the speed $v_i(t)$ (that remains in an interval $[0, v_{max}]$). At any time *t*, the distance between any vehicle *i*, $j \in \{1, 2, 3\}$ cannot be less than a 2r threshold (in order to avoid the collision of the vehicles), where *r* is a safe radius for vehicle *i*, *j* as shown in Figure. 3.3.

Furthermore, in Figure. 3.3, the communication area is specified for Inter-Vehicle Communication (IVC). After loading the computing at t_0 , vehicles in communication area can exchange the state information and priory anticipatory probability of possible actions before entering the intersection. Due to real-time computing environment, it has been given a deadline for MVS returns the strategies/actions for critical applications at the intersec-



Figure 3.3: Application scenario and main zones characterizing the addressed MVS.

tion. Here, a negotiation area is defined w.r.t. action time t_{app} (data processing deadline) for synchronous cooperative navigation. Because vehicle's initial speed is different, the position of the vehicle begins to collaborate in negotiation area will be different w.r.t. t_{app} . The core area (red block in Figure. 3.3) is more critical zone which contains the possible collision.

It is important to notice that all the vehicles are provided in this setup with the same kind of control devices (or control protocol). Henceforth, vehicles will follow the same algorithm logic and share current states when loading computation at t_0 . As indicated before, we recommend setting a time limit for the solver, which ensures that the PC program will terminate in a reasonable period of time Δt_{sol} (cf. section 3.2.1.2). This motion planning can be applied at time $t_{app} = t_0 + \Delta t_{sol}$. Because vehicles in our system is rolling without slipping (i.e., Pfaffian constraints), we can accurately predict vehicle's states (i.e., position and speed) at t_{app} when loading the algorithm at t_0 . So, it is better to plan motions having the predicted horizon time that starts at $t = t_{app}$ for MVS ($[t_{app}, t_{app} + T]$), where *T* is the prediction horizon. To do this, we can always pursue an optimal solution that guarantee well-coordinated motions in time. It is worth noting that the vehicle will precisely execute its final desirable actions/strategies (as $\Pi_k^i \rightarrow v_i(t)$) during time interval [$t_{app}, t_{app} + T$]. Some additional constraints are highlighted below for applying motion planning Π_k^i at t_{app} :

- The vehicle that has already entered the intersection is not concerned by the optimization process.
- Vehicle *i* keeps constant speed $v_i(0)$ (and less than v_{max}) before executing Π_k^i .
- If vehicle *i* will enter the intersection immediately after t_{app} with current speed $v_i(0)$, then treat it as a "non-collaborative agent" with maximizing self-utility strategy.

Dynamic constraints (e.g., inertial delay in powertrain) and trajectory deviation are not considered in this model. As mentioned above, we designed the PC to run in an iterative way. When the on-board algorithm is launched, it produces a possible action plan based on states of the current agents in communication area with a prior knowledge of each other (cf. Fig 3.3).

The best combined strategies should be saved and updated if new feasible optimal solution is obtained by the MVS. To address the dynamic nature of the transportation system, users are permitted to change the action after adding any other agent in the MVS. This means that the previous plan will be executed with the possibility of few changes occurrence when new cooperative actions become available.

3.2.1.2/ TIME-SLOT-BASED (TSB) MECHANISM

In the above real life application of MVS, the critical issue is the computing of each plan "fast enough" during Δt_{sol} by PC, in order so that the system can react to the changing environment without exceeding its motion capability (defined as maneuverability [57]). In our proposed PC implementation, the data exchange is aimed to be minimal and a solving time of $\Delta t_{sol} = 0.2s$ is targeted ($\Delta t_{sol} = 0.8s$ have been achieved so far for 4 vehicles [373]). Moreover, a reasonable Δt_{sol} leaves the system a maneuverability margin that can be used to move it into a safe configuration or state. To address this concern, we use Time-Slot-Based (TSB) approach to further explain the vehicle's sequential optimization and communication mechanism in PC (see Figure. 3.4).



Figure 3.4: Time-Slot-Based (TSB) communication mechanism for applying PC.

In TSB vehicle communication system, the basic time interval Δt_{sol} is divided into multiple duration. Here, "class regions" are highlighted for different message classes. We can use different wireless bandwidth for these regions. In class A, vehicles transmit the status information to other connected agents and exchange possible actions for entering an intersection. After vehicles get priory anticipatory probability of other vehicles' behaviors, the navigation problem can be formulated as an optimization model and the PC will be run in its default iterations. All the vehicles successively participate in the optimization with on-board PC algorithm at each iteration. They broadcast the updated probability distribution over the set of possible strategies for repeated computing reference. The successive iterations continue until all the updated probability distributions converge to stable distributions. However, long-running can be time-consuming and difficult to optimize. The timeout mechanism in class C helps to limit that time while supplying a satisfactory motion planning. Our optimizer triggers timeout when:

- All vehicles converge to stable probability distributions.
- The algorithm's running time exceeds timeout limits.

The conduct of timeout setup may lead to complete or to incomplete search. We are aware of the fact that vehicles do not guarantee convergence to the global optimum in our standard solving time Δt_{sol} . But the PC algorithm always retains the current best results at each iteration. These strategies tend to produce high quality solutions in short time. The precise timeout limit value is changed depending on the running machines and experiment configuration but very close to the motion planning applied time t_{app} . It is interesting to note that the majority of our proposed PC models are either completed very quickly or they converge very slowly. Therefore, changing the timeout value (not too much) will not dramatically influence the computing results.

3.2.1.3/ MOTION PLANNING FOR CONTINUOUS VEHICLES CROSSING

To the best knowledge of the authors, the PC algorithm has not been used yet to perform repeated optimizations to deal with a continuous flow of vehicles. The closest application was [423] for air traffic management, but the algorithm was demonstrated on fixed initial situations. Since a repeated full optimization of MVS is time consuming, we already defined two real-time motion planning modes of PC for searching feasible solutions in practical application:

- Single vehicle optimization (denoted "single" mode).
- Full optimization ("full" mode).

As above explained, in the single mode, a vehicle runs the PC optimization as soon as it enters the communication area as shown in Figure. 3.3. All the other vehicles are considered to be connected but non-collaborative agents. In this mode, the coordination is sequential more for the collaborative case, since the vehicles decide what to do one by one. This is an important option to reduce calculation time, because ego vehicle only needs to pick up the best actions with respect to fixed strategies of others. Also, the intersection crossing performance with the single mode may be sub-optimal as not all the vehicles coordinate with each other.

In the full optimization mode, the PC algorithm will run in its default iterative mode where all the vehicles participate in an optimization process as shown in Figure. 3.4. We recommend to perform the "full" mode long enough (10s for instance) to ensure vehicle exiting the intersection, and it should be triggered by a predefined event (such as a threshold number of vehicles at intersection). New vehicles entering in the restricted communication area are not allowed to rerun the full optimization mode until the previous optimization is completed.

Later in this section, authors focus in section 3.2.4.4 to prove that the real-time solution of two modes can handle continue vehicles waves in practical real-time applications.

3.2.2/ RISK MEASUREMENT BY 2D TTC

Due to the probabilistic nature of the decision-making problem between vehicles, it is hard and not straightforward to directly convert the constraints to probability space. Therefore, several heuristic repair approaches are applied to narrow the optimal solution [261, 262]. The elevated computational load limits thus the use of the PC approach in hard real-time vehicle. Kulkarni and Tai et al., then handle the constraints by a penalty function method [263] while knowing that the appropriate weights parameters (between sub-criteria) are not easy to be obtained precisely. In the proposed approach, the existing ε -constraint method [323] is used in addition to the PC algorithm to solve the real time multi-criteria safety assignment MVS coordination problem. The navigation characteristics of MVS and main constraints are highlighted in section. 3.2.1.1. Then, in the following paragraph, we introduce Time-To-Collision (TTC) as a constraint indicator. The purpose of MVS is to compute a feasible solution, which serves all the customer in a flexible and risk-sensitive manner. The system objective function in equation (3.1) can offer a combined solution that penalizes low separation and slow average speed. However, as mentioned before, the previous work needs a risk assessment approach to succeed in the road hazard prediction. Thus, the TTC is used as a predictive safety measure of vehicle's trajectory.

TTC is a risk indicator that describes the remaining time for a probable collision (i.e., traffic crash) between two vehicles. It was originally defined by [193] in car following scenarios. Generally, TTC can measure a road-user's time to react (for a critical collision event). The TTC at time instant $t \in [0, T]$ can be calculated according to the first order case (in co-linear navigation case between vehicles) [46]:

$$TTC = \frac{x_{lead}(t) - x_i(t) - 2r}{\dot{x}_i(t) - \dot{x}_{lead}(t)}$$
(3.2)

where $(x_{lead}, y_{lead}), (x_i, y_i) \in \vec{P}, y_{lead} = y_i$, and 2r is the vehicle real length as shown in Figure. 3.5a. In equation (3.2), x_{lead} , if exist, can be measured as the position of leading vehicle for vehicle *i* at x_i with speed $\dot{x}_i(t) > \dot{x}_{lead}(t)$. To calculate TTC in two dimensions, we simply consider a collision of two circles as shown in Figure. 3.5b.

As one may notice that it is a "collision" of two circles (not a real crash of two vehicles). We use these circles to anticipate real accident. Here, 2r can be seen as vehicle length l as in equation (3.2). In spite of sacrificing some accuracy, the TTC between vehicles i, j can be more easily formulated in two dimensions as:

$$[(x_i(t) + \dot{x}_i(t) \cdot TTC_{ij}) - (x_j(t) + \dot{x}_j(t) \cdot TTC_{ij})]^2 + [(y_i(t) + \dot{y}_i(t) \cdot TTC_{ij}) - (y_j(t) + \dot{y}_j(t) \cdot TTC_{ij})]^2 = (2r)^2$$
(3.3)

In equation (3.3), setting $(x_i, y_i) \in \vec{P}_i, (x_j, y_j) \in \vec{P}_j$ is the position of vehicle *i*, *j* at time instant $t \in [0, T]$. $\dot{x}_i(t), \dot{x}_j(t), \dot{y}_j(t)$ denote the relative speeds measured in *x*, *y* directions.



Figure 3.5: A collision of two-vehicles based on circle area.

Accordingly, we can get a quadratic function of TTC_{ij} like:

$$\begin{aligned} & [(\dot{x}_{i}(t) - \dot{x}_{j}(t))^{2} + (\dot{y}_{i}(t) - \dot{y}_{j}(t))^{2}] \cdot TTC_{ij}^{2} \\ & + 2[(x_{i}(t) - x_{j}(t))(\dot{x}_{i}(t) - \dot{x}_{j}(t)) + (y_{i}(t) - y_{j}(t))(\dot{y}_{i}(t) - \dot{y}_{j}(t))] \cdot TTC_{ij} \\ & + [(x_{i}(t) - x_{j}(t))^{2} + (y_{i}(t) - y_{j}(t))^{2} - (2r)^{2}] = 0 \end{aligned}$$
(3.4)

Equation (3.4) can be solved by quadratic discriminant. If there are real roots in (3.4), we can take the positive lower value as the nearest TTC in the prediction horizon. For cases equation (3.4) has no real roots, it presents the collision that will never happen with this dynamic. To avoid any confusion, we defined the solutions in equation (3.4) as "2D TTC" in the following of this section. The objective of MVS is to maximize the final agents' 2D TTC to improve the navigation safety. Thus, the corresponding objective function is defined as:

$$max \qquad J_{TTC}(\mathbf{Y}) = \min_{\substack{i,j \in \{1,2,\dots,n\} (i \neq j)\}} \{TTC_{ij}(\mathbf{Y})\}$$

subject to
$$\mathbf{Y} = [\Pi_k^1, \Pi_k^2, \dots, \Pi_k^n](k = 1, \dots, N)$$

$$\Pi_k^i \in \mathbf{\Pi}^i = \{\Pi_1^i, \dots, \Pi_N^i\}(i = 1, \dots, n)$$
(3.5)

Where $\min\{TTC_{ij}(Y)\}$ represents the minimum 2D TTC value of the most critical situation between *n* agents in the prediction horizon $t \in [0, T]$ within vehicles' combined actions/strategies *Y*. J_{TTC} aims to maximize the critical 2D TTC value for more safety response to the concerned situation. Above all, an optimization problem can be formulated by considering equation (3.1) and (3.5). To handle the TTC constraint, ε -PC algorithm is addressed in next section (cf. section 3.2.3).

3.2.3/ RISK MANAGEMENT BASED ON ε -PC FRAMEWORK

The original PC algorithm focuses on a straightforward task with only one objective function as shown in equation (3.1). Nevertheless, the MVS needs to deal with RAMS as suggested by the discussion given in section 3.1.2.4. The ε -constraint method, which was firstly proposed in [187], can be introduced to handle this trade-off problem. Only one objective function is optimized in the method, while others are converted into constraints with a permitted value ε by a limited range. In our case, the objective function J_{TTC} in equation (3.5) can be adopted as a constraint during optimizing the main objective function *J* in equation (3.1). Hence, the transformed optimization problem is formulated as below:

$$\begin{array}{ll} \min & J(\mathbf{Y}) \\ subject \ to & \mathbf{Y} = [\Pi_k^1, \Pi_k^2, \dots, \Pi_k^n](k = 1, \dots, N) \\ & \Pi_k^i \in \mathbf{\Pi}^i = \{\Pi_1^i, \dots, \Pi_N^i\}(i = 1, \dots, n) \\ & J_{TTC}(\mathbf{Y}) \ge \varepsilon \end{array}$$

$$(3.6)$$

According to the model, the optimal results could be given by the following theorems. The interested readers may consult [330] for more details:

Theorem 1: If objective *J* and vector ε = (ε₁,..., ε_m) exist, such that *Y*^{*} is an optimal solution to the problem (3.6), then *Y*^{*} is a weakly Pareto optimal solution.

• *Theorem 2*: Y^* is a strict Pareto optimal solution if and only if, for objective *J*, there exists a vector $\varepsilon = (\varepsilon_1, \ldots, \varepsilon_m)$, such that Y^* is the unique objective vector corresponding to the optimal solution of the problem given in equation (3.6).

3.2.3.1/ A SELECTION OF ε VALUE

An advantage of the ε -constraint method, presented in equation (3.6), is that we do not need to scale different objective functions by adding weights. The obtained solution, if it exists in equation (3.6) with a given parameter $\varepsilon = (\varepsilon_1, \ldots, \varepsilon_m)$, is proved to a weakly Pareto optimal solution as Theorem 1 and 2. Actually, the Pareto front can be obtained by varying the vector ε . To find an efficient solution (that means close to a strict Pareto optimal solution) in problem (3.6), selecting an appropriate ε is the key. Accordingly, for calculating a more efficient solution, we must have at least the range of constraint objective function J_{TTC} . Unfortunately, the calculation of the J_{TTC} range in searching space is not a trivial task. The worst value is hard to compute, while we can get the best value in an individual optimization. Hence, a general selection of ε_m can be provided by equation (3.7):

$$J_{TTC}(\boldsymbol{Y}_{inf}^*) \le \varepsilon_m \le J_{TTC}(\boldsymbol{Y}_{sup}^*)$$
(3.7)

Where Y_{inf}^* is the optimal solution of single optimal problem (3.1) for minimum objective function *J* without any constraint, and Y_{sup}^* is the optimal solution for single optimal problem that maximize J_{TTC} in equation (3.5) in a predefined searching space. After that, for the bounded value in equation (3.7), we define the range of normal J_{TTC} values as $J_{TTC}(Y_{sup}^*) - J_{TTC}(Y_{inf}^*)$ in problem (3.6). Note that, with the ε -constraint, we can get different efficient solutions close to a strict Pareto optimal solution. Therefore, a more rich and flexible solutions are favorable in the applied traffic scenario. Thus, we can divide the ε range into *p* equal intervals by *p* + 1 "grids points" [323] like the following:

$$\varepsilon_m = J_{TTC}(\mathbf{Y}^*_{inf}) + (J_{TTC}(\mathbf{Y}^*_{sup}) - J_{TTC}(\mathbf{Y}^*_{inf})) \cdot (\frac{m}{p}), (m = 0, 1, \dots, p)$$
(3.8)

Let consider equation (3.8), we can also get efficient solutions by properly adjusting the the number of "grid points" gradually increasing ε_m by referential signs and linear logic. An indicator $\sigma(\varepsilon_m)$ to interpret the linear relationship between *J* and J_{TTC} with different ε_m is calculated as:

$$\sigma(\varepsilon_m) = \begin{cases} 1 & \text{if } \varepsilon_m = \varepsilon_p \\ \frac{\varepsilon_m - J_{TTC}(Y^*_{inf})}{J_{TTC}(Y^*_{sup}) - J_{TTC}(Y^*_{inf})} & others \\ 0 & \text{if } \varepsilon_m = \varepsilon_0 \end{cases}$$
(3.9)

For the bounded value in equation (3.7), we define the range of normal ε_m by two bound values J_{TTC} with respect to the individual optimal problems. As a matter of fact, the proposed " ε -Constraint" in the bounds is to correctly estimate the trade off between crossing time and risk which we aimed to achieve a good trajectory schedule. To guarantee the feasible solution in the bounds, we divide ε_m into several equal intervals as a constraint in original PC algorithm. Only the feasible solutions afford the constraints will be reserved in the PC searching procedure as depicted in Figure. 3.6. It is also essential to note that too small bound intervals will lead to ineffective 2D TTC constraint for a safety-sensitive solution. A simple remedy in order to bypass the difficulty of estimating lower bound is to define reservation values as shown in [323]. We capture minimum 2D TTC threshold as

1.5*s* for a reference in this section [96]. Because the strategy hypotheses include full stop actions to avoid the extreme situation (i.e., conflict immediately), thus ε -PC can filter the decision states while remains optimal feasible solutions.

To sum up, the advantages of ε -constraint method in MVS are:

- ε-constraint method in PC algorithm avoids scaling multi-objective function in a complex target function by adding too much weights, which are never so simple to fix.
- we can control the number of efficient vehicle's actions by properly adjusting ε_m with predefined grid points p. A membership function σ can indicates the degree of optimization in different objective functions.
- the feasible solutions obtained after the optimization are indeed Pareto optimal solutions.

A simply remedy in order to bypass the difficulty of estimating the worst values of the searching results (e.g., $J_{TTC}(Y_{inf}^*)$ with optimal Y_{inf}^* in (3.1) for minimum *J* is to define reservation values for the objective functions [323]. Thus, we only need to calculate the maximum $J_{TTC}(Y_{sup}^*)$ in conventional PC algorithm. In the context of the proposed MVS, several approximate block solvers are recommended as initialization fast algorithms. For example, adopting max-min resource method in [228] to calculate $J_{TTC}(Y_{sup}^*)$. It is also essential to note that too small equal intervals will lead ineffective 2D TTC constraint for safety sensitive solution. Therefore, setting 2D TTC constraint indicators are expected to regard real-life situations. Furthermore, the ε -PC algorithm can be explained with detailed flow diagram as shown in Figure. 3.6.

3.2.3.2/ ε-PC FRAMEWORK

In Figure. 3.6, as the original PC method, vehicle *i* assigns uniform probabilities $q(\Pi_k^i)$ to its strategies/actions set Π^i (for example, $q(\Pi_k^i) = 1/N$ is a distribution over Π^i) at t = 0. From a vantage point of associate a probability for the strategy Π_k^i , vehicle *i* can further compute the *N* corresponding expected system objective function values w.r.t. its strategies set Π^i . Thus, when vehicle *i* in turn to run its PC algorithm, it can help to optimize the distribution $q(\Pi^i)$ (for ego-vehicle) in an Expectation function like equation (3.10).

$$min \qquad E(q(\mathbf{\Pi}^{i})) = \sum_{k=1}^{N} J(\mathbf{Y}_{k}^{i})q(\Pi_{k}^{i}) \prod_{(i)} q(\Pi_{2}^{(i)})$$

subject to
$$(\mathbf{Y}_{k}^{i} = [\Pi_{2}^{1}, \Pi_{2}^{2}, \dots, \Pi_{k}^{i}, \dots, \Pi_{2}^{n}])$$
$$q(\Pi_{k}^{i}) \in q(\mathbf{\Pi}^{i}) = \{q(\Pi_{1}^{i}), \dots, q(\Pi_{N}^{i})\}$$
$$\sum_{k=1}^{N} q(\Pi_{k}^{i}) = 1, \quad q(\Pi_{k}^{i}) \ge 0$$
(3.10)

Where (*i*) represents every vehicle other than *i*, and $\Pi_2^{(i)}$ is the other vehicle's strategies selected randomly (with question marks "?") depending on its probability $q(\Pi_2^{(i)})$. It is important to underline that $q(\Pi_2^{(i)})$ is a priory anticipatory probability of the actions/strategies of all the other agents. For vehicle *i* in turn to minimize expectation function $E(q(\Pi^i))$,



Figure 3.6: Flowchart of the proposed ε -PC algorithm: at the initial state, for vehicle *i* in turn to run the ε -PC algorithm; a combined strategy Y_k^i including its own strategy Π_k^i and other selected strategies is given after each vehicle's iteration. Additionally, MVS keeps track of current favorable actions Y_{cur}^* and compares them to earlier ones. Finally, as the solution converges, the final optimal solution Y_{opt}^* is chosen.

a combined strategy Y_k^i include its own strategy Π_k^i and other randomly selected strategy $\Pi_2^{(i)}$ (i.e., $Y_k^i = [\Pi_2^1, \Pi_2^2, \dots, \Pi_k^i, \dots, \Pi_2^n]$). Thus, we can underline these the combined probabilities distribution of in Y_k^i w.r.t. each Π_k^i as the following:

$$q(\mathbf{Y}_{1}^{i}) = [q(\Pi_{2}^{1}), q(\Pi_{2}^{2}), \dots, q(\Pi_{1}^{i}), \dots, q(\Pi_{2}^{n})]$$

$$\vdots$$

$$q(\mathbf{Y}_{k}^{i}) = [q(\Pi_{2}^{1}), q(\Pi_{2}^{2}), \dots, q(\Pi_{k}^{i}), \dots, q(\Pi_{2}^{n})]$$

$$\vdots$$

$$q(\mathbf{Y}_{N}^{i}) = [q(\Pi_{2}^{1}), q(\Pi_{2}^{2}), \dots, q(\Pi_{N}^{i}), \dots, q(\Pi_{2}^{n})]$$
(3.11)

Thanks to the cost function in equation (3.10), it is easier to optimize probability than the original problem in (3.6). Such a method is referred as the homotopy approach that converts the primal problem into the probability space. Next, a key attraction, and most important Maximum-Entropy (MaxEnt) principle in PC algorithm is applied, so that we formulate $E(q(\mathbf{\Pi}^i))$ into equation (3.12):

$$L(q(\mathbf{\Pi}^{i}), Temp) = E(q(\mathbf{\Pi}^{i})) - Temp \times E_{free}$$

= $\sum_{k=1}^{N} J(\mathbf{Y}_{k}^{i})q(\Pi_{k}^{i}) \prod_{(i)} q(\Pi_{2}^{(i)}) - Temp \times (-\sum_{k=1}^{N} q(\Pi_{k}^{i}) \log_{2} q(\Pi_{k}^{i}))$ (3.12)

The objective function (3.12) is called MaxEnt Lagrangian and is widely used in statistical physics considering free energy as E_{free} [511]: a parameter *Temp* called temperature is specific for the Simulated Annealing (SA) process. At the beginning of the ε -PC algorithm, the parameter $Temp \in [0, \inf)$ is huge, which weights more the entropy term. Then we can always get uniform probabilities. Since function E_{free} can stand for the largest uncertainty (highest entropy) at the beginning (this means each vehicle's actions has probability 1/N of being most favorable). Shannon entropy is a general choice for function $E_{free} = -\sum_{k=1}^{N} q(\Pi_k^i) \log_2 q(\Pi_k^i)$, where it can be proved mathematically that: $argmax(E_{free}) \rightarrow q(\Pi_k^i) = 1/N$.

The formulation of the Maxent Lagrangian $L_i(q(\mathbf{\Pi}^i), Temp)$ is very appropriate in the original PC theory, since the probability nature may handle the rest of work to solve in a convex space of probability distribution. To obtain the updated probability, the Broyden–Fletcher–Goldfarb–Shannon (BFGS) method is used in PC algorithm for the reformulated optimization problem in equation (3.13):

min
$$L(q(\Pi^{i}), Temp)$$

subject to

$$\sum_{k=1}^{N} q(\Pi_{k}^{i}) = 1$$

$$q(\Pi_{k}^{i}) \ge 0$$
(3.13)

In equation (3.13), the expected global utility $L(q(\Pi^i), Temp)$ based on the combined strategy is calculated under a specific temperature Temp. Vehicle *i* updates the probabilities $q(\Pi^i)$ of all the actions after each iteration. However, the adoption of BFGS method cannot keep the probability value within [0, 1]. Even though standardization may be used to

handle such case, the interior point method [511] is recommended in the proposed ε -PC algorithm. Because it can be guaranteed that the probability keep within [0, 1] during each iteration. Interior point method have been proven to efficiently solve linear (or-nonlinear) constraints with less iterations.

After that every vehicle runs the ε -PC algorithm, the probability distribution of its actions will be updated. Combined strategies $J(Y_{cur}^*)$ with the minimum value $J(Y^*)$ will be saved as current preferred solution. It must be mentioned that, the accepted combined strategies Y_{cur}^* as current preferred solution at an iteration afford $J_{TTC}(Y_{cur}^*) \ge \varepsilon_m$. Otherwise, discard $J(Y_{cur}^*)$ and retain previous objective function value with related actions. At last, if any of the three criteria listed below is valid, then, accept the current system best solution Y_{cur}^* as the final optimal strategy Y_{opt}^* of all the vehicle.

- if temperature $Temp = Temp_{end} \rightarrow 0$
- · if maximum number of iterations exceeded
- if the difference of objective function $J(Y^*_{cur})$ between two iterations reaches the prescribed threshold of Δ

Above all, the main difference between ε -PC algorithm and the original PC framework can be highlighted as following:

- The process which confirms available constraints to the feasible solutions is considered in the PC framework in a randomly improved way with additional calculation steps. The proposed method uses accessible individual optimization process to define the ranges of constraints in advance. The grid points is inserted in the algorithm, therefore, a feasible solution can always be calculated in due time.
- The interior point method is used in the improved PC algorithm to guarantee the probability value within [0, 1]. It is essential to apply the Monte-Carlo sampling principle based these probability distribution rather than standardization.
- The process of narrowing/updating sampling interval is excluded in the proposed ε -PC algorithm. Because the searching space is well-designed before (see section 3.1.2.1). This fact leads us to the obvious advantage of reducing computational time.

The main interest of the proposed ε -PC algorithm is a proper balance between the highquality solution and acceptable computational speed. The method is flexible and produces approximation algorithm solution rather than global optimal results. It is supposed to be a good decision support system for transportation service and risk assessment. The typical MVS coordination will be tested in section 3.2.4.4.

3.2.4/ SIMULATION RESULTS

3.2.4.1/ COMMUNICATION CHARACTERIZATION FOR THE EXPECTED DATA EXCHANGE BETWEEN VEHICLES

In ε -PC algorithm, we designed a distributed approach in the hypothesis that there are *n* processors with the *i*th processor (vehicle *i*) assigned the responsibility of updating the

*i*th probabilities $q(\Pi_k^i)$ of actions/strategies set Π^i according to equation (3.13). Processor *i* inform processor *j* (e.g., vehicle *j*) to calculate its preferred actions/strategies relying information of preceded probabilities $q(\Pi_k^i)$. All the processors repeat the process until the convergence the aforementioned iterative sequence (cf. Figure. 3.4).

Due to the limitations in the measurement and control units, it is often impossible to acquire measurements at an arbitrarily fast speed and to execute the control inputs instantaneously. Thus, the MVS in this section is described in a continuous-time setting while measurements and control inputs are described in a precise constant sampled data $T_{sample} = 0.2s$. It is assumed also that the messages are received correctly within a finite time (but still leave an open way to consider synchronous or asynchronous implementations). In our predefined scenario, for example three connected vehicles in Figure. 3.3, each action/strategy is a float vector of size 50 (10s horizon and 0.2s sampling time) and the set of possible velocity profile has 10 strategies as shown in Figure. 3.2a. That is a total of 1500 floats-5.86kB (kilo Bytes)-for three vehicles. It is done again at the beginning of the re-acceleration phase (cf. section 3.1.2.2). Consequently total 11.72kB is prior data volume needs to be broadcasted in the considered case. Then, for each iteration the vehicle broadcasts its updated probability vector $q(\Pi_t^i)$ of 500 floats (10s horizon and 0.2s sampling time with 10 strategies). The number of iterations is various in different mode according to experimental statistics. So the total broadcast per vehicle is depend on the iterations. Accordingly, the communication demand in "single" (only one vehicle) and "full" modes (for three vehicles) are compared in Table 3.1 as an instance.

Single mode	Value	Full mode	Value
Prior data	11.72 <i>kB</i>	Prior data	11.72 <i>kB</i>
Strategy	10	Strategy	10^{3}
Iterations	10	Iterations	$20 \sim 50$
Probability data	500 <i>floats</i> (1.95 <i>kB</i>)	Probability data	1500 <i>floats</i> (5.86kB)
Data volume	19.53 <i>kB</i>	Data volume	117.19 <i>kB</i> ~ 292.97 <i>kB</i>
Total volume	31.25 <i>kB</i>	Total volume	$128.91 kB \sim 304.69 kB$
Solver time	0.2s	Solver time	0.8s
Network request	0.4MB/s	Network request	0.4MB/s
Physically possible	4MB/s	Physically possible	4MB/s

Table 3.1: Data exchange in "single" mode and "full" mode for three vehicles

We suppose that the required network throughput for all modes will not exceed a magnitude of 0.4MB/s as shown in Table 3.1. Then, the optimization can be achieved in about $0.2s \sim 0.8s$. A faster network throughput (e.g., 4MB/s) is physically possible. Therefore, our ε -PC method may be executed with on-board processors that have a better C++ implemented code for a faster running. But it is important to note that the experiments in this section were all run by a program developed in MATLAB with a computer of Core i5-6300HQ, 2.30GHz and 8GB RAM.

3.2.4.2/ PARAMETER SETTING AND RESULTS EVALUATION

Indeed, the single and full modes are both processed by ε -PC. The main difference between the two strategies is that the single mode only considers its best options w.r.t. the fixed strategies of others. It is a special case of full optimization that vehicles have several actions that may be chosen for self-interested behavior. Therefore, it is discussed in what follows only full optimization of MVS. To explain better the performance of the proposed intersection navigation scheme for the MVS based on ε -PC, let us decompose the experiments into several parts to evaluate the characteristics of the algorithm. The main parameters in our proposed algorithm are presented in Table 3.2:

Parameter	Value	Parameter	Value
(x_1, y_1)	[-20, -2.5] [m]	V _{min}	0 [m/s]
(x_2, y_2)	[20, 2.5] [m]	<i>v_{max}</i>	10 [m/s]
(x_3, y_3)	[2.5, -20][m]	r	1.5[m]
$v_1(0)$	6.0[m/s]	T _{sample}	0.2[s]
$v_2(0)$	5.0 [m/s]	T	10
$v_{3}(0)$	5.5 [m/s]	t _{act}	3
W_{sep}	1	N _{samples}	10
W _{cross}	10	Nvehicles	3
Δ	0.01	TTC_{min}	1.5[s]
ε_1	1.5	$\sigma(\varepsilon_0)$	0%
ε_2	1.97	$\sigma(\varepsilon_1)$	33.33%
ε_3	2.43	$\sigma(\varepsilon_2)$	66.66%

Table 3.2: Parameters and initial states

To evaluate the proposed method, contract experiments between original PC and ε -PC are given in a simulation. Three vehicles cooperative navigation at an intersection with predefined trajectories which include left-turn maneuvers for vehicle 1 and vehicle 2 (as shown in Figure. 3.7a and Figure. 3.8a). One of the important property highlighted in the simulation is the safety of the proposed navigation strategy and the ability to avoid collisions. In original PC, the cost function considered by MVS includes the average crossing time (altruistic objective) and the separation distance as shown in equation (3.1). The simulation results are illustrated in Figure. 3.7b. Because the control effort has been focused on the crossing time ($W_{cross} > W_{sep}$), the original PC method attempted to maintain a fast crossing speed. Thus, there is a low probability but high impact to choose extreme strategy which allows all vehicles to accelerate as shown in Figure. 3.7b (a). Such speed growth had led to a collision as the distance indicator exhibited in Figure. 3.7b (b): the distance between vehicle 2 and 3 (purple line) violated the safety limit of 2r = 3m. The 2D TTC profile of this two-vehicle also collapsed to zero during the collision as Figure. 3.7b (c).

In comparison, ε -PC made vehicle 1 to maintain current speed at the beginning two seconds in order to increase the distance between adjacent vehicles as indicated in Figure. 3.8b (a). Due to the threshold constraint of $\varepsilon \ge TTC = 1.5s$ with respect to equation (3.6), a 100% free collision navigation can be guaranteed in the whole time horizon [0s, 10s] as the indicator of distance and 2D TTC underlined in Figure. 3.8b (b) and Figure. 3.8b (c).

Several ε -constraint values are carried out to highlight the performance of the proposed method in the previous scenario (cf. Figure. 3.8a). The approximated maximum $J_{TTC}(Y^*_{sup}) = 2.8975$ is fixed in predefined MVS by initially heuristic searching (cf.



(b) Performance indicators: distance, velocity and 2D TTC in MVS by original PC.

Figure 3.7: Three vehicles navigation by original PC (simulation video https://bit.ly/ 3kkRJ9v).

Sec 3.2.3.1). Here, the reservation value $J_{TTC}(Y_{inf}^*) = 1.5$ for a minimum 2D TTC in whole time horizon [0s, 10s]. After that, it is used several grids point (cf. equation (3.8)) in the range of $1.5 \le \varepsilon_m \le 2.8975$ to get a constraint set ε_i and membership function $\sigma(\varepsilon_i)$ (cf.



(b) Performance indicators: distance, velocity and 2D TTC in MVS by ε -PC.



equation (3.9)) as presented in Table 3.3.

A comparison of average intersection crossing time (in presence of three vehicles) during four trails is shown in Table 3.3. It is instructive to note that the original PC algorithm, which does not use any 2D TTC constraint (i.e., $\varepsilon = 0$), shows a fastest crossing time

with as expected a lowest performances of 2D TTC. Increasing ε_i with weighting more on the membership function $\sigma(\varepsilon_i)$ can generally provide more better temporal margins of 2D TTC while increasing vehicles average crossing time. However, ϵ -PC can still avoid conservative actions/strategies with low crossing time in dangerous situation as in trial 4 (about 1.33s late than trial 1). Moreover, the increasing iteration numbers implies that the convergence of the model needs more execution time. Therefore, MVS has potential applications in different navigation environments when a proper selective ε -constraint model is designed.

	ε_i	$\sigma(\varepsilon_i)$	Average crossing time	Iterations	J_{TTC}
Trial 1	0	-	3.60 [s]	20	0.04 [s]
Trial 2	1.5	0%	4.70 [s]	21	2.07 [s]
Trial 3	1.97	33.33%	4.70 [s]	21	2.07 [s]
Trial 4	2.43	66.66%	4.93 [s]	26	2.51 [s]

Table 3.3: Performance co	omparison of different $arepsilon_i$
---------------------------	--------------------------------------

3.2.4.3/ SCALABILITY PROPERTIES

To enable the qualified strategy to fulfill certain specific situations in MVS navigation (the long tail challenge), it is important to reserve enough strategies in searching space. In order to have a clear picture of the computational demanding under such applications, the execution time of varied number of strategies for MVS is compared in the Figure. 3.9.



Figure 3.9: Effect of strategy number on the execution time.

The results given in Figure. 3.9 show a slight difference in terms of the execution time

between the original PC and ε -PC. More importantly and within this small decrease in the execution time, the ε -PC outperforms the original method by its capacity in avoiding the potential collision. Based on the results shown in Figure. 3.9, it can be admitted that the ε -PC method guarantees the no collision between vehicles without any additional computational burden to satisfy the minimum safety navigation requests. As a reminder, minimum 2D TTC = 1.5s is specifically considered in all these simulations of ε -PC. Furthermore, it appears that setting $N_{samples} = 10$ improves the satisfaction of the computationally demanding. To conclude, over-or under-estimated static sampling strategy may lead to more execution time to find optimal results that can meet the requirements.

The complexity of the ε -PC algorithm is strongly correlated to the scalability properties of the proposed method. So that we attempt to run ε -PC algorithm with an increasing number of vehicles. The randomized initial parameters (position and speed) are within the same range. Vehicles are generated with different distance to the intersection. The result has further explored the limiting number of vehicles for the application of ε -PC in more exhaustive navigation states (see Figure. 3.10).





Figure 3.10: Effect of vehicle number on the performance of ϵ -PC.

Figure. 3.10 shows results for $N_{vehicles} = 3$ to $N_{vehicles} = 6$. The execution time of per vehicle show a steady increase for both the number of vehicle and the average crossing time. The $\varepsilon = 1.5s$ is fixed in each simulation. This indicates that ε -PC can guarantee a safe navigation with increasing number in MVS. A further analysis of complexity will be carried out in further works, and more realistic execution time will be measured after an optimization of the produced code and its parallelization.

3.2.4.4/ EXPERIMENTS USING CONTINUOUS VEHICLE FLOW

In the case of the "single mode", all the vehicles are non-collaborative. This mode is useful to demonstrate the capabilities of the algorithm to deal with a high number of non-collaborative connected vehicles. Figure. 3.11 illustrates the continuous traffic situation during the simulation.





Figure 3.11: MVS navigation in real-time (simulation video https://bit.ly/3KojvfO).

It is demonstrated also that the use of the algorithm in the special case of single vehicle optimizes the navigation performances. On the other hand, the "full" mode shows the exact opposite as it forces all the vehicles to run the optimization when triggered by a predefined event. This mode has been used in the previous section when an optimization is done on a fixed initial situation with all the vehicles. It should be noted that "single mode" has been preferred for a fast decision making. While "full mode" is more preferred for collaboration in complex environment. In the tackled simulation (see Figure. 3.11), no conflicts did happen between any of the present vehicles, even though they did the optimization one by one.

3.3/ BEYOND LOCAL COOPERATION: SPEED VERSUS RISK FROM A GLOBAL PERSPECTIVE

This section discusses MVS navigation in a protected region that includes multiple intersections. It is expected, in particular, that all vehicles are trained to accomplish internal transportation tasks. As indicated in section 2.2.4, the suggested approach in this PhD thesis is intended to route and schedule vehicles in a manner that strikes an optimal balance between speed and risk. Thus, a MVS is supposed to be regulated by a hierarchical supervision architecture that distributes and schedules pick-up and delivery tasks at the highest level while ensuring intersection safety and trajectory control at the lowest level. Unlike the previous parts (i.e., section 3.1 and section 3.2), which concentrated on cooperation at a single intersection, this section will analyze the top level from a global perspective while also providing a time-dependent estimation of the risk generated by traversing any path at a given time. Furthermore, a local search algorithm involving dynamic programming, which implements the Reinforcement learning (RL) principle, is analyzed with experimental results. More specifically, section 3.3.1 explains the general context and relative definitions. In section 3.3.2, we formally describe global risk management model and state some structural results. In section 3.3.3, we describe the Local Search (LS) heuristic algorithm for Shortest Path under Risk constraint (SPR) problem (short for "LS-SPR") that incorporates the RL principle. Section 3.3.4 is devoted to numerical experiments.

3.3.1/ DEFINITIONS AND GENERAL CONTEXT

As mentioned in section 2.2.2 (Figure 2.8), the hierarchical supervision architecture is proposed to manage a MVS system relying on the 2nd and 3rd level. Recalling that the first level, or micro-level, is defined by the monitoring and sensing devices that are embedded inside the vehicles, they compute the trajectories in real time and adapt them to the possible presence of obstacles. The majority of the robotics community's effort is still focused on this micro-level, which primarily involves optimal control and artificial perception techniques [67, 288, 398]. The second one, or management-level, is in charge of the supervision of small tricky areas, like for instance intersections or loading/unloading spots (see Figure 3.3). Working as a mediator agent, it sends signals and instructions to the vehicles in order to regulate their transit and to avoid them to collide when they get through those areas. This level has been motivating a rise in interest for the last years [118, 391, 550], and sometimes a confusion with the first level: in many cases, hypothesis

is set that all vehicles involved are run by the same embedded software and exchange perfect information; this become equivalent to supposing the existence of a local external mediator. The third one, or macro-level, consists in tactical dynamic planning and routing of the MVS, in order to make vehicles achieve some internal logistics requests [137, 477]. Depending on the complexity and the size of MVS, the management-level¹ may merge with either the micro-level or the macro-level. In any case, a true challenge is the synchronization of those monitoring levels to correspond to different time scales and goals, as well as creating communication protocols that will allow them to interact.

The goal in this section is to deal with the macro-level. A global supervisor is supposed to be deployed at the top level. By some aspects, related problems may be viewed as cases of well-known Pick up and Delivery Problem (PDP, see [53]), since in most cases a task will consist of moving a vehicle in a protected area, performing some loading or unloading transaction. But two specific features are going to bring its specificity to this PDP variant:

- The time horizon of autonomous or semi-autonomous vehicles is usually somewhat short: decisions have to be taken fast, in a dynamic context, and decisional processes must take into account the communication infrastructure [430] and the way the global supervisor can be provided, at any time, with a representation of the current state of the system and its short-term evolution;
- As soon as autonomous or semi-autonomous vehicles are involved, safety is at stake: the global supervisor must compute and schedule routes in such a way that not only are tasks going to be performed fast (standard industrial efficiency) but also that local manager will perform their tasks more easily. In other words, risk minimization should be a criterion for a good schedule (see [19, 46, 362, 454]).

A consequence is that performing the top-level supervision of MVS requires disposing at any time of an accurate representation of the current state of the system and its shortterm evolution. This representation should enable us to quantify the risk induced by an additional vehicle that enters the transit network and is asked to follow a given trajectory. Addressing this risk quantification problem requires converting real-time acquired data about current traffic into risk estimators using statistical data analysis (see [401, 493]). We are not going to directly address this issue. Instead, we will concentrate on how the resulting estimators may be utilized to make routing and scheduling decisions that avoid tricky situations as much as possible and so simplify management of the vehicles at the management and embedded levels. More precisely, we are going to suppose that, at the time when we are trying to schedule this vehicle, supposed to perform a move (or a sequence of moves) from some origin o to some destination d, we are provided with a procedure that, for any arc e = (x, y) of the transit network and any time value t, computes an estimation of the risk resulting from the presence of our vehicle on arc e at time t. Then our goal becomes to compute and schedule the route Γ of our vehicle in such a way that its riding time is minimized and that its induced risk estimation which does not exceed some threshold $Risk_{Max}$. For the sake of simplicity, we shall limit ourselves to a single task tour, which means that Γ will be constrained by its starting point o and its destination point d. Described this way, our problem might be viewed as the search for the constrained shortest path [137]. But the fact that both risk and arc traversal times are time-dependent makes the problem significantly more difficult (see [137]). Similarly,

¹This PhD thesis will be expanded on the discussion of the management-level in Chapter 4.
the on-line capability of a MVS system prevents us from relying on heavy mathematical programming, and forces us to develop highly reactive decision-making tools

We propose three steps for achieving the above-mentioned purpose here:

- The first step is devoted to configuring our SPR: Shortest Path Under Risk (SPR) problem, as well as a discussion of its complexity and some structural aspects. The SPR problem is associated with time versus risk routing issues that can be involved within for instance a warehouse environment.
- The second step is devoted to the design of static-context algorithm LS-SPR: Local Search (LS) heuristic algorithm for SPR problem (i.e., LS-SPR), whose structure can be compared to the structure of Split algorithms, which implement *Route First-Cluster Second* approaches [44] for Vehicle Routing Problem (VRP) and estimate the quality of a given route using a filtered dynamic programming procedure.
- The final step is to implement the RL principle in our searching algorithm: the goal is to convert the above-mentioned static algorithms LS-SPR into reactive algorithms for on-line contexts. According to this hypothesis, we use statistical learning and auto-adaptive RL techniques to correlate ad hoc arc traversal decisions with any present traffic patterns.

3.3.2/ GLOBAL TRANSIT NETWORK AND SPR MODEL

Transit Network and Risk Function: it is supposed that, MVS moves inside a simple planar transit network G = (N, A), N denoting the nodes of G, and A corresponds to the set of arcs, likely to represent, for instance, a constrained area (e.g., a warehouse see Figure 3.12 below).

Every arc e = (x, y) is provided with a maximal speed of V_{Max_e} and a length of L_e . At the time t = 0, when the global supervisor of MVS needs to make a decision about the



Figure 3.12: A warehouse like transit network: at time *t*, both green and black vehicles are scheduled in a risky area within an arc e(A, B) (inspired from [38]).



Figure 3.13: A piecewise function $\Pi^{e}(t)$.

target vehicle *VEH* (short for *VEHICLE*), it knows about the routes followed by the other vehicles and the tasks they are going to perform. This knowledge enables the vehicle to be provided with a rough representation: for any arc e = (x, y) and any future instant t > 0, with an estimation of the number of vehicles and obstacles which are going to be located in *e* at time *t*. For instance in Figure 3.12, both green and black vehicles will travel through an arc e(A, B) at time *t*, e(A, B) can be considered a risky area along the scheduled path. This allows global supervisor to derive a risk estimation $\Pi^e(t)$, the meaning of which is:

• For any small value dt, $\Pi^{e}(t)dt$ is the *Expected Damage* between time t and time t + dt in case *VEH* moves at maximal speed $V_{Max_{e}}$ along e during this period.

Obtaining function $\Pi^{e}(t)$ is not included in this study: the *Expected Damage* is going to be assessed by experimental data and statistical analysis (e.g., observing risk behaviors such as hazardous breaking and steering in real-world driving [401]). In fact, these risky behaviors depend on spatial and temporal factors (e.g., time of day, road type and properties, etc.). When a high frequency of unexpected and dangerous maneuvers is detected, autonomous driving becomes difficult or impossible in a crowded *e*. Intuitively, it will involve number of vehicles related to the arcs. Because the global supervisor must be able to retain such risk estimate functions $\Pi^{e}(t)$ throughout the supervisory process, we must do so in a structure that makes them easy to calculate and update. Thus, it is supposed that any function of $\Pi^{e}(t)$, which translates those configurations into risk, is a piecewise linear function (see Figure 3.13) with a not too large number of break points. Particularly, we call the break points of $\Pi^{e}(t)$ when the values *t* make the value of $\Pi^{e}(t)$ changes. Thus, if *VEH* traverses arc *e* during some interval [*t*, *t* + *dt*] at a speed of $v \leq V_{Max_e}$, then related *Expected Damage* is given by (3.14):

$$Risk^{e}(v,t) = \Phi(v/V_{Max_{e}}) \times \Pi^{e}(t)dt$$
(3.14)

Where Φ is an increasing convex function with values in [0, 1] and such that for any value $u = v/V_{Max_e}$ where $u \in [0, 1]$, $\Phi(u)$ is significantly smaller than u. Those conditions are imposed in order to confirm the intuition which tells that the slower the vehicle moves, the

smaller the resulting risk. Particularly, we define $\Phi(u) = u^2$ in this PhD thesis. As a result, if vehicle *VEH* moves across arc *e* between time *T* and time *T* + σ , according to speed function *v*(*t*), then the related *Expected Damage* can be developed as:

$$Risk^{e}(v,t) = \int_{T}^{T+\sigma} \Phi(v(t)/V_{Max_{e}}) \times \Pi^{e}(t)dt = \int_{T}^{T+\sigma} (u(t))^{2} \times \Pi^{e}(t)dt$$
(3.15)

Routing strategies: it is supposed now that origin *o* and destination *d* are given as nodes of the transit network G = (N, A) (see Figure 3.12). A routing strategy for a vehicle is going to be a pair (Γ, v) , where Γ is a path from *o* to *d* in the network *G*, and *v* is a speed function, which, to any time value $t \ge 0$, v(t) is the corresponding speed of the vehicle. Clearly, if at time *t*, *VEH* in located on arc $e \in \Gamma$, then v(t) must not exceed V_{Max_e} . Specifically, a speed normalization is assumed to have $V_{Max_e} = 1$. Thus, all that matters here is that the speed value $u \in [0, 1]$ in the model. Path Γ may be viewed in a standard way as a sequence $\Gamma = \{e_1, \ldots, e_n\}$ of arcs of *G*. If we set T(0) = 0 and denote by T(i) the time when *VEH* arrives to the end-node of e_i , then values T(i) are completely determined by speed function u(t). Then the global duration and risk of the routing strategy (Γ, u) can be denoted as follows:

$$G_{Time}(\Gamma, u) = T(n)$$

$$G_{Risk}(\Gamma, u) = \sum_{i=1}^{n} \int_{T(i-1)}^{T(i)} \Phi(u(t)) \times \Pi^{e_i}(t) dt$$
(3.16)

Shortest Path under Risk (SPR) Model: then the purpose becomes in a natural way to make vehicle *VEH* move from *o* to *d* while achieving minimal values of G_{Time} and G_{Risk} values. In fact, risk and time play very different roles inside a real industrial system, and so the risk is usually managed as a constraint rather than through a weighted method. Thus, some threshold $Risk_{Max}$ is given and the trajectory (Γ , *u*) of vehicle *VEH* is required to be such that resulting risk G_{Risk} does not exceed threshold $Risk_{Max}$. It comes that our SPR comes in a natural way as follows:

• SPR model: given origin o and destination d, together with some threshold $Risk_{Max}$, compute a routing strategy (Γ, u) such that $G_{Risk} \leq Risk_{Max}$ and G_{Time} is the smallest possible.

Indeed, the SPR model is a multi-objective optimization. A challenge that arises is how to convert risk to time in a way that allows us to solve a mono-objective problem. This leads us to establish the *Risk versus Distance* coefficient λ , which indicates that the anticipated risk in per unit distance dR/dL. One may check the quantity $u(t)\Pi^{e_i}(t)$ means the instantaneous risk per distance at the time *t* when *VEH* move along e_i . Thus, we have:

$$\lambda = \frac{dR}{dL} = u(t)\Pi^{e_i}(t)$$
(3.17)

Reconstructing routing strategies: the above results can significantly simplify the SPR model by replacing the search for speed $t \to u(t)$ (in a strategy (Γ, u)) with a search for risk per distance $e \to \lambda$. Further, seeing that $\frac{1}{2}\Phi'(u)\Pi^{e_i}(t) = u(t)\Pi^{e_i}(t)$ in case $\Phi(u) = u^2$. We define a risk versus distance strategy as a pair $(\Gamma, \lambda_{e_i}^{RD})$ where:

Γ is a path, that includes a sequence {e_i,..., e_n} of arcs, which connects origin node *o* to destinations node *d*;

• $\lambda_{e_i}^{RD}$ is Risk versus Distance coefficient associating with any arc e_i in Γ , which is defined by $\lambda^{RD} = \frac{1}{2} \Phi'(u) \Pi^{e_i}(t)$.

In the PhD thesis, it is supposed that both u(t) and λ_e^{RD} are piecewise constant on e, we see that the knowledge of (Γ, λ_e^{RD}) allows us to reconstruct standard routing strategy $(\Gamma, u(t))$. Therefore, the SPR model may be rewritten as follows (we extend previous notations $G_{Time}(\Gamma, u)$ and $G_{Risk}(\Gamma, u)$ by denoting by $G_{Time}(\Gamma, \lambda^{RD})$ and $G_{Risk}(\Gamma, \lambda^{RD})$ respectively as the time value and risk value of a risk driven routing strategy (Γ, λ^{RD}) :

• *Risk versus Distance SPR Reformulation:* compute risk versus distance strategy (Γ, λ^{RD}) such that $G_{Risk}(\Gamma, \lambda^{RD}) \leq Risk_{Max}$ and $G_{Time}(\Gamma, \lambda^{RD})$ is the smallest possible.

3.3.3/ LOCAL SEARCH ALGORITHM FOR SPR

The method for the SPR issue is based mostly on a notion of *state* and a notion of *decision*. A state is 3-tuple (i, T, R) in which:

- *i* is a node of *G* where vehicle *VEH* is currently located;
- *T* is the times spent in order to reach *i*, and *R* is the amount of risk induced by this process of moving from origin *o* to node *i*.

The *state* denotes the amount of time and risk required to travel an arc *e*. As a result, a transition function for updating the *state* may be created. Notably, as noted above, such a transition is dependent on the risk versus distance coefficient λ . Therefore, our *decision* process is as follows:

- Choosing the arc *e* along which the vehicle is going to move;
- Choosing some parameter λ which is going to determine the speed function u along the arc e.

For the sake of simplicity, we still restrict $\Phi(u) = u^2$. Further, the *Algorithm 1* is given containing the method for the proposed transition function, which is based on the assumptions and equations in section 3.3.2. A heuristic method for local search LS-SPR is designed for the suggested SPR model's decision process, which consists of two steps (as seen in *Algorithm 2*):

The *Update* step is designed in order to modify Γ and improve its quality. More precisely, the arrival time of the vehicle *VEH* at any node *i* along the route Γ is specified in *Evaluate* step. Considering that, the consumed time and/or risk in local pairs of nodes along the entire path Γ may remain high. Thus, the *Update* step will test a change to Γ (which involves some proximity threshold) and implement it only if it improves the performance of Γ . Following is a more detailed overview of these two steps.

Update step: it supposes that some proximity threshold S_{Prox} has been fixed, and that for any two nodes *i*, *j* of the transit network *G* such that length $L_{ij} \leq S_{Prox}$, MVS are provided with a collection Path(i, j) of elementary route from *i* to *j*. This allows us to introduce a local transformation operator $Detour(\Gamma, i, j, \gamma)$, which acts on any path Γ through parameters *i*, *j* and γ : *i* and *j* are two nodes of Γ , such that $L_{ij} \leq S_{Prox}$ and *i* is located before *j* Algorithm 1: Transition function for SPR

	nput: $(i, T, R), \lambda, \{e, \Pi^e(t)\}$						
(Output: (<i>i</i> , <i>T</i> , <i>R</i>)						
/	/* $\Pi^e(t)$ is the piecewise risk function of arc e and λ denotes the						
	decision value obtained in algorithm 2. The output (i, T, R)						
	represents a state transition between two nodes with associated						
	time and risk. */						
1	nitialization: $i = 0, T = 0, R = 0, L = 0$, and NotS to $p = true$;						
2 ۱	while $NotStop == true$ do						
3	$\pi = \Pi^{e}(t), u = Inf(1, \lambda/\pi); $ // Refer to equation (3.17)						
4	$t = Inf((length(e) - L)/u, t_q);$ // $t_q \in [t_1,, t_Q]$ denotes the breakpoints						
5	$L = L + ut, T = T + t, R = R + u^2 \pi t;$ // Refer to equation (3.16)						
6	i = getUpdateNode(e, L); // update the node <i>i</i> by current position						
7	if $\underline{R > Risk_{max}}$ then						
8	break						
9	if $\underline{L} > length(e)$ then						
10	$\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ $						
\sqsubseteq 11 return (i, T, R)							

Algorithm 2: LS – S PR: A local search algorithm for SPR

Input: G = (N, A), (o, d)Output: Γ^* /* Input the graph G = (N, A) with the vehicle's desired origin and */ destination (o, d) and output the best scheduled path Γ^* 1 Initialization Γ from *o* to *d* and *NotStop* = *true*; 2 while NotStop == true do $\Gamma_1 \leftarrow Evaluate \quad \Gamma$; // Running Dynamic Programming (DP) procedures 3 $\Gamma_2 \leftarrow Update \quad \Gamma_1$; // Running Update algorithm to improve Γ_1 4 if Improve failed then 5 NotStop = False;6 $\Gamma^* = \Gamma_2$; // keep the best solution Γ ever obtained 7 8 return Γ^*

Algorithm 3: Update algorithm

	nput: Γ					
(Output: Γ^{ipv}					
1 \	while <i>NotStop</i> == <i>true</i> do					
2	<i>Generate</i> (i, j, γ) ;	<pre>// Generate a 3-tuple</pre>				
3	Schedule $Detour(\Gamma, i, j, \gamma)$;	<pre>// Using a Detour operator</pre>				
4	if $\underline{TIME(Detour)} \leq \underline{TIME}(\Gamma)$ then					
5	NotStop = False;					
6	$\Gamma^{ipv} = Detour(\Gamma, i, j, \gamma) ;$	// Replace Γ by Detour				
7 ľ	7 return I ^{vpv}					

on Γ . Noting that, the *Detour* replace $\Gamma_{ij} \in \Gamma$ from *i* to *j* by path $\gamma \in Path(i, j)$. The Update algorithm is given in *Algorithm 3*.

Due to the possibility that *Detour* may include a significant number of parameter values (Γ, i, j, γ) , we must describe how those values will be created prior to applying *Detour*. Intuitively, we find pairs of nodes (i, j) in path that have a significant crowd coefficient $(R_j - R_i)/(T_j - T_i)$ and then choose one of these pairs (i, j). Then, we select a route in Path γ that is not too crowded between times T_i and T_j (it means ratio $(R_j - R_i)/(T_j - T_i)$ is not large with respect to the mean risk $\Pi^e(t)$ between time T_i and T_j).

Evaluate step: as illustrated before, *Evaluate* step means the key point inside the *decision* process given in *Algorithm 2*. Basically, it consists in a dynamic programming (DP) procedure DP_{eval} whose main features come as follows:

- the arcs of current path Γ is denoted by $\{e_1, \ldots, e_n\}$, and by $\{i_0, \ldots, i_n\}$ related nodes. The time space of DP_{eval} comes in a natural way as the set $\{0, 1, \ldots, n\}$ and a state (or label) at *i* is a pair (T, R), where *T* and *R* are defined as before in *state*. For any *i*, we shall denote as S tate(i) the set of *state* computed in relation with *i*. Those states will be used in order to move along arc e_i . Clearly, initial state is (0, 0) and final state should be any pair (T, R) such that $R \le Risk_{Max}$.
- a decision at time becomes a value λ (equal to λRD referring to section 3.3.2) and such a decision induces a state transition (cf. *Algorithm 1*): (*i*, *R*, *T*) → (*i* + 1, *T*°, *R*°) with a cost *R*° −*R* (or *T*° −*T*). This decision is feasible if it does not induce *R* > *Risk_{max}* (*Algorithm 1*, line 7).
- the Bellman principle is performed in the DP procedure DP_{eval} : applying Bellman principle means eliminating state (T1, R1) that not in Pareto optimal set from State(i+1). For instance, state (T1, R1) are not in Pareto optimal set if there does not exists (T^1, R^1) in State(i + 1), such that $T1 \leq T^\circ$ and $R1 \leq R^\circ$. One at least of those inequalities is being strict. In other words only non-dominated solution set $R1 \leq R^\circ$ are reserved into State(i + 1).

Generate λ -decision set Λ : the above DP procedure DP_{eval} rely on an instruction to generate any decision λ^{RD} which is the part of an optimal solution. However, λ^{RD} values are continuous ones and if we want to implement both above algorithms, we must proceed in a heuristic way and decide about the way to generate the λ -decision set Λ . In this PhD thesis, a mean value for λ^{RD} is defined as:

$$\lambda_{Mean}^{RD} = \frac{Risk_{Max}}{L_{o,d}^*}$$
(3.18)

Where, $L_{o,d}^*$ is the whole length from o to d. Then a way to generate λ -decision set Λ is to choose an odd number 2K + 1 of λ^{RD} values as the decision number. Next, we set a geometric step value $\delta > 0$, and sampled the decision sets $\Lambda = \Lambda_1 \cup \Lambda_2 \cup \Lambda_3$ by equation (3.19). According to this, Λ can be determined by the parameters K and δ .

$$\Lambda_{1} = \{\lambda_{Mean}^{RD}\}$$

$$\Lambda_{2} = \{(1 + \delta)^{k} \lambda_{Mean}^{RD}, k = 1, \dots, K\}$$

$$\Lambda_{3} = \{(1 + \delta)^{-k} \lambda_{Mean}^{RD}, k = 1, \dots, K\}$$
(3.19)

Bounding state through learning: as mentioned in section 3.3.2, SPR puts computing costs are at stake. Above DP_{eval} procedures should be very fast, and so State(i) should be constrained, or it may expand in size as the number of arcs in the route transit network increases. To address this problem, we are going to rank pairs (T, R) in State(i) according to growing values $\omega T + R$ and maintain the top S ones while eliminating the others. Particularly, ω is a risk rate per unit of time, and S is the limited number of states. In addition, ω can be denoted by equation (3.20) with an optimal SPR value (T^*, R^*) :

$$\omega = \frac{Risk_{Max}}{T^*} \tag{3.20}$$

However, the objective in this case is to learn ω value from the SPR model's pre-defined values. Thus, supposing that the mean value of $\Pi^{e}(t), e \in A$ is Δ . The following functions may be deduced from equation (3.15):

$$u^{2}\Delta T = Risk_{max}$$

$$uT = L^{*}$$
(3.21)

Then, we can deduce the expected $T = \frac{\Delta L^{*2}}{Risk_{max}}$ from equation (3.21). Referring to equation (3.20), this will allow us to initialize ω as:

$$\omega = \frac{Risk_{Max}^2}{\Delta L^{*2}} \tag{3.22}$$

Nonetheless, under such a chosen mechanism, a ω value that is not fully well-fit may be obtained, resulting in an unbalanced *State(i)*. Thus, using RL techniques, we may consider to learn ω value in an auto-adaptive way. More precisely:

- We fix the number 2K + 1 in λ -decision set Λ , and impose a threshold S on the size of state with initialed ω value (w.r.t. equation (3.22)). Those values K, S and ω become parameters of the LS-SPR algorithm. During some process to compute State(i+1) through transition function (cf. *Algorithm 1*), the current state State(i) may provide a subset State(i + 1) whose size is like to exceed S by applying decisions in Λ . Then we rank state (T, R) of State(i + 1) according to $\omega T + R$. Ideally, states (T, R) ordered this way should make S best states (T, R) be balanced in the sense that risky states should get along within a reasonable level.
- According to this, the ranked state (T, R) in State(i+1) should have a ratio $R/Risk_{Max}$ centered around the ratio TIME(o, i+1)/TIME(o, d). If, for instance, those values are centered significantly above this ratio, then we deduce that we are moving in a too risky way and must make ω decrease. Conversely, if those best values are centered above this ratio, then we are too careful and must make ω increase. Then we drive ω values in an auto-adaptive way by implementing some RL principles.

We implement this principle by performing a kind of statistical analysis of those best values in *State*(*i* + 1). In order to derive, from those *S* best states (*T*,*R*), an indicator *Risk_{Balance}*, which takes symbolic values {*Risky*, *Normal*, *Careful*} depending on the way the mean *R*/*Risk_{Max}* is located with respect to *TIME*(*o*, x_{i+1})/*TIME*(*o*, *d*). The filtering and learning Procedure can be addressed as given in *Algorithm 4*.

Algorithm 4: Filtering learning Procedure

```
Input: State(i)
   Output: (T, R)^* for State(i + 1)
   /* Input the current state State(i) and output the S best states (T,R)^*
       in State(i + 1).
                                                                                               */
1 State(i+1) \leftarrow State(i);
                                                  // Compute State(i + 1) by Algorithm 1
2 Rank (T, R) of S tate(i + 1);
                                                   // Rank states according to \omega T + R
3 (T, R) \leftarrow State(i + 1);
                                           // Select S best (T, R) according to Rank
4 compute Risk<sub>Balance</sub>;
5 if Risk<sub>Balance</sub> == Normal() then
      (T,R)^* = (T,R);
                                      // keep only the S best states in State(i+1)
6
7 else
       if \underline{Risk_{Balance}} == Risky() \text{ or } Risk_{Balance} == Careful() then
8
           S_1, S_2 \leftarrow Split \quad State(i+1); \quad // Split \quad State(i+1) \text{ into two subsets}
9
            with same size
           if \underline{Risk_{Balance}} == Risky then
10
               Keep only the S/2 best states in S_1, S_2 in State(i + 1) and decrease \omega;
11
           else
12
              Keep only the S/2 best states in S_1, S_2 in State(i + 1) and increase \omega;
13
14 return (T, R)^*
```

3.3.4/ COMPARISON EXPERIMENTS

We performed several numerical experiments with the purpose of getting information about the following points:

- The ability of the different algorithms to get good solutions under small computational costs, and the dependence of their behavior to the size of the transit network.
- The sensitivity of those algorithms to the parameter S and K, which bounds, for every algorithm, the numbers of possible states and decisions.
- The sensitivity of our algorithms to the structure of the piecewise functions Π^e, and on the intensity of current traffic inside the transit network at the time when the algorithms are applied.

In order to do so, we used the A^* like algorithm, run with large *S* and *K* values as an almost exact algorithm, which provided us with reference results. Additionally, a greedy approach that removes the *update* step in the LS-SPR method (cf. *Algorithm 2*) is used for comparison.

Instances: We generated networks G = (N, A) as connected symmetric partial grids, which means grids $n \times m$, modified through removal of a percentage $\rho = 30\%$ of nodes and arcs. The arc connecting two points is a one-way direction. Finally, those partial grids are summarized through their number |N| of nodes and their number |A| of arcs. Additionally, each arc's length is randomly ranged from 0.5 to 2, with a normalized maximum speed of specified $V_{Max_e} = 1$. Function Φ is taken as function $u \to \Phi(u) = u^2$. Next, function Π^e are generated by fixing a time horizon T_{Max} , fixing a mean number *B* of break points

 t_i^e in $[0, T_{Max}]$, and an average value Δ for value $\Phi(u)$. More precisely, $\Pi^e(t_i^e)$ values are generated within a finite set $\{2\Delta, 3\Delta/2, \Delta, \Delta/2, 0\}$. As for threshold $Risk_{Max}$, we follow a pair (o, d) related path Γ with length L_{Man} , where L_{Man} is the corresponding Manhattan length between o and d. Then it comes that $Risk_{Max} = L_{Man}\Delta$. Additionally, we calculate the *Mean Risk*, which is the average of all the risk functions in the graph G = (N, A). The related presetting parameters in instances can be found in Table 3.4:

Instance	N	A	В	Δ	L_{Man}	Mean Risk
1	20	65	1	2	44.54	5.73
2	18	65	2	1.5	59.03	2.85
3	19	65	1	3	47.15	5.64
4	54	160	1	4	42.3	11.12
5	52	160	2	2.5	71.66	5.75
6	51	160	1	3.5	64.23	4.56
7	82	250	2	3	77.57	5.73
8	83	250	3	3	139.39	5.67
9	88	285	2	4.5	87.27	5.71
10	92	285	3	5	82.68	8.51

Table 3.4: Characteristics of the instances

For each instance, the following comparative studies were conducted, involving three algorithms:

- **LS-SPR algorithm:** computing the Risk value R_{LS} , the Time value T_{LS} in Algorithm 2, the number of iterations *ITER* of its main loop (modification of Γ) and related CPU time CPU_{LS} .
- Greedy algorithm: the Risk value R_{GR} , the Time value T_{GR} computed by the greedy algorithm GR_{SPR} , together with related CPU time CPU_{GR} .
- A^* like algorithm: almost exact Risk value and Time value R_{Opt} , T_{Opt} computed by the A^* like algorithm, performed with large *S* and *K* values, together with related CPU time CPU_{A^*} .

Obtained results are summarized in the following Figure 3.14 and Figure 3.15:

The A^* like algorithm provided nearly accurate Risk and Time values R_{Opt} , T_{Opt} in each case. Thus, as seen in Figure 3.14, the proposed LS-SPR method's results are near to the optimal risk and time values and outperforms the Greedy algorithm in terms of path Γ 's time consumption. Results obtained through GR_{SPR} are rather erratic, because this algorithm relies on the current state of shortest path Γ from o to d, which can be bad at the travel time when we did not launch the *update* step. A^* tends performs better than LS-SPR as for the accuracy, but is more time consuming for algorithm running (see Figure 3.15). Depending on the cases, results of A^* may be also significantly impacted by parameters values S and K. Finally, we also notice that obtaining almost exact optimal values is rather time costly, even on small instances. In order to improve it, we should find a way to provide a criterion which could identify, at any times, whether a decision λ_e^{RD} has to be tried or not.



(a) Risk value for every instance.



(b) Time value for every instance.



Figure 3.14: Risk versus time for every instance.



(b) Related CPU time by LS-SPR and Greedy algorithm.

Figure 3.15: Reference CPU computing time.

The simulations reveal that our proposed LS-SPR may give a pretty good solution within an ideal time-risk trade-off. The computing time is rather quick, and the method appears to be suitable for use in a real-time scenario. However, it is necessary to emphasize that the risk function must still be demonstrated empirically and statistically, which are not included in this PhD thesis. We simplified the instance by using a fixed break point and maximum risk value. Maintaining the same assumptions in complicated urban navigation contexts is challenging. Section 3.3 may be viewed as an exploratory approach to highlevel modeling and decision making, which is also a goal of this PhD thesis. Nonetheless, due to the model core difficulty (uncertain urban environments), the following chapter's focus will be on management-level, which will not use the suggested LS-SPR method.

3.4/ CONCLUSION

This chapter proposed a distributed optimal approach for dynamic MVS CN with risksensitive management strategy. A distributed architecture of MVS system has been formulated. The proposed formulation employs the PC theory, which enables a competitive solving time $(0.2s \sim 0.8s)$ by approximating the optimal solution in real-time applications. The TSB communication mechanism was adopted for bilateral or multilateral negotiation. Furthermore, a key safety indicator 2D TTC was adopted as a risk assessment measure to impact vehicle's decision. We model the 2D TTC in two dimensions and assume that the minimum 2D TTC value in MVS is the corresponding objective function. Such an application with the proposed ε -PC lead to promising results. Particularly, the proposed ε -PC can be applied in trajectories tracking, maneuvers warning/assisting or directly as feedback control law in Model Predictive Control (MPC). The experiments shown in this chapter prove the efficiency of the proposed ε -PC framework for real-time intersection management. Additionally, this chapter develops a local heuristic search algorithm called LS-SPR for obtaining an approximate optimal solution to the SPR problem involving a trade-off between time and risk. A learning process for LS-SPR with a limited number of decisions and states is discussed, and experiments demonstrate that it is possible to get a solution that is close to the optimal value while saving computational time. However, given the more difficult settings associated with urban traffic management, our assumptions regarding a constrained transit network (a warehouse area) may not be valid. As a result, it will not be applied in the subsequent chapter. Generally, the issues discussed in this chapter include MVS cooperation at the micro-control level and routing strategy from the perspective of a global supervisor. Both approaches for motion planning can be extended to the transportation network. However, when things get more complicated, a hierarchical design needs to be dealt with a reliable and practical way. In the following chapter, we will go through these issues in further details.

4

MIMAFC-BASED HIERARCHICAL TRAFFIC MANAGEMENT ARCHITECTURE IN TRANSPORTATION NETWORKS

This chapter considers vehicle decision-making in a transportation network with multiple intersections. In fact, advanced intersection control systems have been created to alleviate traffic congestion (cf. Chapter 2 for a literature review). Regular MVS using Cooperative Navigation (CN) principles make it easier to accomplish a shared purpose than traditional traffic. Nonetheless, due to whole uncertainty in transportation network, conventional motion planning for local areas may lead to undesirable effects in long run. Previous work in Chapter 3 developed a safe and flexible CN strategy for resolving local conflicts at an unsignalized intersection. This chapter will design a Micro-Macro Flow Control (MiMaFC) technique for investigating MVS' navigation at unsignalized intersection networks. More precisely, it is intended to get a better understanding of micro control and how it might be used to influence the behavior of traffic flows based on the proposed CN strategy.

As mentioned before, our work will employ hierarchical architecture (cf. section 4.2) to investigate the global navigation performance of MVS in complex environments/situations. Next, a macroscopic flow model for interpreting the real-time status of a dynamic transportation network is provided in section 4.3. Correspondingly, CN protocol used in the addressed architecture is specifically developed for MVS that continually cross intersections (cf. section 4.4). Further, spatio-temporal velocity adaptation is presented in this work. Depending on the vehicles' location and speed, MVS might use either the MiMaFC-based or self-interested speed approach. Simulation results (cf. section 4.5), which include a congested traffic network, are shown to demonstrate the suggested method's potential. It finds that the suggested motion planning framework may increase urban network mobility over a pure, distributed, non-supervised MVS system.

4.1/ BACKGROUND AND CHALLENGES

Urban transport systems are expected to be enormously improved thanks to the connected Intelligent Vehicles [8, 275]. The efforts made by the academic institutions, car manufacturers and Big Tech companies permit to avoid the in-road hazards through MVS [166], improve fuel economy and reduce emissions [175] by more efficient CN technologies. Under this circumstance, a review research work presented in [331] investigated the "ripple effect" of the vehicle automation. This work has built a ripple model to classify the implications of vehicle automation (the core of the ripple) on three levels. In the first level, the travel cost/choice and traffic capacity are emphasized for the near future. It will directly impact the vehicle ownership and sharing, residential location choices, land use and transport infrastructures as the sequential effects in the second level. Finally, the third level is related to broader social challenges such as energy use and pollution and public health, etc. As a result, it's critical to create modern traffic management systems that can adapt to the process of autonomous driving and pay attention to its sequential effects. Obviously, MVS may contribute in a better manner to boost the public transportation in urban areas and regulate their navigation in an arranged way [124, 275, 363]. From this perspective, an important question arises: how can MVS help to fulfill the increasing mobility demands and better adapt intelligent transportation development in the future? Therefore, developing novel mobility platforms, which are focusing on improving the arterial traffic performance, is attracting a considerable interest in the transportation systems community.

As illustrated in section 1.1.1, the majority of studies indicates that MVS will become a reality in the near future (although at very modest market penetration rates) and will have the capacity to significantly decrease traffic congestion, road accidents, and car emissions [127, 363, 377]. Additionally, research on autonomous highways has proven significant promise in a variety of traffic scenarios, facilitating the deployment of MVS [428]. In high-ways/instrumented intersections, a number of detecting technologies are used, including in-road sensors (loop detectors, magnetic detectors, and magnetometers) and over-roadway sensors (cameras, radars, ultrasonic, infrared, and acoustic sensors) [82, 163, 299]. Intersections may be instrumented to broadcast messages to neighboring cars through DSRC (SAE standard J2735-Dedicated Short Range Communications message set dictionary), which contains information about the signal phase, time and geometry of the intersection [181]. Due to advancements in both the on-board system and sophisticated sensors in surrounding environments, research under the premise of all MVS on the road has also attracted the traffic control community's attention. Most urban road users feel that implementing MVS technologies (such as cooperative trajectory prediction and motion planning) will considerably increase urban road capacity at a corridor or a road network level in a fully automated setting [24, 241, 244, 519]. In this sense, these MVS applications address comparable traffic-control issues. In [552], researchers used game theory to an unsignalized intersection, suggesting that an intersection controller collect data from automated vehicles on the road network and minimize overall trip delay. A decentralized Model Predictive Control (MPC) approach provided by [316] is used to determine the time necessary to arrive at an intersection for each vehicle in the network. A survey related to a review of traffic control and its links to vehicle connectivity can be found in [41] and [284]. In particular, many decentralized approaches for MVS CN still requires a local manage agent in charge of synchronizing the vehicle to cross the intersection [445]. It is important to note that the dynamics and distributed controller are qualities of the individual MVS, but the information flow network and macro traffic control are aspects of the entire MVS system. Consequently, most of the research on on-ramp/intersection coordination in road networks use a multi-layer/multi-level control framework [406, 446, 498].

In summary, there are two main challenges that need to be addressed in the context of MVS CN in a complex intersection network (non-highway scenarios). First, how to create a specialized large-scale optimization framework for addressing the issue of largescale vehicle operation. Due to the fact that unexpected driver behaviors often contribute to traffic uncertainty, cars are prone to unnecessary delay and, as a result, slow traffic [195]. Additionally, increased traffic data must be analyzed inside a MVS system. There is no established concept, standard or platform for resolving this problem at the moment. Rather than that, a framework that unifies the protocol for CN. Second, we are concerned not only with the safe movement of a single agent inside MVS, but also with the coordinated and safe movement of the whole system. Nonetheless, in the expanded macro traffic flow model, the cooperative motion planning methods remain inexact and suboptimal. Additionally, the trajectory chosen to resolve conflicts in a local region may result in an increase in travel time and fuel consumption over time. As a result, speed guidance (at a targeted vehicle speed or a constrained speed) is required to account for macro-traffic circumstances.

4.2/ MIMAFC-BASED HIERARCHICAL ARCHITECTURE

The overview of the addressed circumstances is explained in Fig. 4.1. This chapter assumes that all navigation vehicles can interact with one another and with the infrastructure via communication. The designated paths of MVS have been computed based on the state information from global supervisor. A module named local supervisor (S_{Loc_A}) is located at each intersection region. Additionally, assume that downstream traffic information are provided by roadside sensors implanted along the mid-blocks between two intersections. For the sake of simplicity, a S_{Loc_A} is assumed to receive updated approaching traffic flow data without considering measurement errors (and/or delays) induced from the Infrastructure-to-infrastructure (I2I) communication. In the micro control level, it is thought



Figure 4.1: Urban road network for autonomous vehicle CN with local supervisor S_{LocA} at each intersection region.

that all interactive vehicles might make decisions after communication with neighboring vehicles. (see Fig. 4.1). Vehicles originating from various points of entry into the urban network are assigned to predetermined destinations. Further, S_{Loc_A} is assumed to obtain the updated approaching traffic flow messages without considering measurement errors and delays.

4.2.1/ SUPERVISED TRAFFIC MANAGEMENT SCHEME.

As mentioned in section 2.2.2. The current tendency is toward the establishment of a hierarchical supervisory architecture with two or three levels for managing such a system. The following summarizes the roles of the newly established global/local supervisor and MVS in this Chapter.

- Global supervisor: at the macro level, the global supervisor is responsible for tactical dynamic planning and routing of the vehicle, as well as task allocation among the vehicles (e.g., pick-up and delivery). In section 3.3, we developed a transit network model and demonstrated how to balance speed and risk on the road using the Local Search (LS) heuristic algorithm for Shortest Path under Risk constraint (SPR) problem (i.e., LS-SPR algorithm). In contrast, this chapter focuses mostly on a management-level agent that is located at traffic intersections. To simplify the scheme, the global supervisor might still get all information while assigning just the vehicle's path.
- Local supervisor: local supervisors *S*_{LocA}, at the management level, are responsible for supervising minor difficult locations, including as intersections (or loading/unloading zones in warehouse-like areas). As a mediator, it communicates with the vehicles, regulating their transit and avoiding collisions as they pass through specific zones. This level has sparked renewed interest in recent years (see [118, 391]), and occasionally caused confusion with the micro control level: in many cases, the hypothesis is made that all involved vehicles are run by the same embedded software and exchange perfect information; this becomes equivalent to supposing the existence of a local external mediator. Clearly, in an urban environment, the goal of *S*_{LocA} is to prevent congestion and car collisions. However, it can only be provided partial information about the traffic status.
- Multi-Vehicle System (MVS): the MVS or CAVs is defined at the micro-level, by the monitoring and sensing devices installed inside the vehicles; these devices calculate real-time paths and modify them to the presence of potential obstacles. The majority of the robotics community's attention is still directed at the micro-level, where optimum control and artificial perception approaches are predominantly utilized [67, 288, 398]. As a component of the transportation system, the MVS prefers the adaptable and safe CN. Additionally, proactive motion planning is foreseen in our planned system, which facilitates traffic flow fluency.

4.2.2/ HIERARCHICAL AND HYBRID DECISION-MAKING ARCHITECTURE

The proposed Multi-layer Hybrid Control Policy and Motion Planning (MHCP-MP) framework is shown in Fig. 4.2. The MHCP-MP architecture is divided into two primary layers for each intersection: a macroscopic flow model and a trajectory-based optimization



Figure 4.2: Basic schematic of the proposed Multi-layer Hybrid Control Policy and Motion Planning (MHCP-MP) framework.

model for AIM. Local supervisors S_{Loc_A} observe the 4-way downstream traffic flow status. Thus, the traffic aggregated speeds and the rights of passage are then disseminated for upstream vehicles¹ in different directions. MVS are therefore considered to have onboard system to retrieve a hybrid control policy from S_{Loc_A} . It is important to note that all the approaching MVS in the 4-way upstream are investigated as a distributed cooperative system based on their network topology structure. Additionally, the microscopic strategies and optimization solver are addressed by a previously introduced ε -Probability Collective (PC) algorithm [1, 2]. In so doing, the MVS system will guarantee an optimal (or suboptimal) trajectories-based planning for lower control layer. Finally, the CN for multiple MVS systems is conducted regarding the traffic flow fluctuation while ensuring locally efficient and safe navigation. The main idea standing behind the proposed architecture is to construct a feasible link between the developed macroscopic flow strategy and the microscopic motion control as seen in the prior work [4]. A more detailed explanations of the macroscopic flow model regarding urban network (cf. section 4.3) and MVS CN method based on MiMaFC policy (cf. section 4.4) will be provided in the following sections.

4.3/ MIMAFC STRATEGY IN TRAFFIC FLOW MODEL

The macroscopic flow model interprets the real-time state of the dynamic transportation network including multiple intersections. Firstly, a primitive car-following model is illus-trated by ordinary difference equations in section 4.3.1. Next, the urban road network is explained in section 4.3.2.

¹Vehicles that will cross imminently the intersection (cf. Fig. 4.1).

4.3.1/ FORMULATION OF VEHICLE DRIVING MODEL

The vehicle mobility is developed referring Krauss model [259] where each vehicle can perform two motion states (i.e., free motion and interacting motion). Firstly, the proposed car-following model is illustrated by Ordinary Difference Equations (ODE) like follows:

$$\begin{cases} a_i(t) = u_i(t) + \varepsilon_i(t) \\ v_i(t + \Delta t) = v_i(t) + a_i(t) \times \Delta t \\ x_i(t + \Delta t) = x_i(t) + v_i(t) \times \Delta t + \frac{1}{2}a_i(t) \times \Delta t^2 \end{cases}$$
(4.1)

Where $x_i(t)$ and $v_i(t)$ denote respectively the displacement and velocity of the vehicle *i* at time instant *t*. $a_i(t)$ is the corresponding acceleration for a time interval Δt . Besides, $a_i(t)$ in (4.1) is addressed by the control input $u_i(t)$ and an uncertain disturbance factor $\varepsilon_i(t)$ which is related to the perception and sensing errors. It is supposed that $u_i(t)$ corresponds to a movement of a particle with constant acceleration between two instants, which has been defined by Δt . Thus, considering the relative distance $\Delta x_{i,i-1}(t) = x_{i-1}(t) - x_i(t)$ and relative speed $\Delta v_{i,i-1}(t)$ between two successive vehicles (i.e., ego vehicle *i* and vehicle i - 1 ahead). It assumes that MVS either perform Cruise Control (for free motion) to maintain a preset speed v_{ref} or Adaptive Cruise Control (for interacting motion) when a vehicle ahead is detected within a distance $\Delta x_{i,i-1}(t) \leq R_w$. R_w is assumed to be a fixed sensing range. Thus, a reference distance d_{ref} is defined for Adaptive Cruise Control:

$$d_{ref} = d_{safe} + \Delta x_{i,i-1}^* \tag{4.2}$$

In (4.2), d_{safe} is the preset standstill safe distance. $\Delta x_{i,i-1}^*$ is the desired distance at current speed. Thus, the control law $u_i(t)$ for vehicle *i* can be addressed in (4.3):

$$u_{i}(t) = \begin{cases} -k_{0} \cdot (v_{i}(t) - v_{ref}) & \text{if } \Delta x_{i,i-1}(t) > R_{w} \\ k_{1} \cdot (\Delta x_{i,i-1}(t) - d_{ref}(t)) + k_{2} \cdot \Delta v_{i,i-1}(t) & \text{if } \Delta x_{i,i-1}(t) \le R_{w} \end{cases}$$
(4.3)

Where $\{k_0, k_1, k_2\}$ are positive control gains. It is important to remark that $\Delta x_{i,i-1}^*$ represents the preferred distance for each vehicle in (4.2). MVS can be assigned with stochastic space policy like human driver applying invasive or conservative following strategy on road. In this chapter, the desired distance $\Delta x_{i,i-1}^*$ is defined by the stochastic time headway th_i : $\Delta x_{i,i-1}^* = th_i \cdot v_i(t)$. Further, it is assumed that th_i is sampled based on a shifted log-normal distribution as introduced in the literature [287]. Therefore, the i^{th} vehicle is supposed to generate an independent and identically distributed (i.i.d.) $t\hat{h}_i$ as self-preferred time gap (i.e., $th_i = t\hat{h}_i$). Consequently, $t\hat{h}_i$ is specified by (4.4):

$$t\hat{h}_i \sim \text{Log-}N(\mu_v, \sigma_v)$$
 (4.4)

Where μ_{ν} and σ_{ν} are the predefined velocity dependent parameters in log-normal distribution (cf. appendix B). These parameters can be estimated using statistical inference by empirical datasets like the Next Generation Simulation (NGSIM) [461] described in section 4.5.

4.3.2/ FORMULATION OF TRAFFIC FLOW FUNDAMENTALS

In this section, in order to have enough representative example, we will consider an area of 9 neighborhood intersections which are combined together as an urban road network



Figure 4.3: CAVs navigation in transportation network with Multiple unsignalized intersections: the origins (destinations) are noted by $O_i(D_i)$. Each 4-way intersection is in charged by a local supervisor S_{LoAi} . The 3-tuple of traffic density, velocity and flow rate are defined for each link { $K_{L_i}, V_{L_i}, Q_{L_i}$ }, destination lane { $K_{D_i}, V_{D_i}, Q_{D_i}$ } and intersection area { $K_{S_i}, V_{S_i}, Q_{S_i}$ }.

like in Fig. 4.1. The whole transportation network contains 48 links and as mentioned above 9 junctions. The Origins and Destinations (O-D) are set to manage the flows input/output at the borders of the traffic network (see Fig. 4.3). Clearly, destinations points are located in 12 links (thus, define the destination links number: $n_D = 12$) which are not belong to any of the internal intersection areas (red lines in Fig. 4.3). In such neighborhood-sized sections of urban area, it is assumed that the traffic load is homogeneously distributed at the initial state. Further, the external conditions especially for the time-dependent traffic flow are supposed to change slowly. In addition, the traffic flow characteristics linking flow, speed and density can be uniformly defined and revealed by what is called Macroscopic Fundamental Diagram (MFD) (cf. Fig. 4.4 developed in our prior work [3]). For exploiting the MFD, interested reader may refer to [158, 247, 315].

There are many ways to interpret the value of fundamental traffic flow characteristics (flow, speed and density). Here, the measured space mean speed V, traffic density K and calculated travel flow Q are addressed in this section (see Fig. 4.3 for example). Generally, we consider in the proposed modeling that the infrastructure has the possibility to cyclically collect instantaneous vehicle speed v_i and vehicle number N_i for each lane. Thus, the lane density at every instant can be defined as follows:

$$K_{L_i} = K_{D_i} = \frac{N_i}{LGTH_i} \tag{4.5}$$

Where $LGTH_i$ is the length of a link, K_{L_i} represents the lane's density towards the intersection (black arrows in Fig. 4.3) and K_{D_i} is the destination lane's density (red arrows in Fig. 4.3). Accordingly, the density of intersection K_{S_i} (combined by n_{L_i} links, for instance,

a group of $n_{L_i} = 4$ yellow arrows in Fig. 4.3) is defined as:

$$K_{S_i} = (\sum_{L_i=1}^{n_{L_i}} K_{L_i})/n_{L_i}$$
(4.6)

Moreover, the overall intersections density K_S and exits lane density K_D are developed as follows (n_S and n_D are respectively the overall number of intersections and the number of destinations):

$$\begin{cases} K_{S} = (\sum_{i=1}^{n_{S}} K_{S_{i}})/n_{S} \\ K_{D} = (\sum_{D_{i}=1}^{n_{D}} K_{D_{i}})/n_{D} \end{cases}$$
(4.7)

The space-mean speed¹ has been more commonly adopted to reveal current traffic state than time-mean speed² (which overestimates the influence of faster vehicles [255, 453]). Hence, the space-mean speed, which is also calculated as a harmonic mean of collected vehicle speeds, is used in this section as the traffic velocity associated with a specified length of roadway. Thus, the traffic velocity for flow input lane, exit lane and intersection area are respectively defined as { V_{L_i} , V_{D_i} , V_{S_i} }, see (4.8):

$$\begin{cases} V_{L_i} = N_i / \sum_{i=1}^{N_i} (1/v_i) \\ V_{D_i} = N_{D_i} / \sum_{i=1}^{N_{D_i}} (1/v_i) \\ V_{S_i} = N_{S_i} / \sum_{i=1}^{N_{S_i}} (1/v_i) \end{cases}$$
(4.8)

Where $\{N_i, N_{D_i}, N_{S_i}\}$ are the vehicle quantity in the corresponding areas. Similarly, the traffic velocity for overall intersections (contain N_S vehicles) and destination lanes (for N_D vehicles) are written as:

$$\begin{cases} V_D = N_D / \sum_{i=1}^{N_D} (1/v_i) \\ V_S = N_S / \sum_{i=1}^{N_S} (1/v_i) \end{cases}$$
(4.9)

In such a manner, the calculated flow rate can be also addressed as $Q_{L_i} = K_{L_i} \times V_{L_i}$ (for flow input lane), $Q_{D_i} = K_{D_i} \times V_{D_i}$ (for destination lane) and $Q_{S_i} = K_{S_i} \times V_{S_i}$ (for each intersection area) at every instant (see also in Fig. 4.3). In so doing, we avoid to hourly measure flow rate which reflect the equal traffic knowledge as calculated flow rate in the experimental intersection network.

The current proposed work aims to develop proactive vehicle motion planning regarding the dynamic changes in urban environments between controlled intersections. Therefore, the study develops a global supervisor to broadcast the aforementioned traffic characteristics to the preassigned S_{Loc_A} on a regular basis. A global supervisor observes the status of traffic and encourages intelligent vehicles to take specific routes, as shown in the urban setting in Fig. 4.1. It is also compatible with a GPS device, which can suggest the best route (consuming shortest travel time) to get to the targeted destination [38]. Next, the obtained traffic key factors are formulated to the macroscopic fundamental diagram (see

116

¹The space-mean speed is the average speed of vehicles traveling a given segment of roadway during a specified period of time and is calculated using the average travel time and length for the roadway segment (i.e., $S paceMeanS peed = \frac{distance traveled}{avg. travel time}$ as seen in [453], Chapter 1).

²The time-mean speed is defined by the arithmetic average speed of all vehicles for a certain duration of time.



Figure 4.4: The Macroscopic Fundamental Diagram (MFD) of theoretical correlation between traffic density and traffic flow.

Fig. 4.4) by the notable Greenshields model [176] for every single lane as follows:

$$\begin{cases} \mathbb{V}_{L_{i}} = V_{f}(1 - \frac{K_{L_{i}}}{K_{jam}}) \\ \mathbb{Q}_{L_{i}} = V_{f}(K_{L_{i}} - \frac{K_{L_{i}}^{2}}{K_{jam}}) \end{cases}$$
(4.10)

where V_f and K_{jam} are respectively the free-flow speed (maximum desired speed) and jam density (where automobiles are unable to move) regarding the current road conditions. $[\mathbb{V}_{L_i}, \mathbb{Q}_{L_i}]$ are the dependents on the measured lane density K_{L_i} in the proposed traffic fundamental diagram (cf. Fig. 4.4). It is worth acknowledge that the maximum road flow $\mathbb{Q}_{L_i}^{max}$ is theoretically defined at the vertex of the parabola in Fig. 4.4 where $\partial \mathbb{Q}_i / \partial K_{L_i} = 0$ in (4.10). If the vehicle number increasing consistently (while $\partial \mathbb{Q}_i / \partial K_{L_i} \leq 0$) till to the maximum road capacity, the flow rate will decrease and even collapse to zero at the jam density K_{jam} .

4.3.3/ FORMULATION OF MIMAFC STRATEGY

Ultimately, MVS are supposed to adopt appropriated actions to ensure a proper spatiotemporal strategy at intersection areas w.r.t. the policy info from the local supervisor S_{Loc_A} as depicted in Fig. 4.2. When a car driving at the decision-making points (bounds of a local area), the crossing policy is assigned for each one including the downstream lane's aggregated speed (flow speed) \mathbb{V}_{L_i} referring to (4.10) and the corresponding "road-weights" W_{r_i} referring to upstream density. The road-weights W_{r_i} for each upstream approaching car from the lane *i* can be defined as (4.11):

$$W_{r_i} = \left(\frac{K_{L_i}}{K_{S_i}}\right)^{\sigma_{S_i}} \times \varphi_{r_i} \tag{4.11}$$

Where:

$$\varphi_{r_i} = \begin{cases} \Pi^{e1}(t), \text{ if } \partial Q_i / \partial K_{L_i} > 0\\ \Pi^{e2}(t), \text{ if } \partial Q_i / \partial K_{L_i} \le 0 \end{cases}$$
(4.12)

Note that $[\sigma_{S_i}, \Pi^{e_1}, \Pi^{e_2}]$ are independent model parameters. σ_{S_i} is designed to amplify the W_{r_i} impacts in the corresponding intersection. The piecewise-defined function φ_{r_i} indicates the free or congested traffic flow state w.r.t. the calculated flow-density ratio $\partial Q_i / \partial K_{L_i}$. Every function $t \to \Pi^e$ is constant whereas Π^{e_2} is greater than Π^{e_1} to significantly increase the congested road-weights W_{r_i} among others. To summarize, the macroscopic flow model (implemented by an intersection supervisor S_{Loc_A}) was created primarily to monitor traffic conditions and transmit relevant intersection crossing policy. In fact, the suggested crossing policy, which is based on the aggregated speed and road-weights of lanes (i.e., { W_{L_i}, W_{r_i} }), will be used for CN strategy at the microscopic motion planning level (cf. section 4.4.2).

4.4/ MIMAFC STRATEGY FOR CN

A systemic approach is implemented in this section to deal with consecutive vehicles' CN at every single intersection (see Fig. 4.5). Indeed, our research focuses on both macrotraffic management and micro-vehicle decision-making, a complicated system in which it is hard to focus on every element. In order to better comprehend such a system without splitting it into much more independent sub-systems (for a more analytical program), the systemic approach is used to analyze all attributes of the full cooperative system under the supervisory scheme. Thus, this latter (for a clearer picture of the microscopic layer in the MHCP-MP framework) is divided into two succeeding parts to highlight the proposed standardized navigation protocol (cf. section 4.4.1 as seen in Fig. 4.6) and local conflicts processing model (cf. Section 4.4.2).

4.4.1/ PROPOSED INTERSECTION NAVIGATION PROTOCOL

As seen in Fig. 4.5, before reaching the Initial Decision-Making Point (IDMP) P_0 , a car approaching the intersection will decelerate at a specific distance S_1 (e.g., $S_1 = 50m$ which corresponds to the bounds of local interactive area). When a vehicle arrives at IDMP P_0 , a local supervisor S_{Loc_A} (cf. Fig. 4.5) immediately sends the local policy, which instructs the vehicle's cooperative motion planning. It's also worth noting that cars arriving at the IDMP P_0 are supposed to have all of the information they need to make decisions (including other vehicles' predicted paths). As a result, if two or more cars join IDMP P_0 (from different directions) at the same time, a local supervisor S_{Loc_A} will appoint the vehicle that makes the decision first at random. The interactive area is divided into three parts labelled by different colors in Fig. 4.5. If a vehicle is identified as a cooperative agent, it will firstly run the MiMaFC-based speed strategy (cf. section 4.4.2.1) which creates the most chance to satisfy the S_{LocA} policy relating the traffic aggregated speed (i.e., the proposed traffic flow velocity). Unfortunately, the crossing strategy is sensitive to the initial speed and it may be re-planned in the proposed cooperation protocol as mentioned in the following paragraph. Thus, MiMaFC-based Decision-Making Area (MiMaFC-DMA,



Figure 4.5: Individual vehicles approach the intersection and make decision from IDMP P_0 to participate in CN. The interactive area supervised by S_{LoA} is divided into three parts (i.e., MiMaFC-DMA, IND-DMA and CA identified by different colors), each of which implements different corresponding navigation protocol for cooperative strategy.

the green area in Fig. 4.5) is reserved with a length S_2 for vehicles to find the bestsampled strategy to compose a feasible cooperative trajectory at other DMP P_i (as seen in Fig. 4.5). Secondly, when a vehicle still can not find a feasible solution to satisfy the local policy (or have to recompute a solution for CN) after MiMaFC-DMA, it will adopt the selfinterested strategy (cf. section 4.4.2.1) in the INDependent Decision-Making Area (IND-DMA, the yellow area in Fig. 4.5). In so doing, safety solution can be lastly guaranteed when a vehicle closing to the intersection Core Area (CA) with red color in Fig. 4.5. In such a case, the S_{LocA} policy can not be fully satisfied. But the vehicle's priority (road weights) are still considered in the optimized trajectory. Thirdly, after the Final Decision-



Figure 4.6: Flow-chart showing the proposed cooperative strategy for traffic network navigation diagram. Making Point (FDMP) P_f (see Fig. 4.5), vehicles in CA are not permitted to modify the targeted speed profile. Namely, vehicles in CA that have already calculated a sufficiently safe trajectory will be excluded from any optimization.

A basic CN protocol in the local supervised area (see Fig. 4.5, for example) has been addressed in our previous work [4] for unidirectional traffic flow or in a single arterial road. The main contribution in this work is to develop reliable and adaptable measures to counter more complicated traffic flows (e.g., left and right turning is possible) in multiple intersection networks. The overall traffic network intersection navigation diagram is provided in Fig. 4.6 for more clearly understanding. Through this framework the augmented cooperative protocol which integrally considers both vehicle's conflicts and multi-intersection management at the level of large area in urban environment is proposed in this section. The suggested protocol, in particular, are customized for various regions (i.e., highlight by different colors in Fig. 4.6) within the supervised interactive area. In the following sections, we will further describe the primary contents and underline adapted zones of the augmented protocol.

4.4.1.1/ PREPARATION AREA: VEHICLE SORTING

Firstly, a new car added in the MVS system executes sorting algorithm (cf. *Algorithm 5*) to identify the interactive vehicles. The cooperative vehicles need to further choose their intersection strategies (either to search a feasible optimal trajectory or help other vehicle to avoid conflicts). Notably, the preceding entering vehicle that owns the assigned crossing strategy usually does not cooperate with a new arriving vehicle until there is a conflict that cannot be avoided by the succeeding cars' own efforts. Particularly, vehicle sorting occurs regularly (at a short interval, for instance 0.1s) in all interactive areas for MVS supervised by S_{LoA} .

A sorting algorithm (as shown in *Algorithm 5*) is firstly performed to identify the interactive vehicles at IDMP P_0 (see Fig. 4.5). Let us assume that the embedded motion planner of each vehicle in the distributed MVS system *SYS* can update the coordination state at every instant. Then, the Boolean's values are correctly assigned for the labeled states such as: collaboration flag V_{Col} , optimization flag V_{opt} , conflict flag $V_{conflict}$ and remain in intersections flag V_{rem} , etc. When a vehicle equipped with an embedded system enters a local monitored area, it will initialize all of these default flags (at IDMP P_0). The detailed steps to distinguish between the collaborative and non-collaborative vehicles are given in *Algorithm 5*.

4.4.1.2/ DECISION AREA: SELECTING STRATEGY

Secondly, the collaborative vehicle calculates its minimum Time-To-Collision which is a risk indicator to describe the remaining time for a probable collision between any two vehicles. The developed 2D TTC in our prior work [2] (this article focuses on developing risk-sensitive intersection cooperation strategies) is revisited to identify the cars that have potential conflicts with other vehicles. Noting that, a threshold of TTC_{min} is used to select the violated 2D TTC. A vehicle that does not hit the minimum threshold TTC_{min} will execute a constant accelerating strategy by predefined a_{ref} (if it is the only vehicle) or maintain current speed (if there are other vehicles). Notably, the risk valuing mechanism that is first implemented at IDMP P_0 is capable of handling a trivial situation (e.g., not too

122

Algorithm 5: Sorting algorithm for collaboration /* Sorting algorithm is executed periodically in supervised areas */ **Input:** SYS, V_{opt}, V_{conflict} and V_{rem} **Output:** V_{Col} */ /* Initial $V_{opt}, V_{conflict}, V_{rem}, V_{Col}$ are set to the default value 0 1 if vehicle in local area then for all $i \in SYS$ do 2 if $V_{opt} == 0$ then 3 // The vehicle without optimal strategy should 4 $V_{Col} = true$; participate in collaboration else 5 $V_{Col} = false;$ 6 7 if $V_{conflict} == 1$ then // The vehicle participates in collaboration 8 $V_{Col} = true$; when an inefficient strategy induced conflict if $V_{rem} == 1$ then 9 // The vehicle participates in collaboration 10 $V_{Col} = true$; when it stays at the local area after predicted time 11 else $V_{Col} = false$ 12 **13** return V_{Col} ;

many automobiles in an interaction) without requiring additional optimization. Additionally, it is conducted on a frequent basis to monitor the collision risk in all local regions. Next, the labelled cooperative agent (i.e., $V_{Col} = true$ after *Algorithm 5*) will run ε -PC algorithm which is impacted by the aforementioned MiMaFC-based speed strategy from S_{LocA} in MiMaFC-DMA with length S_2 (green part in Fig. 4.5). The aggregated speed and lane priority are both considered among a utility maximizing model (cf. Section 4.4.2.3). The feasible solution within a time horizon $T_{horizon}$ will be adopted for the vehicle's control system. Particularly, if a vehicle does not find any solution in former step, it will identify the conflict vehicle in the current state.

4.4.1.3/ IMPLEMENTATION AREA: CONFLICTS PROCESSING

As mentioned before, vehicle in CA (see Fig. 4.5) is not allowed to re-plan their trajectory. In such a context, a vehicle will perform decelerate strategy (i.e., with a constant deceleration) in order to find a better-sampled strategy in the next time instant during it is driving in S_2 . It's worth noting that the car following model continues to operate even when subsequent cars decelerate in MiMaFC-DMA. In addition, the detected conflict vehicles are permitted to do multi-agent PC algorithm immediately, if they are both out of the CA. The MiMaFC strategy from S_{LocA} is still adopted in the cooperation. However, if cooperative vehicles are placed in IND-DMA with length S_3 (which is typically characterized as being shorter than S_2 and closer to CA; see the yellow region in Fig. 4.5), they will solely follow their own self-interest regardless of the local aggregated speed policy. In such an IND-

DMA of length S_2 , it is assumed to prioritize a safe intersection crossing strategy before FDMP P_f (as seen also in Fig. 4.5). Finally, vehicles will either discover an admissible solution to avoid any conflicts corresponding the desired strategy policy (before FDMP P_f) or decelerate to wait a chance to find better sampled-strategy at next DMP P_i belong to MiMaFC-DMA or IND-DMA (as seen in Fig. 4.5).

In summary, the augmented protocol described in this section ensures the safe (nonconflict scene) and/or optimal (conflict scene) operation of subsequent vehicles by allowing them to make decisions using the MiMaFC strategy (from S_{Loc_A} in MiMaFC-DMA with length S_2) or a self-interested strategy (in IND-DMA with length S_3). With regard to the above mentioned protocol for strategy selecting (cf. section 4.4.1.2), the MVS system is required a deliberate effort on approximate optimal solutions integrating the S_{Loc_A} policy and the CN utility. As a result, the following section proposes the spatio-temporal velocity adaptation approach (cf. section 4.4.2.1) for MVS along with an local optimization objective (cf. section 4.4.2.3) to select the proposed strategy.

4.4.2/ DEVELOPING AND EXECUTING CN STRATEGY

The introduced traffic navigation diagram (see Fig. 4.6) is primarily intended to facilitate vehicle cooperation during intersection conflicts. Other trivial cases (non-violation occurred at intersection) are excluded from the cooperative strategy execution. Further, the detected conflict (by risk valuating, cf. section 4.4.1.2) between any two vehicles will be optimized immediately referring to current system states. An intersection with multiple conflicted vehicles is depicted in Fig. 4.7(a). In this section, the aforementioned MiMaFC-based speed strategy and self-interested strategy will be further developed inside the spatio-temporal velocity adaptation approach (cf. section 4.4.2.1). Additionally, an objective function in PC framework (cf. section 4.4.2.3) is discussed in relation to searching the optimal strategy.

4.4.2.1/ VELOCITY-PLANNING-BASED STRATEGY

The conflict resolutions were developed with the aim of producing a low-complexity and rapid optimization strategy for the intersecting network. In fact, the vehicle's path is supposed to be fixed during the movement in the local area. Therefore, the only degree of freedom to re-plan a conflict-free trajectory is the speed for each of the collaborative agents. As seen in Fig. 4.7(a), the vehicles (e.g., the green rectangles) are assigned paths (e.g., the blue arrow) with origins O_i and destination D_i before crossing the intersection. For simplify, a red circle of radius r is defined to surround the car during movement. Any two circles in the 2D graph can not violate a center distance less than 2r when a vehicle follows its path. Therefore, a velocity planning-based optimization problem for MVS is formulated in this section. Particularly, the formulated model only uses the information of the displacements in the path without concerning the path geometry. In so doing, the algorithm is also independent of the topology of the intersection as long as the possible paths are defined. Our previous work in [1, 2] have applied a version of PC algorithm to search feasible solutions with self-interested speed strategy (see Fig. 4.7(b)) in such a single (or adjacent) intersection(s). Readers interested in the creation of self-interested speed strategy are encouraged to consult earlier works [373]. However, the bounded conditions (e.g., the initial speed) are sensitive for MVS system navigation in the pro-



(b) Sampled self-interested speed-time profiles.

Figure 4.7: An illustration of the possible CAVs trajectories with sampled self-interested speed profiles.

posed road network. As a result, the strategy of sampled speed profiles is refined by a spatio-temporal velocity adaption approach in this section.

As mentioned in previous section, a car will decelerate in S_1 until reaching IDMP P_0 (see Fig. 4.5) to identify whether the considered vehicles will participate in the optimization.

See Fig 4.8(a), in the proposed spatio-temporal velocity adaption approach, the collaborative vehicle can firstly choose the actions at IDMP P_0 . If it can not find any feasible solution at IDMP P_0 , the vehicle will go on decelerating regarding previous speed during a specified interval (e.g., after an interval of 0.1s). Thus, the vehicle rerun the optimization after arbitrary time steps at DCM P_i (belong to MiMaFC-DMA) and generate a set of possible speed profiles with lower initial speed. Generally, as seen in Fig 4.8(a), the speed profiles (blue line) have a constraint (red dotted line) of the acceptable range and final target speed (red line). The addressed MiMaFC-based speed strategy \mathbb{V}_{L_i} are defined as the reference speed v_{ref} for each direction. Additionally, the sampled speed must keep constants when entering CA (see Fig. 4.5) to ease the system complexity associated with developing more practical vehicle crossing solutions. Thus, the sampled speed interval can be addressed regarding different constant speeds in CA within the upper/lower bound.



(a) Displacement-velocity profiles in a local area.



(b) Sampled speed-time profiles.

Figure 4.8: A spatio-temporal velocity adaptation approach for CAVs' developing speed strategy at the supervised intersection.

4.4.2.2/ SPATIO-TEMPORAL VELOCITY ADAPTATION APPROACH

The generation of predefined speed profiles from the spatio-temporal velocity adaption approach are inspired by [212, 264]. Indeed, the speed profile set is calculated based on predictive time horizon, as defined by Model Predictive Control (MPC). Thus, a cost function *f* with initial state $x_k = (v_k - v_{ref}, a_k)^T$ (recall that v_k, a_k is the ego vehicle speed/acceleration, v_{ref} is the reference speed for exiting an intersection) is created in this section. Moreover, jerk (links which influence the physiological aspects of the passenger) is denoted as the input signal $u_k \in [u_{min}, u_{max}]$ in *f*. Thus, the running-cost (integral-cost) is modeled as follows:

$$f = \sum_{i=1}^{N_{opt}} \mathbf{x}_{k+i}^T Q \mathbf{x}_{k+i} + \mathbf{u}_{k+i-1}^T R \mathbf{u}_{k+i-1}$$
(4.13)

Where, Q and R are the positive-definite matrix weights to penalize the state error and system input respectively. N_{opt} is the maximum optimization step number after the discretization of the predicted horizon. Assuming that, the step size is defined by Δt . Then, the dynamics model of the proposed system can be explicitly defined as follows:

$$\boldsymbol{x}_{k+1} = A\boldsymbol{x}_k + B\boldsymbol{u}_k$$

$$A = \begin{bmatrix} 1 & \Delta t \\ 0 & 1 \end{bmatrix}, B = \begin{bmatrix} 0 \\ \Delta t \end{bmatrix}, \boldsymbol{x}_k = \begin{bmatrix} v_k - v_{ref} \\ a_k \end{bmatrix}$$
(4.14)

Hence, it is possible to recast the quadratic optimization problem into the whole prediction time horizon with any initial state x_k by introducing the vectors \overline{x}_{k+1} , \overline{u}_k , \overline{Q} , and \overline{R} in the form:

$$\overline{\boldsymbol{x}}_{k+1} = \begin{bmatrix} \boldsymbol{x}_{k+1} \\ \boldsymbol{x}_{k+2} \\ \vdots \\ \boldsymbol{x}_{k+N_{opt}} \end{bmatrix}, \overline{\boldsymbol{u}}_{k} = \begin{bmatrix} \boldsymbol{u}_{k} \\ \boldsymbol{u}_{k+1} \\ \vdots \\ \boldsymbol{u}_{k+N_{opt}-1} \end{bmatrix}, \overline{\boldsymbol{Q}} = \begin{bmatrix} \boldsymbol{Q} \\ \ddots \\ \boldsymbol{Q} \end{bmatrix}, \overline{\boldsymbol{R}} = \begin{bmatrix} \boldsymbol{R} \\ \ddots \\ \boldsymbol{R} \end{bmatrix}$$
(4.15)

The running-cost function f in (4.13) can be rewritten as:

$$f = \overline{\mathbf{x}}_{k+1}^{I} Q \overline{\mathbf{x}}_{k+1} + \overline{\mathbf{u}}_{k}^{I} R \overline{\mathbf{u}}_{k}$$
(4.16)

Further, the state space model in (4.14) is correspondingly formulated as:

$$\overline{\boldsymbol{x}}_{k+1} = \overline{A}\boldsymbol{x}_k + \overline{B}\overline{\boldsymbol{u}}_k$$

$$\overline{A} = \begin{bmatrix} A \\ A^2 \\ \vdots \\ A^{N_{opt}} \end{bmatrix}, \overline{B} = \begin{bmatrix} B & 0 & \cdots & 0 \\ AB & B & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ A^{N_{opt}}B & A^{N_{opt}-1}B & \cdots & B \end{bmatrix}$$
(4.17)

Finally, we substitute (4.17) into (4.16) to reserve only the input matrix \overline{u} by a standard quadratic form:

$$f(\overline{\boldsymbol{u}}_{k}) = \frac{1}{2} \overline{\boldsymbol{u}}_{k}^{T} H \overline{\boldsymbol{u}}_{k} + f \overline{\boldsymbol{u}}_{k} + d_{k}$$

$$H = 2(\overline{\boldsymbol{B}}^{T} \overline{\boldsymbol{Q}} \overline{\boldsymbol{B}} + \overline{\boldsymbol{R}}_{k})$$

$$f = 2(\overline{\boldsymbol{A}} \boldsymbol{x}_{k})^{T} \overline{\boldsymbol{Q}} \overline{\boldsymbol{B}}$$

$$d_{k} = (\overline{\boldsymbol{A}} \boldsymbol{x}_{k})^{T} \overline{\boldsymbol{Q}} \overline{\boldsymbol{A}} \boldsymbol{x}_{k}$$

$$(4.18)$$

Where the quadratic part described by H and linear part described by f will influence the input \overline{u}_k . Therefore, the independent part d_k (constant related to the initial state x_k) in (4.18) can be eliminated to make the objective function f running more compactly. Thus, the proposed quadratic optimization problem can be defined as:

$$\min_{\overline{u}_{k}} f^{*}(\overline{u}_{k}) = \frac{1}{2}\overline{u}_{k}^{T}H\overline{u}_{k} + f\overline{u}_{k}$$
subject to $A_{ineq}\overline{u}_{k} \leq b_{ineq}$
 $A_{eq}\overline{u}_{k} = b_{ineq}$
(4.19)

Where \overline{u}_k belongs to the bounds of $[\overline{u}_{lower}, \overline{u}_{upper}]$ regarding the inequality constrains. The equality constrains are used to enforce the speed to keep constant at CA (see Fig. 4.5). If a car enter the conflict area at i_1 and exit at i_2 where $i_1, i_2 \in [1, \dots, N_{opt}]$, the constraints can be addressed as:

$$A_{ineq} = \begin{bmatrix} \mathbf{I} \\ -\mathbf{I} \end{bmatrix}, b_{ineq} = \begin{bmatrix} \overline{\mathbf{u}}_{upper} \\ -\overline{\mathbf{u}}_{lower} \end{bmatrix}, A_{eq} = \begin{bmatrix} [\overline{CB}]_{i_1,*} \\ [\overline{CB}]_{i_2,*} \end{bmatrix}, b_{eq} = \begin{bmatrix} [-\overline{CA}\mathbf{x}_k]_{i_1,*} + \overline{v}_{ref} \\ [-\overline{CA}\mathbf{x}_k]_{i_2,*} + \overline{v}_{ref} \end{bmatrix}$$
(4.20)

Where:

$$\overline{C} = \begin{bmatrix} C & & \\ & \ddots & \\ & & C \end{bmatrix}, C = [1, 0]$$

$$\overline{v}_{ref} = v_s - v_{ref}$$

$$v_{ref} = min\{\mathbb{V}_L, v_{upper}\}, \quad \mathbb{V}_L \in [\mathbb{V}_{L_1}, \mathbb{V}_{L_2}, \mathbb{V}_{L_3}, \mathbb{V}_{L_4}]$$

$$(4.21)$$

In (4.20), $[\overline{CB}]_{i,*}$ and $[-\overline{CA}x_k]_{i,*}$ stand for the row *i* in matrix \overline{CB} and $-\overline{CA}x_k$. The sampled speed v_s (between the bounds of $[v_{lower}, v_{upper}]$) are defined as the constant speed in CA. The reference speed v_{ref} is defined according to the aggregated speed \mathbb{V}_L in the targeted direction or local speed limit. More precisely, the final speed of the vehicle either tends toward the maximum allowed speed v_{upper} in the intersection or reach the traffic aggregated speed (if $\mathbb{V}_L < v_{upper}$). Therefore, the different solution (i.e., the possible speed profiles) for the quadratic problem are given by various sampled v_s . As seen in Fig 4.8(b), the red plus signs present the time vehicle entering the conflict area by 10 sampled speeds $v_s \in [7m/s, 17m/s]$. All proposed speed profiles are converged to the lane aggregated speed $\mathbb{V}_L = 10m/s$ (red line.) at the end of the time horizon. The next section will further explore the optimal sampled velocity profiles.

4.4.2.3/ MIMAFC-BASED CN STRATEGY

Considering the search space for MVS (a distributed system), the suggested objective function for selecting the optimal sampled speed profile is as follows:

$$J = W_{sep} \sum_{i_{v} \neq i_{self}} \sum_{k=1}^{max} \frac{1}{d_{k}(i_{v}, i_{self})^{2}} + W_{speed} \sum_{k=1}^{max} (v_{k} - v_{end,i_{v}})^{2} + \sum_{i_{v}} W_{cross,i_{v}} T_{i_{v}}$$
(4.22)

In (4.22), W_{sep} , W_{speed} and W_{cross,i_v} are respectively characterized as weights for the vehicle's separation d_k , deviation for reference exit speed v_{end,i_v} and intersection crossing time T_{i_v} . k is the interval indicator computed by the discretization step regarding a predefined time horizon. v_k is the ego vehicle's speed at every k instant. Moreover, the S_{Loc_A} policy is considered by the second term (v_{end,i_v}) and the third term (W_{cross,i_v}) in relation with downstream aggregated speed [\mathbb{V}_1 , \mathbb{V}_2 , \mathbb{V}_3 , \mathbb{V}_4] and "road-weight" W_{r_i} (cf. Section 4.3) for each vehicle like:

$$\begin{cases} v_{end,i_v} = v_{ref}^{i_v} \\ W_{cross,i_v} = W_{r_i}^{i_v} \end{cases}$$

$$\tag{4.23}$$

Where, $W_{r_i}^{i_v}$ is defined through (4.11) in the lane of vehicle i_v and v_{end,i_v} is defined corresponding to the reference speed $v_{ref}^{i_v}$ in (4.21). It is worth noting that the first term in (4.22) is devoted to guarantee a safe spacing between vehicles in an isolated intersection. Besides, the second and third terms are linked to the intersection policy from S_{Loc_A} to achieve the proposed MiMaFC strategy. To do that, the upstream vehicle can acquire a consensus speed at the beginning of the entrance of the downstream traffic flow. Furthermore, the third term in (4.22) is specified to alleviate the congestion upstream. $W_{t}^{t_v}$ will be significant increased value if the upstream (the corresponding downstream of the previous adjacent intersection) traffic flow fall in the congestion state. Under such situation, the vehicles i_{v} in higher density road will ensure more efforts to have a short crossing time. Therefore, the collaborative vehicles in MVS will reserve the preferred trajectory making vehicles i_v own priority to cross the intersection. Finally, such a technique may be used to alter the density of traffic on crowded roads. In particular, our prior work [1, 2] emphasized the benefits of employing the ε -PC optimization framework for modeling and managing distributed systems. The purpose of this research is to demonstrate how the proposed MiMaFC method can be used to improve the performance of the MVS system navigation in intersection networks by modifying both candidate speed profiles and the objective function in the ε -PC framework. The next section's simulation results will provide further information on the aforementioned strategy.

4.5/ SIMULATION RESULTS

128

To illustrate the performance of the proposed two-layer MHCP-MP framework, the Next Generation Simulation (NGSIM) data sets [461] are explored to characterize the stochastic headway distribution in (4.4). Further, the simulations in Matlab considering multiple unsignalized intersections are executed within a computer of Core i7-10750H, $2.60GH_z$ and 16GB RAM. The main parameters adopted in the tackled scenario are summarized in Table 4.1.

The verified scenario can be seen in Fig. 4.9. The overall MHCP-MP framework was run in 3×3 urban road networks. The unidirectional flows arrive from outside of the network according to a Poisson distribution with the default parameter $\lambda = 1.5 veh/s$. All the vehicles were set up with the initial speed 10m/s in the velocity bounds [0, 20][m/s] (as given in Table 4.1). Vehicles on the road were provided with in-vehicle embedded system for running CN algorithm considering hybrid control policy from S_{Loc_A} . An Adaptive Cruise Control (ACC) system is adopted for maintaining a desired reference speed $v_{ref} = 20m/s$ or time headway linking the log normal distribution $\{\mu_v, \sigma_v\}$ (see Table 4.1). To highlight the advantages of the proposed method, this section performs a baseline model

Parameters	Notation	Value	Units
Simulation time	Tend	200	S
Sampled time interval	T _{sample}	0.2	S
Vehicle safe radius	r	3	m
Lane length	L	410	m
Lane speed limit	$[v_{min}, v_{max}]$	[0, 20]	m/s
Bounds on acceleration	$[a_{min}, a_{max}]$	[-3, 3]	m/s^2
Bounds on jerk	$[j_{min}, j_{max}]$	[-2, 2]	m/s^3
Initial speed for all the vehicle	Vinitial	10	m/s
Decision-Making area	$\{S_1, S_2, S_3\}$	{50 35 10}	m
Minimum 2D TTC	TTC_{min}	10	S
The radius of S_{Loc_A}	R	65	т
Road weight parameters	$\left\{\sigma_s,\Pi^{e1},\Pi^{e2}\right\}$	$\{1, 1, 10\}$	-
CC reference speed	v _{ref}	20	m/s
Control gains	$\{k_0, k_1, k_2\}$	{1, 1, 3}	-
Log-normal distribution parameters for ACC	$\{\mu_{v},\sigma_{v}\}$	$\{0.73, 0.52\}$	-
Standstill safe distance	d_{safe}	6	т
Sensing range	R_w	30	т
Strategy number	N_s	10	-
Matrix weight for strategy	$\{Q, R\}$	$\left\{ \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}, \begin{bmatrix} 0 & 0 \\ 0 & 1 \end{bmatrix} \right\}$	-
Prediction horizon	Thorizon	15	S
Weight on exit speed	W_{speed}	1	-
Weight on separation	W_{sep}	10	-
Weight on crossing time	W _{cross}	0.5	-
Max number of iterations	Niteration	100	-

Table 4.1: Parameters adopted in the tackled scenario

which did not include S_{Loc_A} and navigation protocol (cf. section 4.4.1). The baseline model reserve the same kinematic characters as the proposed car-following model on the road. Nevertheless, vehicles without S_{LocA} simply choose the fixed time strategy to reach the target speeds (spanned the sampled velocity range) at the covered area of the intersection.

As seen in Fig. 4.10, the up-left and up-right velocity diagram give a global view of baseline model without S_{Loc_A} and the proposed method with S_{Loc_A} . The desired exit speed in the baseline model is equal to the maximum speed in their strategy set. Although vehicles expect to leave the intersection as fast as possible, MVS have to wait for their turn to participate in the decision-making. In such an approach, the maximum allowed agents can participate the optimization is five. Therefore, the remained MVS in the local area have to slow down until permit to participate in cooperative optimization. Moreover, the initial speed is very sensitive to vehicle's decision-making. A decelerate policy was widely adopted before entering the intersection in the unsupervised MVS system. Simulations show that vehicle decelerate to around 5m/s in order to have the ability to find the maximum admissible crossing strategy in predefined conditions. Nevertheless, due to the speed fluctuation, the desired velocity causally collapsed to 0m/s when the vehicles increase during the second half simulation time.



Figure 4.9: Unidirectional flow of CAVs navigation in traffic network for both supervised and unsupervised system. The red circles stand for the coved range of a local supervisor S_{LoA} noted from 1 to 9 (simulation videos can be found at https://bit.ly/3zcDwAu).

In contrast, MVS adopting the I2V technology can obtain the real time traffic policy by S_{Loc_A} . The vehicle exit speed was expected to be close to the aggregated speed, which can help to harmonize the flow fluctuation. The augmented navigation protocol also wins the chance to find optimal (or suboptimal) crossing strategy at relative high speed. The up-right velocity diagram in Fig. 4.10 shows most vehicles can deal with the CN task when a self speed greater than 10m/s. As a consequence, the proposed MHCP-MP framework including S_{Loc_A} can improve the average velocity (blue line in bottom-right graph in Fig. 4.10) comparing with the whole distribute MVS system's average speed (red line in bottom-left graph in Fig. 4.10). In addition, all the adjacent vehicles keep a safe distance $d_{safe} = 6m$ in the addressed approach. On the contrary, twice violations of inter-vehicle distance are observed in the baseline model as seen in Fig. 4.11. In brief, the overall cooperative MVS system can benefit from the augmented navigation protocol and assigned real-time traffic policy of S_{Loc_A} to guarantee reliable, smooth, and safe running.

The corresponding traffic fundamental diagram for each intersection and the exits of the whole urban network can be seen in Fig. 4.12. The color bar stands for the time in the



Figure 4.10: A Comparison of CAVs velocities between MHCP-MP framework with S_{Loc_A} and baseline method without S_{Loc_A} . An augmented navigation protocol was also implemented in the MHCP-MP framework.



Figure 4.11: A comparison of inter-vehicle distance in traffic network.

depicted flow-density diagram. One can find that the vehicle density in the unsupervised MVS system was unevenly distributed at various intersections. The flow decreases with the system running (e.g., intersection 1). Correspondingly, the output flow of the intersection network was showed at the bottom-left of Fig. 4.12. The approximate maximum output flow is 300Vehs/hr. After that, it dropped to a lower value. Nevertheless, the proposed method maintains a promising traffic flow-density performance within the same input flow rate. The S_{LocA} can regulate the traffic flow by considering the "road weights" linking to the road density. Therefore, a proportional increase in the flow-density dia-



Figure 4.12: Traffic flow-density diagrams for intersections and output flow in whole exits.



Figure 4.13: The traffic displacement diagram for the whole vehicles
gram has been observed during the whole simulation time. Additionally, the output flow with S_{LocA} remarkably increase to 700Vehs/hr at the end (see the bottom-right graph of Fig. 4.12). Similarly, the accessibility of the MVS system can be explained by Fig. 4.13. The displacement-time graphs were exhibited in up-left (unsupervised system) and up-right (supevised system) graph. A shock wave (congestion state) was induced between vehicles in the same lane when the ahead agents change their speeds. In contrast, the traffic congestion was alleviated in the supervised system which can adjust the traffic state properly. In addition, traffic accessibility can be improved by the proposed system with S_{LocA} . See bottom graph of Fig. 4.13, the color bar represents the displace for each vehicle. Roughly, vehicles in the supervised system shows a better accessibility of transportation.

Finally, the average flow of the compared MVS systems is recorded in Table 4.2:

	Average flow [Vehs/hr]			
Intersection	Without S _{Loc_A}	With S Loc _A		
	-	Self-Interested Protocol	MiMaFC-based Protocol	
1	99.9748	146.9410	148.9427	
2	103.4904	130.8983	134.6613	
3	101.9862	161.7955	160.3477	
4	97.2312	137.6753	138.6609	
5	87.7476	126.0182	126.2508	
6	96.4529	127.0486	127.8843	
7	106.0823	141.2156	142.5730	
8	91.8322	117.2555	118.9887	
9	89.1932	117.4012	118.0336	
flow out	50.8298	166.4516	169.0731	

Table 4.2: Average flow in local area

Noting that, the MVS system with basic protocol [4] was also included. Vehicle can participate in the optimization whenever it was labelled as a "conflict agent" in basic protocol. In general, the proposed approach with local supervisor S_{Loc_A} can guarantee relatively high traffic flow rates comparing to the total distributed MVS system without I2V technology in the different intersection. The vehicles with proposed augmented protocol even have better transportation performance comparing to vehicles implementing basic protocol. Seeing that, the average traffic flow rate (around 166Vehs/hr) out of the network with S_{Loc_A} is greater than the flow-out rate (51Vehs/hr) with non-supervised road network. While the augmented protocol can keep a higher output flow of 169Vehs/hr. It indicates that MVS system in the signal-free intersection with S_{Loc_A} and augmented protocol has the potential to improve traffic mobility. Briefly, the designed intelligent S_{Loc_A} can be beneficial as novel urban mobility management platforms to handle arterial traffic transportation.

4.6/ CONCLUSION

In this chapter, we explored the MHCP-MP architecture for MVS' CN in a road network based on the prior works in Chapter 3. First, an overview of the planned MHCP-MP traffic

134

scenario is given. We then built a macroscopic flow model to determine proposed MVS navigation references (speeds, passing rights) based on traffic flow fluctuation. Thus, the high-hierarchical MiMaFC-based policy can ensure that the reference behavior in micro motion control is applied robustly in response to traffic flow changes. Based on the developed MHCP-MP architecture, this study offered a CN protocol to efficiently decide each agent's motion planning approach. In a highly networked environment, the procedure was done by a local supervisor. Furthermore, the spatio-temporal velocity adaptation technique proposed in this chapter also improves the microscopic MVS optimization model for AIM. The local conflict resolution strategy can be easily calculated by adopting the velocity adaptation approach. The ε -PC algorithm's optimization target is redesigned to determine the recommended velocity strategy for crossing the intersection for micro motion planning. Simulation results indicate an overall improvement in traffic flow control. At an unsignalized intersection, the local supervisor S_{Loc_A} can guide the MVS' motion planning, so enhancing CN efficiency.

GENERAL CONCLUSION AND FUTURE WORK

GENERAL CONCLUSION

Several contributions dealing with multi-vehicle navigation in complicated environments/situations have been proposed throughout this PhD manuscript. The real objective of using the Collective Intelligence (CI) theory-based probability method is to increase the reliability and flexibility of the navigation system and, hence, on-road safety. As a risksensitive method, the suggested probability approach has been established in an attempt to avoid excessively conservative or too invasive procedures for cooperative intersection navigation. Under the risk preference hypothesis, the crossing ego-vehicles could make a quick decision while negotiating with other involved agents. Correspondingly, a logic risk indicator relationship is built between the measurement circumstances and the vehicle's current state. Furthermore, a well-suited trade-off between speed and risk from a global perspective is addressed in order to improve the vehicle navigation trajectory in a transit network. To deal with multi-vehicle navigation in a transportation network, in particular, the management-level agent that acts as a mediator between the global system and local cooperative vehicles are characterized. The Macroscopic Fundamental Diagram (MFD) allows us to monitor the present traffic status and assign policy in a distributed hierarchical control framework, providing instructions to regulate all vehicle trajectories. The intelligent intersection navigation systems are established with uncertain traffic flows under different directions, which combines the hierarchical traffic management architecture and the probability cooperative method for multi-vehicle systems. Finally, the developed vehicle cooperation system contributes to affecting the overall traffic state in a decentralized system while also enabling the mobility of the overall road network.

The first part of this thesis studied a variety of vehicle navigation strategies in typical multiple-vehicle driving environments. There has been a greater emphasis on intersection navigation systems, which comprise the management architecture and associated motion planning approaches. Moreover, we specifically evaluated the literature relating to transit network trajectory optimization.

Chapter 1 details real-world examples involving cooperative driving between multiple vehicles. The ultimate objective is to identify critical characteristics that contribute to the development of safe trajectory planning systems that account for inherent on-road uncertainty. More specifically, this chapter discussed the advantages and disadvantages of existing cooperative navigation techniques as well as current developments in autonomous intersection management systems. As a primary result of the state-of-the-art investigation, the usage of hierarchical design was deemed to be a very promising option for multiple vehicle cooperation. The centralized and distributed driving techniques, in particular,

may be combined to create a complementary set of capabilities.

Subsequently, Chapter 2 was devoted to digging more into the cooperative navigation system used in a transportation network. More precisely, decision-making for IVs is primarily focused on resolving local conflicts, and the impact of this micro-level control on overall traffic efficiency has not been adequately explored. The scope of the examined literature includes traffic control and cooperative operation of multiple vehicles. The research community's attempts to improve both of these targets were emphasized. It has been found that although traffic lights remain the most relevant approach, cooperative navigation technology should carefully analyze all conceivable decision-making strategies in urban traffic networks, ranging from passive feedback control to feedforward/response control. Additionally, the probabilistic approach and risk management are emphasized in order to execute a cooperative driving strategy that includes the development of short-term driving plans via bilateral or multilateral negotiations.

The successive chapters in the second part are devoted to explaining cooperative navigation methods for a group of vehicles, from a single intersection to a transit network. For a negotiated system, a probabilistic approach is taken while considering the research gaps described in the first part.

Chapter 3 firstly introduced a novel flexible and risk-sensitive cooperative navigation technique at a single intersection. The new approach enhances an existing Probability Collective (PC) algorithm based on a framework for multiple vehicle negation by explicitly accounting for the diverse risk preferences represented by ε -constraint values. ε -PC is a flexible negotiation scheme that allow vehicle to find a feasible crossing strategy between a group of vehicles in the proximity of an intersection. This chapter's main contribution is to include an on-road risk indicator as a major component of the proposed ε -PC's analysis scheme in order to assure safe autonomous navigation through a real-time communication mechanism. Simulation results prove that the decentralized negotiating system and probabilistic approach based on heuristics enable the cooperative navigation method to maintain a competitive computing time while processing uncertainties in an undetermined (but feasible) way. Secondly, the inherent issue of balancing risk and speed for IVs is examined from the viewpoint of a global supervisor in order to operate in a transit network. A heuristic local search algorithm LS-SPR is designed and appropriate learning principles is used for a speedy run in routing strategies based on the formulation of the short route under risk constraint (SPR) problem. Different instances are used to validate the simulations with bounded state and decision numbers. The findings demonstrated the efficacy of the suggested LS-SPR in obtaining a speedy solution in routing methods while taking risk and speed into account.

Finally, in chapter 4, a Micro-Macro Flow Control (MiMaFC) strategy is proposed, with a local supervisor acting as a mediator between a global traffic supervisor and local intelligent vehicles. Its application is suitable for multi-vehicle cooperative navigation in a network without signalized intersections. Further, MiMaFC policy is combined with a decentralized navigation framework called Multi-layer Hybrid Management Policy and Motion Planning (MHCP-MP) in order to provide proactive traffic flow control. This goal is what highlighted in the previous part of the state-of-the-art. Also, a macroscopic flow model was made using the elements associated with traffic fundamental diagrams. Correspondingly, a car-following model with an uncertain following strategy was used to simulate fluctuating traffic flow. Additionally, suitable speed-position-based decision-making procedures were presented to define the attributes of the autonomous navigation system in order to offer an adequate intersection navigation protocol for continuous vehicle flow. More precisely, the protocol consists of three steps: vehicle sorting, strategy selection, and conflict resolution in order to ensure the suggested MHCP-MP architecture. In particular, this chapter presented the spatio-temporal velocity adaptation method in order to improve the feasibility of cooperative solutions. Intelligent vehicles may use either the MiMaFC-based or self-interested speed strategy, depending on their location and speed. Thus, the heuristic searching procedure will not fail in a planning/re-planning process. The preceding chapter's suggested PC algorithm is implemented into the decision-making architecture. Finally, simulation results demonstrated that models developed using MiMaFC inside the MHCP-MP framework are capable of dealing with fluctuating vehicle flow in a traffic network and improving mobility compared to an unsupervised system.

PERSPECTIVES AND FUTURE WORK

The various contributions made in this thesis may result in an introduction of a heterogeneous traffic flow navigation framework. The flexible and safe cooperative navigation approach enables the management of unsignalized intersections. Indeed, this will allow us the exploration and extension of various research topics in the field of MVS cooperation, microscopic control model, and traffic flow management. The following summarizes the most prominent works that we plan to perform in near future.

Extend the approach of Cooperative Navigation (CN) to meet diverse objectives: our work is primarily concerned with the issue of intersection cooperation in motion planning decision-making. However, several optimization objectives such as constrained time to destination for MVS, fuel economy for cooperative eco-driving, cooperative maneuvering for comfort driving, and collision avoidance/mitigation remain unresolved (see Figure 4.14). Because the majority of the methods examined for MVS dynamic driving tasks have been developed under strict conditions. The suggested targets will add complexity to the CN strategy design process. How to use the bottom-up method to produce a fully available modular system remains an unanswered question. It could be interesting to use our decentralized collaboration framework to accomplish the suggested tasks.

Improve the MVS microscopic control model for mixed traffic flow: a heterogeneous MVS navigation framework can integrate various traffic participants, such as humandriven cars, emerging vehicles, and/or commercial/public vehicles, among others. The addressed risk assessment/management and cooperative navigation protocol must be reviewed in various layers of the proposed hierarchical control architecture. Nevertheless, since the suggested ε -PC technique is probabilistic in nature, it may simply be used to put the traffic participants' movements into probability space. Thus, Intelligent Vehicles (IVs) might still execute the suggested intersection navigation approach without distinguishing between an automated or human-driven vehicle. Additionally, we may use Reinforcement Learning (RL) technologies to accelerate the formulation of the navigation strategy space for conventional vehicles. Similarly, the formalization of the 2D TTC should be expanded to include more complicated situations involving uncertainty, such as overtaking maneuvers by non cooperative traffic players.

Contribute to the enhancement of macroscopic modeling: due to the several layers of assured performance, this hierarchical navigation system with partially centralized decision-making would possess significant capabilities for avoiding traffic conges-



Figure 4.14: Past and potential evolution towards Cooperative Navigation (CN) [49].

tion. However, the cooperative architectures proposed in this thesis may be upgraded to a completely transportation navigation framework within which the global supervisor may operate more effectively. One may observe that the proposed MiMaFC strategy does not contain a global route strategy that incorporates the ideal trade-off between risk and speed (i.e., our proposed LS-SPR algorithm). This is because the proposed SPR issue is discrete decision-making model in nature, imposing constraints on the (piecewise) speed function (cf. Appendix A). Additionally, a cooperative navigation technique for multiple vehicles is presented in this PhD thesis based on an optimum control issue with a regular continuous speed function. Thus, integrating these two layers (macro and micro) in order to ensure overall performance remains a difficulty. Because the global supervisor represents the system's current condition and short-term development, navigation risk may be significantly reduced when following a pre-scheduled low-risk path. Thus, the local supervisor and/or on-board embedded system can execute their task more simply.

Simulation with real-world data and estimation techniques: the majority of the proposed methodologies have been proven only through extensive simulation work. As a result, the proposed PC method and MiMaFC strategy should be implemented on multiple real-world vehicles or a large-scale simulation based on real-world traffic data in the near future. This will create a number of technical challenges, including as the reliable implementation of software components and the functional safety of automobile operating systems. Furthermore, it is uncertain how to instance these hierarchical layers, which include various components that allow a collaborative function. Indeed, the test ground's multiple coordination of various signal-free intersections has been just getting started. Additional efforts are required to implement the navigation protocols while dealing with the many types of interference in vehicle communication. A more practical research may be done at the Institut Pascal laboratory, which has various Véhicule Individuel Public et Autonome (VIPALAB) and Plate-forme d'Auvergne pour Véhicules INtelligents (PAVIN) test ground. Additionally, reliable traffic status estimation remains an unresolved issue. As noted in state of the art (section 2.1.2), traffic assessments based simply on data collection would be challenging in the absence of actual measurement tools. Combining data collecting and estimation techniques for our traffic state estimation may be a potential method in a future research.

ANNEXES

Α

STRUCTURE RESULTS FOR SHORTEST PATH UNDER RISK CONSTRAINT (SPR) PROBLEM

As it is said in section 3.3, SPR problem appears to be more of an optimal control problem than a combinatorial one, because the speed function $t \rightarrow u(t)$ can be thought of as a regular continuous function. However, as we will see shortly, we can apply constraints on this speed function, bringing the SPR model closer to a discrete choice model. These constraints will greatly simplify the formulation of SPR. The first is about the shape of the speed function u(t) which can be chosen as piecewise linear with severe breakpoint constraints:

Proposition 1: Optimal solution (Γ , u) of SPR may be chosen in such a way that u is piecewise constant, with breakpoints related to the times t_i when vehicle *VEH* arrives at the end-nodes of arcs t_i , i = 1, ..., n, and to the breakpoints of function Π^{ei} , i = 1, ..., n.

Proof. Let us suppose that *VEC* is moving along some arc $e = e_i$, and that δ_1, δ_2 are 2 consecutive breakpoints in above sense of proposition 1. If function $t \to u(t)$ is not constant between δ_1 and δ_2 then we may replace u(t) by the mean value u^* of function $t \to u(t)$ between δ_1 and δ_2 . Time value $Time(\Gamma, u)$ remains unchanged, while risk value $Risk(\Gamma, u)$ decreases because of the convexity of function Φ . So we conclude.

It comes that we may impose function u(t) to be piecewise constant in section 3.3, with breakpoints which corresponds to the times when vehicle *VEC* shifts from an arc *e* to its successor in Γ and to the breakpoints of functions $t \to \Pi^e(t)$.

B

THE STOCHASTIC SPACE POLICY IN MACROSCOPIC FLOW MODEL

It assumes that MVS can be predefined with preferred car following strategy (e.g., invasive or conservative behaviors). Therefore, the desired distance d_{ref} can be defined by stochastic time headway (d_{safe} standstill safe distance, cf. section 4.3.1)

$$d_{ref} = d_{safe} + th_i \times v_i(t) \tag{B.1}$$

We further assume that th_i sampled based on a shifted log-normal distribution. Vehicle is supposed to generate an independent and identically distributed (i.i.d.) $t\hat{h}_i$ as selfpreferred time gap (i.e., $th_i = t\hat{h}_i$)

$$th_i \sim Log - N(\mu_v, \sigma_v)$$
 (B.2)

Where μ_{ν} and σ_{ν} are the predefined velocity dependent parameters in log-normal distribution as illustrated in Table B.1. Notably, we use the NGSIM [461] data set to match the log-normal distribution over four distinct speed ranges as seen in Figure B.1. Generally, as the speed of the vehicle decreases, the vehicle's time headway increases. This implies that while a vehicle's speed is low, it is more sensitive to the safe gap between it and the vehicle ahead of it. Thus, in section 4.5, we select the time headway based on the speed shown in Table B.1. A more detailed $t\hat{h}_i$ distribution based on the NGSIM data set can be found in Figures B.2.

Table B.1: A test of the log-normal distribution's parameters

	$t\hat{h}_i$ based on a log-normal distribution			
NO.	Speed range	Parameters in a log-normal distribution		
	-	μ_v	σ_v	
1	[0m/s, 5m/s]	1.3115	0.520339	
2	[5m/s, 10m/s]	0.8800	0.419708	
3	[10m/s, 15m/s]	0.7765	0.435128	
4	$[15m/s, +\infty m/s]$	0.7337	0.521951	
Confidence interval	-	[0.7266, 1.3169]	-	



Figure B.1: A comparison of different time headway $t\hat{h}_i$ distribution.



Figure B.2: The time headway distribution $t\hat{h}_i$ at speeds of [0m/s, 5m/s], [5m/s, 10m/s], [10m/s, 15m/s] and $[15m/s, +\infty m/s]$.

LIST OF MY PUBLICATIONS

- [1] <u>ZHU, Z.</u>, ADOUANE, L., AND QUILLIOT, A. A decentralized multi-criteria optimization algorithm for multi-unmanned ground vehicles (mugvs) navigation at signal-free intersection. *IFAC Symposium on Control in Transportation Systems* (CTS) 54, 2 (Lile, France, June 8-10 2021), 327–334.
- [2] <u>ZHU, Z.</u>, ADOUANE, L., AND QUILLIOT, A. Flexible multi-unmanned ground vehicles (mugvs) in intersection coordination based on ε-constraint probability collectives algorithm. International Journal of Intelligent Robotics and Applications 5, 2 (2021), 156–175.
- [3] <u>ZHU, Z.</u>, ADOUANE, L., AND QUILLIOT, A. Hierarchical control for trajectory-based intelligent navigation in urban adjacent intersections. 2021 IEEE International Intelligent Transportation Systems Conference (ITSC) (Indianapolis, USA, September 19-22 2021), 948–954.
- [4] <u>ZHU, Z.</u>, ADOUANE, L., AND QUILLIOT, A. Intelligent traffic based on hybrid control policy of connected autonomous vehicles in multiple unsignalized intersections. In 2021 IEEE SmartWorld, Ubiquitous Intelligence & Computing, Advanced & Trusted Computing, Scalable Computing & Communications, Internet of People and Smart City Innovation (SmartWorld/SCALCOM/UIC/ATC/IOP/SCI) (Atlanta, USA, October 18-21 2021), pp. 416–424.

BIBLIOGRAPHY

- [5] Abboud, K., Omar, H. A., et Zhuang, W. (2016). Interworking of dsrc and cellular network technologies for v2x communications: A survey. *IEEE transactions on* vehicular technology, 65(12):9457–9470.
- [6] Aboudolas, K., et Geroliminis, N. (2013). Perimeter and boundary flow control in multi-reservoir heterogeneous networks. *Transportation Research Part B: Method*ological, 55:265–281.
- [7] Abubakr, M., Abucharr, O., Al-Stouhi, S., Buchard, M., Das, S., Ibrahim, U., Kailas, A., Konet, H., Kumar, V., Meier, J., et others (2016). Eco-approach and eco-departure planning study. Technical Report, Crash Avoidance Metrics Partnership.
- [8] Adouane, L. (2016). Autonomous vehicle navigation: from behavioral to hybrid multi-controller architectures. Taylor & Francis CRC Press, ISBN: 9781498715584, 228 pages.
- [9] Adouane, L. (2017a). Reactive versus cognitive vehicle navigation based on optimal local and global pelc. *Robotics and Autonomous Systems*, 88:51–70.
- [10] Adouane, L. (2017b). Toward fully autonomous vehicle navigation: From behavioral to hybrid multi-controller architectures. In 2017 11th International Workshop on Robot Motion and Control (RoMoCo), pages 85–98. IEEE.
- [11] Ahmed, M. S., Hoque, M. A., Rios-Torres, J., et Khattak, A. (2017). Freeway merge assistance system using dsrc. In *Proceedings of the 2nd ACM International Work*shop on Smart, Autonomous, and Connected Vehicular Systems and Services, pages 83–84.
- [12] Ahmed, M. S., Hoque, M. A., Rios-Torres, J., et Khattak, A. (2018). A cooperative freeway merge assistance system using connected vehicles. *arXiv preprint arXiv:1805.00508*.
- [13] Al-Dabbagh, M. S. M., Al-Sherbaz, A., et Turner, S. (2018). The impact of road intersection topology on traffic congestion in urban cities. In Proceedings of SAI Intelligent Systems Conference, pages 1196–1207. Springer.
- [14] Al Islam, S. B., et Hajbabaie, A. (2017). Distributed coordinated signal timing optimization in connected transportation networks. *Transportation Research Part C: Emerging Technologies*, 80:272–285.
- [15] Alexandru, C. (2019). Dynamic simulation of the adaptive sun tracking system used for an electric unmanned ground vehicle. In *IOP Conference Series: Materials Science and Engineering*, volume 568, page 012019. IOP Publishing.
- [16] Alonso, J., Milanés, V., Pérez, J., Onieva, E., González, C., et De Pedro, T. (2011). Autonomous vehicle control systems for safe crossroads. *Transportation research part C: emerging technologies*, 19(6):1095–1110.

- [17] Alonso-Mora, J., Siegwart, R., et Beardsley, P. (2014). Human-robot swarm interaction for entertainment: From animation display to gesture based control. In Proceedings of the 2014 ACM/IEEE international conference on Human-robot interaction, pages 98–98.
- [18] Altan, O. D., Wu, G., Barth, M. J., Boriboonsomsin, K., et Stark, J. A. (2017). Glidepath: Eco-friendly automated approach and departure at signalized intersections. *IEEE Transactions on Intelligent Vehicles*, 2(4):266–277.
- [19] Altché, F., Qian, X., et de La Fortelle, A. (2016). Least restrictive and minimally deviating supervisor for safe semi-autonomous driving at an intersection: An miqp approach. In 2016 IEEE 19th International Conference on Intelligent Transportation Systems (ITSC), pages 2520–2526. IEEE.
- [20] Althoff, M., Koschi, M., et Manzinger, S. (2017). Commonroad: Composable benchmarks for motion planning on roads. In 2017 IEEE Intelligent Vehicles Symposium (IV), pages 719–726. IEEE.
- [21] Amazon.com, I. (2021). Amazon.com: Prime air. available at https://www.amazon. com/primeair.
- [22] Ambrose, R. O., Aldridge, H., Askew, R. S., Burridge, R. R., Bluethmann, W., Diftler, M., Lovchik, C., Magruder, D., et Rehnmark, F. (2000). Robonaut: Nasa's space humanoid. *IEEE Intelligent Systems and Their Applications*, 15(4):57–63.
- [23] Amoozadeh, M., Deng, H., Chuah, C.-N., Zhang, H. M., et Ghosal, D. (2015). Platoon management with cooperative adaptive cruise control enabled by vanet. *Vehicular communications*, 2(2):110–123.
- [24] Anderson, J. M., Nidhi, K., Stanley, K. D., Sorensen, P., Samaras, C., et Oluwatola, O. A. (2014). Autonomous vehicle technology: A guide for policymakers. Rand Corporation.
- [25] Aoki, R. (2017). Report on autonomous mode disengagements for waymo selfdriving vehicles in california. Technical Report, Technical report, WAYMO.
- [26] Arai, T., Pagello, E., Parker, L. E., et others (2002). Advances in multi-robot systems. *IEEE Transactions on robotics and automation*, 18(5):655–661.
- [27] Archer, J. (2005). Indicators for traffic safety assessment and prediction and their application in micro-simulation modelling: A study of urban and suburban intersections. PhD thesis, Kungliga Tekniska högskolan.
- [28] Arkin, R. C., Arkin, R. C., et others (1998). Behavior-based robotics. MIT press.
- [29] Åström, K. J., et Kumar, P. R. (2014). Control: A perspective. Automatica, 50(1):3–43.
- [30] Atallah, R. F., Khabbaz, M. J., et Assi, C. M. (2015). Vehicular networking: A survey on spectrum access technologies and persisting challenges. Vehicular Communications, 2(3):125–149.
- [31] Au, T.-C., et Stone, P. (2010). Motion planning algorithms for autonomous intersection management. In Workshops at the Twenty-Fourth AAAI Conference on Artificial Intelligence.

- [32] Azimi, R., Bhatia, G., Rajkumar, R., et Mudalige, P. (2013). V2v-intersection management at roundabouts. SAE Int. J. Passeng. Cars-Mech. Syst, 6(2):681–690.
- [33] Azimi, S. R., Bhatia, G., Rajkumar, R., et Mudalige, P. (2011). Vehicular networks for collision avoidance at intersections. SAE International Journal of Passenger Cars-Mechanical Systems, 4(1):406–416.
- [34] Aziz, H. A., et Ukkusuri, S. V. (2012). Unified framework for dynamic traffic assignment and signal control with cell transmission model. *Transportation Research Record*, 2311(1):73–84.
- [35] Baber, J., Kolodko, J., Noel, T., Parent, M., et Vlacic, L. (2005). Cooperative autonomous driving: intelligent vehicles sharing city roads. IEEE Robotics & Automation Magazine, 12(1):44–49.
- [36] BačÍK, J., DUrovský, F., Biroš, M., Kyslan, K., Perdukova, D., et Padmanaban, S. (2017). Pathfinder-development of automated guided vehicle for hospital logistics. *leee Access*, 5:26892–26900.
- [37] Badue, C., Guidolini, R., Carneiro, R. V., Azevedo, P., Cardoso, V. B., Forechi, A., Jesus, L., Berriel, R., Paixao, T. M., Mutz, F., et others (2021). Self-driving cars: A survey. *Expert Systems with Applications*, 165:113816.
- [38] Baïou, M., Mombelli, A., Quilliot, A., Adouane, L., et <u>Zhu, Z.</u> (2021). Algorithms for the safe management of autonomous vehicles. In 2021 16th Conference on Computer Science and Intelligence Systems (FedCSIS), pages 153–162. IEEE.
- [39] Ban, X. J., Hao, P., et Sun, Z. (2011). Real time queue length estimation for signalized intersections using travel times from mobile sensors. *Transportation Research Part C: Emerging Technologies*, 19(6):1133–1156.
- [40] Barnes, P. J., et Swan, B. (2019). High definition mapping and data needs for connected and automated vehicles. Technical Report, Institute for Public Administration.
- [41] Baskar, L. D., De Schutter, B., Hellendoorn, J., et Papp, Z. (2011). Traffic control and intelligent vehicle highway systems: a survey. IET Intelligent Transport Systems, 5(1):38–52.
- [42] Bazzan, A. L., De Oliveira, D., et da Silva, B. C. (2010). Learning in groups of traffic signals. Engineering Applications of Artificial Intelligence, 23(4):560–568.
- [43] Beak, B., Head, K. L., et Feng, Y. (2017). Adaptive coordination based on connected vehicle technology. *Transportation Research Record*, 2619(1):1–12.
- [44] Beasley, J. E. (1983). Route first—cluster second methods for vehicle routing. Omega, 11(4):403–408.
- [45] Ben Lakhal, N. M., Adouane, L., Nasri, O., et Hadj Slama, J. B. (2019). Risk management for intelligent vehicles based on interval analysis of ttc. *IFAC-PapersOnLine*, 52(8):338–343. 10th IFAC Symposium on Intelligent Autonomous Vehicles IAV 2019: Gdansk, Poland, 3–5 July 2019.

- [46] Ben Lakhal., N. M., Nasri., O., Adouane., L., et Slama., J. (2020). Reliable modeling for safe navigation of intelligent vehicles: Analysis of first and second order setmembership ttc. In Proceedings of the 17th International Conference on Informatics in Control, Automation and Robotics - Volume 1: ICINCO,, pages 545–552, Paris-France. INSTICC, SciTePress.
- [47] Ben Ticha, H., Absi, N., Feillet, D., et Quilliot, A. (2018). Vehicle routing problems with road-network information: State of the art. *Networks*, 72(3):393–406.
- [48] Bendali, F., Kamga, E. M., Mailfert, J., Quilliot, A., et Toussaint, H. (2021). Synchronizing energy production and vehicle routing. *RAIRO-Operations Research*, 55(4):2141–2163.
- [49] Bengler, K., Dietmayer, K., Farber, B., Maurer, M., Stiller, C., et Winner, H. (2014). Three decades of driver assistance systems: Review and future perspectives. *IEEE Intelligent transportation systems magazine*, 6(4):6–22.
- [50] Benmimoun, M., Pütz, A., Zlocki, A., et Eckstein, L. (2013). eurofot: Field operational test and impact assessment of advanced driver assistance systems: Final results. In *Proceedings of the FISITA 2012 World Automotive Congress*, pages 537–547. Springer.
- [51] Benzerrouk, A., Adouane, L., et Martinet, P. (2013). Obstacle avoidance controller generating attainable set-points for the navigation of multi-robot system. In 2013 IEEE Intelligent Vehicles Symposium (IV), pages 487–492.
- [52] Benzerrouk, A., Adouane, L., et Martinet, P. (2014). Stable navigation in formation for a multi-robot system based on a constrained virtual structure. *Robotics and Autonomous Systems*, 62(12):1806–1815.
- [53] Berbeglia, G., Cordeau, J.-F., Gribkovskaia, I., et Laporte, G. (2007). Static pickup and delivery problems: a classification scheme and survey. *Top*, 15(1):1–31.
- [54] Berbeglia, G., Cordeau, J.-F., et Laporte, G. (2010). Dynamic pickup and delivery problems. *European journal of operational research*, 202(1):8–15.
- [55] Bergenhem, C., Shladover, S., Coelingh, E., Englund, C., et Tsugawa, S. (2012). Overview of platooning systems. In *Proceedings of the 19th ITS World Congress, Oct 22-26, Vienna, Austria (2012).*
- [56] Bernardin Jr, V. L., Mansfield, T., Swanson, B., Sadrsadat, H., et Bindra, S. (2019). Scenario modeling of autonomous vehicles with trip-based models. *Transportation Research Record*, 2673(10):261–270.
- [57] Bertolazzi, E., Biral, F., et Da Lio, M. (2007). Real-time motion planning for multibody systems. *Multibody System Dynamics*, 17(2-3):119–139.
- [58] Bieniawski, S. R. (2005). Distributed optimization and flight control using collectives. PhD thesis, Stanford University.
- [59] Binas, J., Neil, D., Liu, S.-C., et Delbruck, T. (2017). Ddd17: End-to-end davis driving dataset. arXiv preprint arXiv:1711.01458.

- [60] Blatt, A., Pierowicz, J., Flanigan, M., Lin, P.-S., Kourtellis, A., Lee, C., Jovanis, P., Jenness, J., Wilaby, M., Campbell, J., et others (2015). Naturalistic driving study: Field data collection. Report S2-S07-RW-1, SHRP 2 Rep.
- [61] Blum, J., et Eskandarian, A. (2004). The threat of intelligent collisions. IT professional, 6(1):24–29.
- [62] Blum, J. J., Tararakin, A., et Eskandarian, A. (2008). Efficient certificate distribution for vehicle heartbeat messages. In 2008 IEEE 68th Vehicular Technology Conference, pages 1–5. Calgary, Alberta, September 21–24.
- [63] Borja, R., De La Pinta, J. R., Álvarez, A., et Maestre, J. M. (2013). Integration of service robots in the smart home by means of upnp: A surveillance robot case study. *Robotics and Autonomous Systems*, 61(2):153–160.
- [64] Bouraoui, L., Petti, S., Laouiti, A., Fraichard, T., et Parent, M. (2006). Cybercar cooperation for safe intersections. In 2006 IEEE Intelligent Transportation Systems Conference, pages 456–461. IEEE.
- [65] Brooks, R. (1986). A robust layered control system for a mobile robot. *IEEE journal on robotics and automation*, 2(1):14–23.
- [66] Bruni, L., Colombo, A., et Del Vecchio, D. (2013). Robust multi-agent collision avoidance through scheduling. In 52nd IEEE Conference on Decision and Control, pages 3944–3950. IEEE.
- [67] Bsaybes, S., Quilliot, A., et Wagler, A. K. (2018). Fleet management for autonomous vehicles using multicommodity coupled flows in time-expanded networks. In 17th International Symposium on Experimental Algorithms (SEA 2018). Schloss Dagstuhl-Leibniz-Zentrum fuer Informatik.
- [68] Buehler, M., lagnemma, K., et Singh, S. (2007). The 2005 DARPA grand challenge: the great robot race, volume 36. Springer.
- [69] Buisson, C., et Ladier, C. (2009). Exploring the impact of homogeneity of traffic measurements on the existence of macroscopic fundamental diagrams. *Trans*portation Research Record, 2124(1):127–136.
- [70] Burger, C., Orzechowski, P. F., Taş, Ö. Ş., et Stiller, C. (2017). Rating cooperative driving: A scheme for behavior assessment. In 2017 IEEE 20th International Conference on Intelligent Transportation Systems (ITSC), pages 1–6.
- [71] Burns, L. D. (2013). A vision of our transport future. Nature, 497(7448):181–182.
- [72] Cadena, C., Carlone, L., Carrillo, H., Latif, Y., Scaramuzza, D., Neira, J., Reid, I., et Leonard, J. J. (2016). Past, present, and future of simultaneous localization and mapping: Toward the robust-perception age. *IEEE Transactions on robotics*, 32(6):1309–1332.
- [73] Caesar, H., Bankiti, V., Lang, A. H., Vora, S., Liong, V. E., Xu, Q., Krishnan, A., Pan, Y., Baldan, G., et Beijbom, O. (2020). nuscenes: A multimodal dataset for autonomous driving. In *Proceedings of the IEEE/CVF conference on computer vision* and pattern recognition, pages 11621–11631.

- [74] Calabrese, F., Colonna, M., Lovisolo, P., Parata, D., et Ratti, C. (2010). Real-time urban monitoring using cell phones: A case study in rome. *IEEE transactions on intelligent transportation systems*, 12(1):141–151.
- [75] Cao, Y. U., Kahng, A. B., et Fukunaga, A. S. (1997). Cooperative mobile robotics: Antecedents and directions. In *Robot colonies*, pages 7–27. Springer.
- [76] Carl, A. (1962). Petri. kommunikation mit automaten. PhD, University of Bonn, West Germany.
- [77] Ceylan, H., et Bell, M. G. (2004). Traffic signal timing optimisation based on genetic algorithm approach, including drivers' routing. *Transportation Research Part B: Methodological*, 38(4):329–342.
- [78] Chai, L., Cai, B., ShangGuan, W., et Wang, J. (2017). Connected and autonomous vehicles coordinating method at intersection utilizing preassigned slots. In 2017 IEEE 20th International Conference on Intelligent Transportation Systems (ITSC), pages 1–6.
- [79] Chan, E. (2016). Sartre automated platooning vehicles. *Towards innovative freight and logistics*, 2:137–150.
- [80] Chan, E., Gilhead, P., Jelinek, P., Krejci, P., et Robinson, T. (2012). Cooperative control of sartre automated platoon vehicles. In 19th ITS World CongressERTICO-ITS EuropeEuropean CommissionITS AmericaITS Asia-Pacific.
- [81] Charalampidis, A. C., et Gillet, D. (2014). Speed profile optimization for vehicles crossing an intersection under a safety constraint. In 2014 european control conference (ecc), pages 2894–2901. leee.
- [82] Chen, C., Petty, K., Skabardonis, A., Varaiya, P., et Jia, Z. (2001). Freeway performance measurement system: mining loop detector data. *Transportation Research Record*, 1748(1):96–102.
- [83] Chen, C., Xiang, H., Qiu, T., Wang, C., Zhou, Y., et Chang, V. (2018a). A rear-end collision prediction scheme based on deep learning in the internet of vehicles. *Journal of Parallel and Distributed Computing*, 117:192–204.
- [84] Chen, D., Ahn, S., Chitturi, M., et Noyce, D. A. (2017). Towards vehicle automation: Roadway capacity formulation for traffic mixed with regular and automated vehicles. *Transportation research part B: methodological*, 100:196–221.
- [85] Chen, L., et Englund, C. (2015). Cooperative intersection management: A survey. IEEE Transactions on Intelligent Transportation Systems, 17(2):570–586.
- [86] Chen, N., Wang, M., Alkim, T., et van Arem, B. (2018b). A robust longitudinal control strategy of platoons under model uncertainties and time delays. *Journal of Advanced Transportation*, 2018.
- [87] Chen, Q. A., Yin, Y., Feng, Y., Mao, Z. M., et Liu, H. X. (2018c). Exposing congestion attack on emerging connected vehicle based traffic signal control. In *NDSS*.

- [88] Chen, X., Xu, B., Qin, X., Bian, Y., Hu, M., et Sun, N. (2020). Non-signalized intersection network management with connected and automated vehicles. *IEEE Access*, 8:122065–122077.
- [89] Chen, Y., Wang, J., Li, J., Lu, C., Luo, Z., Xue, H., et Wang, C. (2018d). Lidar-video driving dataset: Learning driving policies effectively. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, pages 5870–5878.
- [90] Chen, Y., Zha, J., et Wang, J. (2019). An autonomous t-intersection driving strategy considering oncoming vehicles based on connected vehicle technology. IEEE/ASME Transactions on Mechatronics, 24(6):2779–2790.
- [91] Cheng, Y., Qin, X., Jin, J., et Ran, B. (2012). An exploratory shockwave approach to estimating queue length using probe trajectories. *Journal of intelligent transportation systems*, 16(1):12–23.
- [92] Chin, H.-C., et Quek, S.-T. (1997). Measurement of traffic conflicts. Safety Science, 26(3):169–185.
- [93] Chou, F.-C., Shladover, S. E., et Bansal, G. (2016). Coordinated merge control based on v2v communication. In 2016 IEEE Vehicular Networking Conference (VNC), pages 1–8. IEEE.
- [94] Chu, T., Wang, J., Codecà, L., et Li, Z. (2019). Multi-agent deep reinforcement learning for large-scale traffic signal control. IEEE Transactions on Intelligent Transportation Systems, 21(3):1086–1095.
- [95] Chung, Y., Choi, Y.-H., et Yoon, B.-J. (2018). Safe operation guidelines for electronic toll collection systems: a case study in korea. International Journal of Civil Engineering, 16(3):281–288.
- [96] Coffey, S., et Park, S. (2020). Part-time shoulder use operational impact on the safety performance of interstate 476. *Traffic injury prevention*, 21(7):470–475.
- [97] Comma.ai (2021). openpilot. available at https://github.com/commaai/openpilot.
- [98] Committee, S. O.-R. A. V. S., et others (2021). Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicles. SAE Standard J3016202104.
- [99] Cordeau, J.-F., et Laporte, G. (2007). The dial-a-ride problem: models and algorithms. Annals of operations research, 153(1):29–46.
- [100] Cui, S., Seibold, B., Stern, R., et Work, D. B. (2017). Stabilizing traffic flow via a single autonomous vehicle: Possibilities and limitations. In 2017 IEEE Intelligent Vehicles Symposium (IV), pages 1336–1341. IEEE.
- [101] Daganzo, C. F., Gayah, V. V., et Gonzales, E. J. (2011). Macroscopic relations of urban traffic variables: Bifurcations, multivaluedness and instability. *Transportation Research Part B: Methodological*, 45(1):278–288.
- [102] Daganzo, C. F., Gayah, V. V., et Gonzales, E. J. (2012). The potential of parsimonious models for understanding large scale transportation systems and answering big picture questions. EURO Journal on Transportation and Logistics, 1(1-2):47–65.

- [103] Dai, P., Liu, K., Zhuge, Q., Sha, E. H.-M., Lee, V. C. S., et Son, S. H. (2016). Quality-of-experience-oriented autonomous intersection control in vehicular networks. *IEEE Transactions on Intelligent Transportation Systems*, 17(7):1956–1967.
- [104] Darmoul, S., Elkosantini, S., Louati, A., et Said, L. B. (2017). Multi-agent immune networks to control interrupted flow at signalized intersections. *Transportation Research Part C: Emerging Technologies*, 82:290–313.
- [105] Dávila, A., et Nombela, M. (2011). Sartre-safe road trains for the environment reducing fuel consumption through lower aerodynamic drag coefficient. Technical Report, SAE Technical Paper.
- [106] De Campos, G. R., Falcone, P., et Sjöberg, J. (2013). Autonomous cooperative driving: a velocity-based negotiation approach for intersection crossing. In 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), pages 1456–1461.
- [107] De La Fortelle, A. (2010). Analysis of reservation algorithms for cooperative planning at intersections. In 13th International IEEE conference on intelligent transportation systems, pages 445–449.
- [108] Decker, M., Fischer, M., et Ott, I. (2017). Service robotics and human labor: A first technology assessment of substitution and cooperation. *Robotics and Autonomous Systems*, 87:348–354.
- [109] Deleplanque, S., Quilliot, A., et Bernay, B. (2014). Branch and price for a reliability oriented darp model. In 2014 International Conference on Control, Decision and Information Technologies (CoDIT), pages 081–086. IEEE.
- [110] Deng, J., Dong, W., Socher, R., Li, L.-J., Li, K., et Fei-Fei, L. (2009). Imagenet: A large-scale hierarchical image database. In 2009 IEEE conference on computer vision and pattern recognition, pages 248–255. leee.
- [111] Dias, C., Iryo-Asano, M., Abdullah, M., Oguchi, T., et Alhajyaseen, W. (2020). Modeling trajectories and trajectory variation of turning vehicles at signalized intersections. *IEEE Access*, 8:109821–109834.
- [112] Dinopoulou, V., Diakaki, C., et Papageorgiou, M. (2005). Application and evaluation of the signal traffic control strategy tuc in chania. *Journal of Intelligent Transportation Systems*, 9(3):133–143.
- [113] Dixit, S., Fallah, S., Montanaro, U., Dianati, M., Stevens, A., Mccullough, F., et Mouzakitis, A. (2018). Trajectory planning and tracking for autonomous overtaking: State-of-the-art and future prospects. *Annual Reviews in Control*, 45:76–86.
- [114] Dong, C. J., Shao, C. F., Zhuge, C.-X., et Meng, M. (2012). Spatial and temporal characteristics for congested traffic on urban expressway. *Journal of Beijing Polytechnic University*, 38(8):128–132.
- [115] Dosovitskiy, A., Ros, G., Codevilla, F., Lopez, A., et Koltun, V. (2017). Carla: An open urban driving simulator. In *Conference on robot learning*, pages 1–16. PMLR.

- [116] Dresner, K., et Stone, P. (2004). Multiagent traffic management: A reservationbased intersection control mechanism. In Autonomous Agents and Multiagent Systems, International Joint Conference on, volume 3, pages 530–537. IEEE Computer Society.
- [117] Dresner, K., et Stone, P. (2008). A multiagent approach to autonomous intersection management. *Journal of artificial intelligence research*, 31:591–656.
- [118] Du, Z., HomChaudhuri, B., et Pisu, P. (2018). Hierarchical distributed coordination strategy of connected and automated vehicles at multiple intersections. *Journal of Intelligent Transportation Systems*, 22(2):144–158.
- [119] Duhamel, C., Lacomme, P., Quilliot, A., et Toussaint, H. (2011). A multi-start evolutionary local search for the two-dimensional loading capacitated vehicle routing problem. *Computers & Operations Research*, 38(3):617–640.
- [120] Electronics, S. (2021). Sparkfun autonomous vehicle competition. available at https://avc.sparkfun.com/.
- [121] Elliott, D., Keen, W., et Miao, L. (2019). Recent advances in connected and automated vehicles. Journal of Traffic and Transportation Engineering (English Edition), 6(2):109–131.
- [122] (ERDF), T. E. R. D. F. (2022). Intelligent transportation for dynamic environment. available at https://keep.eu/projects/7040/Intelligent-Transportation-f-EN/.
- [123] Eskandarian, A. (2012). Handbook of intelligent vehicles, volume 2. Springer.
- [124] Eskandarian, A., Wu, C., et Sun, C. (2019). Research advances and challenges of autonomous and connected ground vehicles. *IEEE Transactions on Intelligent Transportation Systems*.
- [125] Evad37 (2022). Main roads western australia's traffic operations centre in northbridge, western australia. available at https://commons.wikimedia.org/wiki/ Main_Page.
- [126] Evestedt, N., Ward, E., Folkesson, J., et Axehill, D. (2016). Interaction aware trajectory planning for merge scenarios in congested traffic situations. In 2016 IEEE 19th International Conference on Intelligent Transportation Systems (ITSC), pages 465–472.
- [127] Fagnant, D. J., et Kockelman, K. (2015). Preparing a nation for autonomous vehicles: opportunities, barriers and policy recommendations. *Transportation Research Part A: Policy and Practice*, 77:167–181.
- [128] Fan, H., <u>Zhu, F.</u>, Liu, C., Zhang, L., Zhuang, L., Li, D., <u>Zhu, W.</u>, Hu, J., Li, H., et Kong, Q. (2018). Baidu apollo em motion planner. arXiv preprint arXiv:1807.08048.
- [129] Fang, F. C., et Elefteriadou, L. (2006). Development of an optimization methodology for adaptive traffic signal control at diamond interchanges. *Journal of Transportation Engineering*, 132(8):629–637.
- [130] Feddema, J. T., Lewis, C., et Schoenwald, D. A. (2002). Decentralized control of cooperative robotic vehicles: theory and application. *IEEE Transactions on robotics and automation*, 18(5):852–864.

- [131] FedEx (2019). Delivering the future: Fedex unveils autonomous delivery robot. available at https://newsroom.fedex.com/newsroom/thefuturefedex/.
- [132] Feng, Y., Head, K. L., Khoshmagham, S., et Zamanipour, M. (2015). A real-time adaptive signal control in a connected vehicle environment. *Transportation Re*search Part C: Emerging Technologies, 55:460–473.
- [133] Feng, Y., Huang, S., Chen, Q. A., Liu, H. X., et Mao, Z. M. (2018). Vulnerability of traffic control system under cyberattacks with falsified data. *Transportation research record*, 2672(1):1–11.
- [134] Feng, Z., Hu, G., Sun, Y., et Soon, J. (2020). An overview of collaborative robotic manipulation in multi-robot systems. Annual Reviews in Control, 49:113–127.
- [135] Fenton, R. E. (1994). lvhs/ahs: Driving into the future. IEEE Control Systems Magazine, 14(6):13–20.
- [136] Ferscha, A., et Riener, A. (2009). Pervasive adaptation in car crowds. In International Conference on Mobile Wireless Middleware, Operating Systems, and Applications, pages 111–117. Springer.
- [137] Fidanova, S., Roeva, O., et Ganzha, M. (2020). Ant colony optimization algorithm for fuzzy transport modelling. In 2020 15th Conference on Computer Science and Information Systems (FedCSIS), pages 237–240. IEEE.
- [138] Fierro, R., Das, A., Spletzer, J., Esposito, J., Kumar, V., Ostrowski, J. P., Pappas, G., Taylor, C. J., Hur, Y., Alur, R., et others (2002). A framework and architecture for multi-robot coordination. *The International Journal of Robotics Research*, 21(10-11):977–995.
- [139] Finn, A., et Scheding, S. (2010). Developments and challenges for autonomous unmanned vehicles. *Intelligent Systems Reference Library*, 3:128–154.
- [140] Fischer, H.-J. (2015). Standardization and harmonization activities towards a global c-its. In Vehicular ad hoc Networks, pages 23–36. Springer.
- [141] Florin, R., et Olariu, S. (2015). A survey of vehicular communications for traffic signal optimization. Vehicular Communications, 2(2):70–79.
- [142] Fonseca, N., Casanova, J., et Valdés, M. (2011). Influence of the stop/start system on co2 emissions of a diesel vehicle in urban traffic. *Transportation Research Part D: Transport and Environment*, 16(2):194–200.
- [143] Foundation, W. (2021). Eureka prometheus project. available at https://en. wikipedia.org/wiki/Eureka_Prometheus_Project.
- [144] Fraade-Blanar, L., Blumenthal, M. S., Anderson, J. M., et Kalra, N. (2018). Measuring automated vehicle safety: Forging a framework. RAND Corporation, Santa Monica, CA.
- [145] Froetscher, A., et Monschiebl, B. (2018). C-roads: Elements of c-its service evaluation to reach interoperability in europe within a wide stakeholder network: Validation steps and comparative elements used in a living lab environment in austria. In 2018 IEEE 87th Vehicular Technology Conference (VTC Spring), pages 1–5.

- [146] Fukuda, T., Takagawa, I., Sekiyama, K., et Hasegawa, Y. (2000). Hybrid approach of centralized control and distributed control for flexible transfer system. In *Distributed Manipulation*, pages 65–85. Springer.
- [147] Gage, D. W. (1995). Ugv history 101: A brief history of unmanned ground vehicle (ugv) development efforts. Technical Report, Naval Command, Control and Ocean Surveillance Center RDT and E Division CA.
- [148] Gallego, J.-L., Farges, J.-L., et Henry, J.-J. (1996). Design by petri nets of an intersection signal controller. *Transportation Research Part C: Emerging Technologies*, 4(4):231–248.
- [149] Gao, J., Shen, Y., Liu, J., Ito, M., et Shiratori, N. (2017). Adaptive traffic signal control: Deep reinforcement learning algorithm with experience replay and target network. arXiv preprint arXiv:1705.02755.
- [150] García García, A., Camacho Torregrosa, F. J., Lopis Castelló, D., et Monserrat del Rio, J. F. (2021). Smart roads classification. Technical Report, World Road Association – PIARC.
- [151] Gartner (2021). Gartner hyper cycle for artificial intelligence, 2021. available at https://www.gartner.com/en/documents/ 4005076-summary-translation-hype-cycle-for-artificial-intelligence-2021.
- [152] Gartner, N. H. (1983). Opac: A demand-responsive strategy for traffic signal control. In 62nd Annual Meeting of the Transportation Research Board, page 75–84.
- [153] Gartner, N. H., et Wagner, P. (2004). Analysis of traffic flow characteristics on signalized arterials. *Transportation Research Record*, 1883(1):94–100.
- [154] Gassner, M., Cieslewski, T., et Scaramuzza, D. (2017). Dynamic collaboration without communication: Vision-based cable-suspended load transport with two quadrotors. In 2017 IEEE International Conference on Robotics and Automation (ICRA), pages 5196–5202. IEEE.
- [155] Gaur, A., et Mirchandani, P. (2001). Method for real-time recognition of vehicle platoons. *Transportation research record*, 1748(1):8–17.
- [156] Geiger, A., Lauer, M., Moosmann, F., Ranft, B., Rapp, H., Stiller, C., et Ziegler, J. (2012a). Team annieway's entry to the 2011 grand cooperative driving challenge. *IEEE Transactions on Intelligent Transportation Systems*, 13(3):1008–1017.
- [157] Geiger, A., Lenz, P., et Urtasun, R. (2012b). Are we ready for autonomous driving? the kitti vision benchmark suite. In 2012 IEEE conference on computer vision and pattern recognition, pages 3354–3361. IEEE.
- [158] Geroliminis, N., et Daganzo, C. F. (2008). Existence of urban-scale macroscopic fundamental diagrams: Some experimental findings. *Transportation Research Part B: Methodological*, 42(9):759–770.
- [159] Geroliminis, N., Haddad, J., et Ramezani, M. (2012). Optimal perimeter control for two urban regions with macroscopic fundamental diagrams: A model predictive approach. *IEEE Transactions on Intelligent Transportation Systems*, 14(1):348– 359.

- [160] Geroliminis, N., et Sun, J. (2011). Properties of a well-defined macroscopic fundamental diagram for urban traffic. *Transportation Research Part B: Methodological*, 45(3):605–617.
- [161] Ghaffarian, H., Fathy, M., et Soryani, M. (2012). Vehicular ad hoc networks enabled traffic controller for removing traffic lights in isolated intersections based on integer linear programming. *IET intelligent transport systems*, 6(2):115–123.
- [162] Ghiasi, A., Li, X., et Ma, J. (2019). A mixed traffic speed harmonization model with connected autonomous vehicles. *Transportation Research Part C: Emerging Technologies*, 104:210–233.
- [163] Gibson, D., Mills, M. K., et Klein, L. A. (2007). A new look at sensors. Public roads, 71(3).
- [164] Girard, A. R., De Sousa, J. B., et Hedrick, J. K. (2001). An overview of emerging results in networked multi-vehicle systems. In *Proceedings of the 40th IEEE Conference on Decision and Control (Cat. No. 01CH37228)*, volume 2, pages 1485–1490. IEEE.
- [165] Giridhar, A., et Kumar, P. (2006). Scheduling automated traffic on a network of roads. *IEEE Transactions on Vehicular Technology*, 55(5):1467–1474.
- [166] Glaser, S., Vanholme, B., Mammar, S., Gruyer, D., et Nouveliere, L. (2010). Maneuver-based trajectory planning for highly autonomous vehicles on real road with traffic and driver interaction. *IEEE Transactions on intelligent transportation* systems, 11(3):589–606.
- [167] Godfrey, J. (1969). The mechanism of a road network. *Traffic Engineering & Control*, 8(8).
- [168] Gokulan, B. P., et Srinivasan, D. (2010). Distributed geometric fuzzy multiagent urban traffic signal control. IEEE Transactions on Intelligent Transportation Systems, 11(3):714–727.
- [169] Goli, M., et Eskandarian, A. (2014). A systematic multi-vehicle platooning and platoon merging: Strategy, control, and trajectory generation. In *Dynamic Sys*tems and Control Conference, volume 46193, page V002T25A006. American Society of Mechanical Engineers.
- [170] González, D., Pérez, J., Milanés, V., et Nashashibi, F. (2015). A review of motion planning techniques for automated vehicles. *IEEE Transactions on Intelligent Transportation Systems*, 17(4):1135–1145.
- [171] Gonzalez-Martín, M., Sepulcre, M., Molina-Masegosa, R., et Gozalvez, J. (2018). Analytical models of the performance of c-v2x mode 4 vehicular communications. *IEEE Transactions on Vehicular Technology*, 68(2):1155–1166.
- [172] Goodall, N. J., Smith, B. L., et Park, B. (2013). Traffic signal control with connected vehicles. *Transportation Research Record*, 2381(1):65–72.
- [173] Goodrich, M. A., et Schultz, A. C. (2008). Human-robot interaction: a survey. Now Publishers Inc.

- [174] Gradinescu, V., Gorgorin, C., Diaconescu, R., Cristea, V., et Iftode, L. (2007). Adaptive traffic lights using car-to-car communication. In 2007 IEEE 65th vehicular technology conference-VTC2007-Spring, pages 21–25. IEEE.
- [175] Greenblatt, J. B., et Shaheen, S. (2015). Automated vehicles, on-demand mobility, and environmental impacts. Current sustainable/renewable energy reports, 2(3):74–81.
- [176] Greenshields, B., Bibbins, J., Channing, W., et Miller, H. (1935). A study of traffic capacity. In *Highway research board proceedings*, volume 1935. National Research Council (USA), Highway Research Board.
- [177] Gregoire, J., Bonnabel, S., et de La Fortelle, A. (2013). **Optimal cooperative** motion planning for vehicles at intersections. *arXiv preprint arXiv:1310.7729*.
- [178] Gregoire, J., Bonnabel, S., et de La Fortelle, A. (2014a). Priority-based coordination of robots. hal-00828976.
- [179] Gregoire, J., Qian, X., Frazzoli, E., De La Fortelle, A., et Wongpiromsarn, T. (2014b). Capacity-aware backpressure traffic signal control. *IEEE Transactions on Control of Network Systems*, 2(2):164–173.
- [180] Gruyer, D., Magnier, V., Hamdi, K., Claussmann, L., Orfila, O., et Rakotonirainy, A. (2017). Perception, information processing and modeling: Critical stages for autonomous driving applications. *Annual Reviews in Control*, 44:323–341.
- [181] Guanetti, J., Kim, Y., et Borrelli, F. (2018). Control of connected and automated vehicles: State of the art and future challenges. Annual reviews in control, 45:18– 40.
- [182] Guestrin, C., Lagoudakis, M., et Parr, R. (2002). Coordinated reinforcement learning. In *ICML*, volume 2, pages 227–234. Citeseer.
- [183] Guo, Q., Li, L., et Ban, X. J. (2019). Urban traffic signal control with connected and automated vehicles: A survey. *Transportation research part C: emerging technologies*, 101:313–334.
- [184] Haddad, J., Ramezani, M., et Geroliminis, N. (2013). Cooperative traffic control of a mixed network with two urban regions and a freeway. *Transportation Research Part B: Methodological*, 54:17–36.
- [185] Hafez, A., Yousef, A., et Harag, N. (2018). Solar tracking systems: Technologies and trackers drive types–a review. *Renewable and Sustainable Energy Reviews*, 91:754–782.
- [186] Hafner, M. R., Cunningham, D., Caminiti, L., et Del Vecchio, D. (2013). Cooperative collision avoidance at intersections: Algorithms and experiments. IEEE Transactions on Intelligent Transportation Systems, 14(3):1162–1175.
- [187] Haimes, Y. (1971). On a bicriterion formulation of the problems of integrated system identification and system optimization. *IEEE transactions on systems, man, and cybernetics*, 1(3):296–297.

- [188] Hale, D. K., Antoniou, C., Brackstone, M., Michalaka, D., Moreno, A. T., et Parikh, K. (2015). Optimization-based assisted calibration of traffic simulation models. *Transportation Research Part C: Emerging Technologies*, 55:100–115.
- [189] Hall, F. L. (1996). Traffic stream characteristics. *Traffic Flow Theory. US Federal Highway Administration*, 36.
- [190] Hang, P., Lv, C., Huang, C., Xing, Y., et Hu, Z. (2021). Cooperative decision making of connected automated vehicles at multi-lane merging zone: A coalitional game approach. *IEEE Transactions on Intelligent Transportation Systems*.
- [191] Hausknecht, M., Au, T.-C., et Stone, P. (2011). Autonomous intersection management: Multi-intersection optimization. In 2011 IEEE/RSJ International Conference on Intelligent Robots and Systems, pages 4581–4586. IEEE.
- [192] Hawkins, A. J. (2021). A day in the life of a waymo selfdriving taxi. available at https://www.theverge.com/2018/8/21/17762326/ waymo-self-driving-ride-hail-fleet-management.
- [193] Hayward, J. C. (1972). Near miss determination through use of a scale of danger. *Report No. TTSC 7715*.
- [194] He, Q., Head, K. L., et Ding, J. (2012). Pamscod: Platoon-based arterial multimodal signal control with online data. *Transportation Research Part C: Emerging Technologies*, 20(1):164–184.
- [195] He, Z., Qi, G., Lu, L., et Chen, Y. (2019). Network-wide identification of turn-level intersection congestion using only low-frequency probe vehicle data. *Transportation Research Part C: Emerging Technologies*, 108:320–339.
- [196] Hedrick, J. K., Tomizuka, M., et Varaiya, P. (1994). Control issues in automated highway systems. IEEE Control Systems Magazine, 14(6):21–32.
- [197] Hegyi, A., De Schutter, B., et Hellendoorn, J. (2005). Optimal coordination of variable speed limits to suppress shock waves. *IEEE Transactions on intelligent transportation systems*, 6(1):102–112.
- [198] Helbing, D. (2009). Derivation of a fundamental diagram for urban traffic flow. *The European Physical Journal B*, 70(2):229–241.
- [199] Hichri, B., Adouane, L., Fauroux, J.-C., Mezouar, Y., et Doroftei, I. (2016). Cooperative mobile robot control architecture for lifting and transportation of any shape payload. In *Distributed Autonomous Robotic Systems*, pages 177–191. Springer.
- [200] Hichri, B., Adouane, L., Fauroux, J.-c., Mezouar, Y., et Doroftei, Y. (2014). Cooperative lifting and transport by a group of mobile robots. In *International Symposium* on Distributed Autonomous Robotic Systems, DARS 2014.
- [201] Hichri, B., Fauroux, J.-C., Adouane, L., Doroftei, I., et Mezouar, Y. (2019). **Design of cooperative mobile robots for co-manipulation and transportation tasks**. *Robotics and computer-integrated manufacturing*, 57:412–421.
- [202] Hillenbrand, J., Spieker, A. M., et Kroschel, K. (2006). A multilevel collision mitigation approach—its situation assessment, decision making, and performance tradeoffs. *IEEE Transactions on intelligent transportation systems*, 7(4):528–540.

- [203] Hofleitner, A., Herring, R., et Bayen, A. (2012). Arterial travel time forecast with streaming data: A hybrid approach of flow modeling and machine learning. *Transportation Research Part B: Methodological*, 46(9):1097–1122.
- [204] Hong, S., Malikopoulos, A. A., Lee, J., et Park, B. B. (2017). Development and evaluation of speed harmonization using optimal control theory: A simulationbased case study at a speed reduction zone 2. In *Proc. 96th Annu. Meeting Transp. Res. Board (TRB).*
- [205] Horst, R. (1991). Time-to-collision as a cue for decision-making in braking. Vision in Vehicles-III.
- [206] Hu, J., Bhowmick, P., Jang, I., Arvin, F., et Lanzon, A. (2021). A decentralized cluster formation containment framework for multirobot systems. *IEEE Transactions on Robotics*.
- [207] Hu, L., Ou, J., Huang, J., Chen, Y., et Cao, D. (2020). A review of research on traffic conflicts based on intelligent vehicles. *leee Access*, 8:24471–24483.
- [208] Huang, C.-F., Bieniawski, S., Wolpert, D. H., et Strauss, C. E. (2005). A comparative study of probability collectives based multi-agent systems and genetic algorithms. In Proceedings of the 7th annual conference on Genetic and evolutionary computation, pages 751–752.
- [209] Huang, G. (2017). Smart signal light settled on jinan, china. available at https: //news.e23.cn/jnnews/2017-04-20/2017042000555.html.
- [210] Huang, S., Sadek, A. W., et Zhao, Y. (2012). Assessing the mobility and environmental benefits of reservation-based intelligent intersections using an integrated simulator. IEEE Transactions on Intelligent Transportation Systems, 13(3):1201–1214.
- [211] Huang, X., Wang, P., Cheng, X., Zhou, D., Geng, Q., et Yang, R. (2019). The apolloscape open dataset for autonomous driving and its application. *IEEE transactions on pattern analysis and machine intelligence*, 42(10):2702–2719.
- [212] Hult, R., Campos, G. R., Falcone, P., et Wymeersch, H. (2015). An approximate solution to the optimal coordination problem for autonomous vehicles at intersections. In 2015 American Control Conference (ACC), pages 763–768. IEEE.
- [213] Hunt, P., Robertson, D., Bretherton, R., et Royle, M. C. (1982). The scoot on-line traffic signal optimisation technique. *Traffic Engineering & Control*, 23(4):190–192.
- [214] Hussain, S., Peng, Z., et Hayee, M. I. (2019). Development and demonstration of merge assist system using connected vehicle technology. Technical Report, Center for Transportation Studies, University of Minnesota.
- [215] Hydén, C. (1987). The development of a method for traffic safety evaluation: The swedish traffic conflicts technique. Bulletin Lund Institute of Technology, Department, 1(70).
- [216] Hyland, M., et Mahmassani, H. S. (2018). Dynamic autonomous vehicle fleet operations: Optimization-based strategies to assign avs to immediate traveler demand requests. *Transportation Research Part C: Emerging Technologies*, 92:278– 297.

- [217] Iberraken, D., Adouane, L., et Denis, D. (2018a). Multi-level bayesian decisionmaking for safe and flexible autonomous navigation in highway environment. In 2018 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pages 3984–3990. IEEE.
- [218] Iberraken, D., Adouane, L., et Denis, D. (2018b). Safe autonomous overtaking maneuver based on inter-vehicular distance prediction and multi-level bayesian decision-making. In 2018 21st International Conference on Intelligent Transportation Systems (ITSC), pages 3259–3265. IEEE.
- [219] IEEE (2021). Look ma, no hands! available at https://www.ieee.org/about/news/ 2012/5september-2-2012.html.
- [220] Institute of Transportation Studies (2021). California partners for advanced transportation technology. available at https://path.berkeley.edu/.
- [221] Institute of Transportation Studies (ITS) (2020). California path biennial report 2018-2020. Standard, University of California, Berkeley, United States.
- [222] International, S. (2016). Dedicated short range communications (dsrc) message set dictionary. Standard, SAE International, United States.
- [223] International, S. (2017). Dedicated short range communication (dsrc) systems engineering process guidance for sae j2945/x documents and common design conceptstm. Standard, SAE International, United States.
- [224] International, S. (2020a). Candidate improvements to dedicated short range communications (dsrc) message set dictionary (sae j2735) using systems engineering methods. Standard, SAE International, United States.
- [225] International, S. (2020b). V2x communications message set dictionary. Standard, SAE International, United States.
- [226] Jameel, F., Chang, Z., Huang, J., et Ristaniemi, T. (2019). Internet of autonomous vehicles: architecture, features, and socio-technological challenges. *IEEE Wireless Communications*, 26(4):21–29.
- [227] Janai, J., Güney, F., Behl, A., Geiger, A., et others (2020). Computer vision for autonomous vehicles: Problems, datasets and state of the art. Foundations and Trends® in Computer Graphics and Vision, 12(1–3):1–308.
- [228] Jansen, K. (2004). Approximation algorithms for the general max-min resource sharing problem: faster and simpler. In *Scandinavian Workshop on Algorithm Theory*, pages 311–322. Springer.
- [229] Ji, Y., et Geroliminis, N. (2012). On the spatial partitioning of urban transportation networks. *Transportation Research Part B: Methodological*, 46(10):1639–1656.
- [230] Jia, B., Jiang, R., et Wu, Q.-S. (2004). Traffic behavior near an off ramp in the cellular automaton traffic model. *Physical Review E*, 69(5):056105.
- [231] Jiang, H., Hu, J., An, S., Wang, M., et Park, B. B. (2017a). Eco approaching at an isolated signalized intersection under partially connected and automated vehicles environment. *Transportation Research Part C: Emerging Technologies*, 79:290– 307.

- [232] Jiang, Y., Li, S., et Shamo, D. E. (2006). A platoon-based traffic signal timing algorithm for major-minor intersection types. *Transportation Research Part B: Methodological*, 40(7):543–562.
- [233] Jiang, Y., Zanon, M., Hult, R., et Houska, B. (2017b). Distributed algorithm for optimal vehicle coordination at traffic intersections. *IFAC-PapersOnLine*, 50(1):11577–11582.
- [234] Jin, Q., Wu, G., Boriboonsomsin, K., et Barth, M. (2012). Multi-agent intersection management for connected vehicles using an optimal scheduling approach. In 2012 International Conference on Connected Vehicles and Expo (ICCVE), pages 185– 190. IEEE.
- [235] Jin, Q., Wu, G., Boriboonsomsin, K., et Barth, M. (2013). Platoon-based multiagent intersection management for connected vehicle. In 16th international ieee conference on intelligent transportation systems (itsc 2013), pages 1462–1467. IEEE.
- [236] Jones, H., et Snyder, M. (2001). Supervisory control of multiple robots based on a real-time strategy game interaction paradigm. In 2001 IEEE International Conference on Systems, Man and Cybernetics. e-Systems and e-Man for Cybernetics in Cyberspace (Cat. No. 01CH37236), volume 1, pages 383–388. IEEE.
- [237] Jorge, V. A., Granada, R., Maidana, R. G., Jurak, D. A., Heck, G., Negreiros, A. P., Dos Santos, D. H., Gonçalves, L. M., et Amory, A. M. (2019). A survey on unmanned surface vehicles for disaster robotics: Main challenges and directions. Sensors, 19(3):702.
- [238] Kamal, M. A. S., Imura, J.-i., Hayakawa, T., Ohata, A., et Aihara, K. (2014). A vehicle-intersection coordination scheme for smooth flows of traffic without using traffic lights. *IEEE Transactions on Intelligent Transportation Systems*, 16(3):1136–1147.
- [239] Kamal, M. A. S., Imura, J.-i., Ohata, A., Hayakawa, T., et Aihara, K. (2013). Coordination of automated vehicles at a traffic-lightless intersection. In 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), pages 922–927. leee.
- [240] Kamal, M. A. S., Mukai, M., Murata, J., et Kawabe, T. (2012). Model predictive control of vehicles on urban roads for improved fuel economy. *IEEE Transactions* on control systems technology, 21(3):831–841.
- [241] Karagiannis, G., Altintas, O., Ekici, E., Heijenk, G., Jarupan, B., Lin, K., et Weil, T. (2011). Vehicular networking: A survey and tutorial on requirements, architectures, challenges, standards and solutions. *IEEE communications surveys & tutorials*, 13(4):584–616.
- [242] Kato, S., Takeuchi, E., Ishiguro, Y., Ninomiya, Y., Takeda, K., et Hamada, T. (2015). An open approach to autonomous vehicles. *IEEE Micro*, 35(6):60–68.
- [243] Kato, S., Tsugawa, S., Tokuda, K., Matsui, T., et Fujii, H. (2002). Vehicle control algorithms for cooperative driving with automated vehicles and intervehicle communications. *IEEE Transactions on intelligent transportation systems*, 3(3):155– 161.

- [244] Katrakazas, C., Quddus, M., Chen, W.-H., et Deka, L. (2015). Real-time motion planning methods for autonomous on-road driving: State-of-the-art and future research directions. *Transportation Research Part C: Emerging Technologies*, 60:416–442.
- [245] Katriniok, A., Kleibaum, P., Ress, C., et Eckstein, L. (2017). Automation of road vehicles using v2x: An application to intersection automation. Technical Report, SAE Technical Paper.
- [246] Kerner, B. S., Klenov, S. L., et Schreckenberg, M. (2011). Simple cellular automaton model for traffic breakdown, highway capacity, and synchronized flow. *Physical Review E*, 84(4):046110.
- [247] Keyvan-Ekbatani, M., Kouvelas, A., Papamichail, I., et Papageorgiou, M. (2012). Exploiting the fundamental diagram of urban networks for feedback-based gating. *Transportation Research Part B: Methodological*, 46(10):1393–1403.
- [248] Khondaker, B., et Kattan, L. (2015). Variable speed limit: A microscopic analysis in a connected vehicle environment. Transportation Research Part C: Emerging Technologies, 58:146–159.
- [249] Khoshnevis, B., et Bekey, G. (1998). Centralized sensing and control of multiple mobile robots. *Computers & industrial engineering*, 35(3-4):503–506.
- [250] Kianfar, R., Augusto, B., Ebadighajari, A., Hakeem, U., Nilsson, J., Raza, A., Tabar, R. S., Irukulapati, N. V., Englund, C., Falcone, P., et others (2012). Design and experimental validation of a cooperative driving system in the grand cooperative driving challenge. *IEEE transactions on intelligent transportation systems*, 13(3):994–1007.
- [251] Kim, K.-D., et Kumar, P. R. (2014a). An mpc-based approach to provable system-wide safety and liveness of autonomous ground traffic. *IEEE Transactions on Automatic Control*, 59(12):3341–3356.
- [252] Kim, S.-W., Liu, W., Ang, M. H., Frazzoli, E., et Rus, D. (2015). The impact of cooperative perception on decision making and planning of autonomous vehicles. IEEE Intelligent Transportation Systems Magazine, 7(3):39–50.
- [253] Kim, S.-W., Qin, B., Chong, Z. J., Shen, X., Liu, W., Ang, M. H., Frazzoli, E., et Rus, D. (2014b). Multivehicle cooperative driving using cooperative perception: Design and experimental validation. *IEEE Transactions on Intelligent Transportation Systems*, 16(2):663–680.
- [254] Klauer, S. G., Guo, F., Sudweeks, J., et Dingus, T. A. (2010). An analysis of driver inattention using a case-crossover approach on 100-car data. Report HS-811 334, Nat. Highway Traffic Saf. Admin., Washington, DC, USA, Tech. Rep.
- [255] Knoop, V., Hoogendoorn, S. P., et van Zuylen, H. (2009). Empirical differences between time mean speed and space mean speed. In *Traffic and Granular Flow'07*, pages 351–356. Springer.
- [256] Koenig, N., et Howard, A. (2004). Design and use paradigms for gazebo, an open-source multi-robot simulator. In 2004 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS)(IEEE Cat. No. 04CH37566), volume 3, pages 2149–2154.

- [257] KOROSEC, K. (2021). Alphabet leads 1 billion investment in lyft. available at https://fortune.com/2017/10/19/alphabet-leads-1-billion-investment-in-lyft/.
- [258] Kousaridas, A., Schimpe, A., Euler, S., Vilajosana, X., Fallgren, M., Landi, G., Moscatelli, F., Barmpounakis, S., Vázquez-Gallego, F., Sedar, R., et others (2020).
 5g cross-border operation for connected and automated mobility: Challenges and solutions. *Future Internet*, 12(1):5.
- [259] Krauß, S., Wagner, P., et Gawron, C. (1997). Metastable states in a microscopic model of traffic flow. *Physical Review E*, 55(5):5597.
- [260] Kulkarni, A. J., Patankar, N. S., et Tai, K. (2016). Constraint handling in probability collectives using a modified feasibility-based rule. International Journal of Computational Science and Engineering, 13(4):303–321.
- [261] Kulkarni, A. J., et Tai, K. (2010a). Probability collectives: a distributed optimization approach for constrained problems. In *IEEE Congress on Evolutionary Computation*, pages 1–8.
- [262] Kulkarni, A. J., et Tai, K. (2010b). Probability collectives: a multi-agent approach for solving combinatorial optimization problems. *Applied Soft Computing*, 10(3):759–771.
- [263] Kulkarni, A. J., et Tai, K. (2011). Solving constrained optimization problems using probability collectives and a penalty function approach. International Journal of Computational Intelligence and Applications, 10(04):445–470.
- [264] Künhe, F., Gomes, J., et Fetter, W. (2005). Mobile robot trajectory tracking using model predictive control. In *II IEEE latin-american robotics symposium*, volume 51. Citeseer.
- [265] Laird, D., Raptis, I. A., et Price, J. (2014). Design and validation of a centimeterscale robot collective. In 2014 IEEE International Conference on Systems, Man, and Cybernetics (SMC), pages 918–923.
- [266] Lakhal, N. M. B. (2021). Risk Assessment and Management based on Interval Analysis for Safe and Reliable Navigation of Intelligent Vehicles. PhD thesis, Université Clermont Auvergne (2017-2021).
- [267] Lategahn, H., Geiger, A., et Kitt, B. (2011). Visual slam for autonomous ground vehicles. In 2011 IEEE International Conference on Robotics and Automation, pages 1732–1737.
- [268] Latombe, J.-C. (2012). Robot motion planning, volume 124. Springer Science & Business Media.
- [269] Le-Anh, T., et De Koster, M. (2006). A review of design and control of automated guided vehicle systems. *European Journal of Operational Research*, 171(1):1–23.
- [270] Lee, J., Gutesa, S., Dimitrijevic, B., Zhang, Y., Spasovic, L., et Singh, J. (2017). Deployment and field evaluation of in-vehicle traffic signal advisory system (itsas). *Information*, 8(3):72.

- [271] Lee, J., et Park, B. (2012). Development and evaluation of a cooperative vehicle intersection control algorithm under the connected vehicles environment. *IEEE Transactions on Intelligent Transportation Systems*, 13(1):81–90.
- [272] Lee, J., Park, B., et Yun, I. (2013a). Cumulative travel-time responsive real-time intersection control algorithm in the connected vehicle environment. *Journal of Transportation Engineering*, 139(10):1020–1029.
- [273] Lee, J., Park, B. B., Malakorn, K., et So, J. J. (2013b). Sustainability assessments of cooperative vehicle intersection control at an urban corridor. *Transportation Research Part C: Emerging Technologies*, 32:193–206.
- [274] Lee, W.-H., Jeng, B.-S., Tseng, S.-S., et Wang, C.-H. (2004). Electronic toll collection based on vehicle-positioning system techniques. In *IEEE International Conference on Networking, Sensing and Control, 2004*, volume 1, pages 643–648.
- [275] Legacy, C., Ashmore, D., Scheurer, J., Stone, J., et Curtis, C. (2019). Planning the driverless city. *Transport reviews*, 39(1):84–102.
- [276] Leitner, J. (2009). Multi-robot cooperation in space: A survey. In 2009 Advanced Technologies for Enhanced Quality of Life, pages 144–151. IEEE.
- [277] Levin, M. W., Fritz, H., et Boyles, S. D. (2016). On optimizing reservationbased intersection controls. *IEEE Transactions on Intelligent Transportation Systems*, 18(3):505–515.
- [278] Levin, M. W., et Rey, D. (2017). Conflict-point formulation of intersection control for autonomous vehicles. *Transportation Research Part C: Emerging Technologies*, 85:528–547.
- [279] Li, H., Tsukada, M., Nashashibi, F., et Parent, M. (2014). Multivehicle cooperative local mapping: A methodology based on occupancy grid map merging. *IEEE Transactions on Intelligent Transportation Systems*, 15(5):2089–2100.
- [280] Li, J., Ma, H., et Tomizuka, M. (2019). Conditional generative neural system for probabilistic trajectory prediction. In 2019 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pages 6150–6156. IEEE.
- [281] Li, J., Yang, F., Tomizuka, M., et Choi, C. (2020a). Evolvegraph: Multiagent trajectory prediction with dynamic relational reasoning. *arXiv* preprint *arXiv:2003.13924*.
- [282] Li, L., Huang, W.-L., Liu, Y., Zheng, N.-N., et Wang, F.-Y. (2016). Intelligence testing for autonomous vehicles: A new approach. *IEEE Transactions on Intelligent Vehicles*, 1(2):158–166.
- [283] Li, L., et Wang, F.-Y. (2006). Cooperative driving at blind crossings using intervehicle communication. *IEEE Transactions on Vehicular technology*, 55(6):1712– 1724.
- [284] Li, L., Wen, D., et Yao, D. (2013). A survey of traffic control with vehicular communications. IEEE Transactions on Intelligent Transportation Systems, 15(1):425– 432.
- [285] Li, N., Yao, Y., Kolmanovsky, I., Atkins, E., et Girard, A. R. (2020b). Game-theoretic modeling of multi-vehicle interactions at uncontrolled intersections. IEEE Transactions on Intelligent Transportation Systems.
- [286] Li, S., Li, K., Rajamani, R., et Wang, J. (2010). Model predictive multi-objective vehicular adaptive cruise control. *IEEE Transactions on control systems technology*, 19(3):556–566.
- [287] Li, S. E., Zheng, Y., Li, K., Wu, Y., Hedrick, J. K., Gao, F., et Zhang, H. (2017). Dynamical modeling and distributed control of connected and automated vehicles: Challenges and opportunities. *IEEE Intelligent Transportation Systems Magazine*, 9(3):46–58.
- [288] Li, Y., Fadda, E., Manerba, D., Tadei, R., et Terzo, O. (2020c). Reinforcement learning algorithms for online single-machine scheduling. In 2020 15th Conference on Computer Science and Information Systems (FedCSIS), pages 277–283. IEEE.
- [289] Li, Y. J. (2010). An overview of the dsrc/wave technology. In International Conference on Heterogeneous Networking for Quality, Reliability, Security and Robustness, pages 544–558. Springer.
- [290] Lian, F., Chen, B., Zhang, K., Miao, L., Wu, J., et Luan, S. (2021). Adaptive traffic signal control algorithms based on probe vehicle data. *Journal of Intelligent Transportation Systems*, 25(1):41–57.
- [291] Liang, X., Guler, S. I., et Gayah, V. V. (2018). Signal timing optimization with connected vehicle technology: Platooning to improve computational efficiency. *Transportation Research Record*, 2672(18):81–92.
- [292] Lin, P., Liu, J., Jin, P. J., et Ran, B. (2017a). Autonomous vehicle-intersection coordination method in a connected vehicle environment. *IEEE Intelligent Transportation Systems Magazine*, 9(4):37–47.
- [293] Lin, S.-H., et Ho, T.-Y. (2019). Autonomous vehicle routing in multiple intersections. In Proceedings of the 24th Asia and South Pacific Design Automation Conference, pages 585–590.
- [294] Lin, T.-Y., Maire, M., Belongie, S., Hays, J., Perona, P., Ramanan, D., Dollár, P., et Zitnick, C. L. (2014). Microsoft coco: Common objects in context. In European conference on computer vision, pages 740–755. Springer.
- [295] Lin, W.-H., et Wang, C. (2004). An enhanced 0-1 mixed-integer lp formulation for traffic signal control. *IEEE Transactions on Intelligent transportation systems*, 5(4):238–245.
- [296] Lin, Y., et Eskandarian, A. (2017b). Experimental evaluation of cooperative adaptive cruise control with autonomous mobile robots. In 2017 IEEE Conference on Control Technology and Applications (CCTA), pages 281–286. IEEE.
- [297] Lindemann, R. A., et Voorhees, C. J. (2005). Mars exploration rover mobility assembly design, test and performance. In 2005 IEEE International Conference on Systems, Man and Cybernetics, volume 1, pages 450–455.

- [298] Lindsey, Q., Mellinger, D., et Kumar, V. (2012). Construction with quadrotor teams. *Autonomous Robots*, 33(3):323–336.
- [299] Lioris, J., Pedarsani, R., Tascikaraoglu, F. Y., et Varaiya, P. (2017). Platoons of connected vehicles can double throughput in urban roads. *Transportation Research Part C: Emerging Technologies*, 77:292–305.
- [300] List, G. F., et Cetin, M. (2004). Modeling traffic signal control using petri nets. *IEEE Transactions on intelligent transportation systems*, 5(3):177–187.
- [301] Little, J., Kelson, M., et Gartner, N. (1981). Maxband: A program for setting signals on arteries and triangular networks. transportation research record 795: Trb, national research council. washington. Technical Report, U.S. Dept. Transp.,Washington, DC,Transp. Res.
- [302] Little, J. D. (1966). The synchronization of traffic signals by mixed-integer linear programming. *Operations Research*, 14(4):568–594.
- [303] LLC, W. (2021). Waymo. available at https://waymo.com/.
- [304] Loke, S. W. (2019). Cooperative automated vehicles: A review of opportunities and challenges in socially intelligent vehicles beyond networking. *IEEE Transactions on Intelligent Vehicles*, 4(4):509–518.
- [305] Lord, D., et Persaud, B. N. (2004). Estimating the safety performance of urban road transportation networks. *Accident Analysis & Prevention*, 36(4):609–620.
- [306] Lowrie, P. (1982). Scats: The sydney co-ordinated adaptive traffic system: Principles, methodology, algorithms. In *IEEE International Conference on Road Traffic Signalling, London, England*, page 67–70.
- [307] Lozenguez, G., Adouane, L., Beynier, A., Martinet, P., et Mouaddib, A.-I. (2011). Map partitioning to approximate an exploration strategy in mobile robotics. In Advances on practical applications of agents and multiagent systems, pages 63–72. Springer.
- [308] Lu, G., Li, L., Wang, Y., Zhang, R., Bao, Z., et Chen, H. (2014). A rule based control algorithm of connected vehicles in uncontrolled intersection. In *17th international IEEE conference on intelligent transportation systems (itsc)*, pages 115–120. IEEE.
- [309] Lu, X.-Y., Tan, H.-S., Shladover, S. E., et Hedrick, J. K. (2004). Automated vehicle merging maneuver implementation for ahs. Vehicle System Dynamics, 41(2):85– 107.
- [310] Ma, J., Li, X., Zhou, F., Hu, J., et Park, B. B. (2017). Parsimonious shooting heuristic for trajectory design of connected automated traffic part ii: computational issues and optimization. *Transportation Research Part B: Methodological*, 95:421–441.
- [311] Ma, Y., Wang, H., Xie, Y., et Guo, M. (2014). Path planning for multiple mobile robots under double-warehouse. *Information Sciences*, 278:357–379.

- [312] MacHardy, Z., Khan, A., Obana, K., et Iwashina, S. (2018). V2x access technologies: Regulation, research, and remaining challenges. *IEEE Communications Surveys & Tutorials*, 20(3):1858–1877.
- [313] Maddern, W., Pascoe, G., Linegar, C., et Newman, P. (2017). 1 year, 1000 km: The oxford robotcar dataset. The International Journal of Robotics Research, 36(1):3–15.
- [314] Mahmassani, H. S. (2016). 50th anniversary invited article—autonomous vehicles and connected vehicle systems: Flow and operations considerations. *Transportation Science*, 50(4):1140–1162.
- [315] Mahmassani, H. S., Hou, T., et Saberi, M. (2013). Connecting networkwide travel time reliability and the network fundamental diagram of traffic flow. *Transportation* research record, 2391(1):80–91.
- [316] Makarem, L., et Gillet, D. (2013). Model predictive coordination of autonomous vehicles crossing intersections. In 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), pages 1799–1804.
- [317] Malik, S., Khan, M. A., et El-Sayed, H. (2021). Collaborative autonomous driving—a survey of solution approaches and future challenges. Sensors, 21(11):3783.
- [318] Malikopoulos, A. A., Beaver, L., et Chremos, I. V. (2021). Optimal time trajectory and coordination for connected and automated vehicles. *Automatica*, 125:109469.
- [319] Manzinger, S., et Althoff, M. (2018). Tactical decision making for cooperative vehicles using reachable sets. In 2018 21st International Conference on Intelligent Transportation Systems (ITSC), pages 444–451. IEEE.
- [320] material handling, et logistics (2021). rise of the warehouse robots. available at https://www.mhlnews.com/technology-automation/article/22054632/ rise-of-the-warehouse-robots.
- [321] Mattingley, J., Wang, Y., et Boyd, S. (2011). Receding horizon control. *IEEE Control Systems Magazine*, 31(3):52–65.
- [322] Mauro, V., et Di Taranto, C. (1990). Utopia. IFAC Proceedings Volumes, 23(2):245– 252.
- [323] Mavrotas, G. (2009). Effective implementation of the ε-constraint method in multi-objective mathematical programming problems. Applied mathematics and computation, 213(2):455–465.
- [324] McKenney, D., et White, T. (2013). Distributed and adaptive traffic signal control within a realistic traffic simulation. Engineering Applications of Artificial Intelligence, 26(1):574–583.
- [325] McLurkin, J., McMullen, A., Robbins, N., Habibi, G., Becker, A., Chou, A., Li, H., John, M., Okeke, N., Rykowski, J., et others (2014). A robot system design for low-cost multi-robot manipulation. In 2014 IEEE/RSJ international conference on intelligent robots and systems, pages 912–918. IEEE.
- [326] Medeiros, A. A. (1998). A survey of control architectures for autonomous mobile robots. *Journal of the Brazilian Computer Society*, 4:35–43.

- [327] Medina, A. I. M., van de Wouw, N., et Nijmeijer, H. (2017). Cooperative intersection control based on virtual platooning. *IEEE Transactions on Intelligent Transportation Systems*, 19(6):1727–1740.
- [328] Mehrabipour, M., et Hajbabaie, A. (2017). A cell-based distributed-coordinated approach for network-level signal timing optimization. *Computer-Aided Civil and Infrastructure Engineering*, 32(7):599–616.
- [329] Meng, Y., Li, L., Wang, F.-Y., Li, K., et Li, Z. (2017). Analysis of cooperative driving strategies for nonsignalized intersections. *IEEE Transactions on Vehicular Technology*, 67(4):2900–2911.
- [330] Miettinen, K. (2012). Nonlinear multiobjective optimization, volume 12. Springer Science & Business Media.
- [331] Milakis, D., Van Arem, B., et Van Wee, B. (2017). Policy and society related implications of automated driving: A review of literature and directions for future research. *Journal of Intelligent Transportation Systems*, 21(4):324–348.
- [332] Milanés, V., Godoy, J., Villagrá, J., et Pérez, J. (2010). Automated on-ramp merging system for congested traffic situations. *IEEE Transactions on Intelligent Transportation Systems*, 12(2):500–508.
- [333] Milanés, V., Villagrá, J., Godoy, J., Simó, J., Pérez, J., et Onieva, E. (2012). An intelligent v2i-based traffic management system. *IEEE Transactions on Intelligent Transportation Systems*, 13(1):49–58.
- [334] Min, W., et Wynter, L. (2011). Real-time road traffic prediction with spatiotemporal correlations. Transportation Research Part C: Emerging Technologies, 19(4):606–616.
- [335] Mirchandani, P. B., et Zou, N. (2007). Queuing models for analysis of traffic adaptive signal control. *IEEE Transactions on Intelligent Transportation Systems*, 8(1):50–59.
- [336] Mirheli, A., Tajalli, M., Hajibabai, L., et Hajbabaie, A. (2019). A consensus-based distributed trajectory control in a signal-free intersection. *Transportation research part C: emerging technologies*, 100:161–176.
- [337] M.Minderhoud (2022). Drip, dynamic route information panel, used in the netherlands for route information. available at https://commons.wikimedia.org/wiki/ Main_Page.
- [338] Mnih, V., Kavukcuoglu, K., Silver, D., Rusu, A. A., Veness, J., Bellemare, M. G., Graves, A., Riedmiller, M., Fidjeland, A. K., Ostrovski, G., et others (2015). Humanlevel control through deep reinforcement learning. *nature*, 518(7540):529–533.
- [339] Mouad, M. (2014). Architecture de COntrôle/COmmande dédiée aux systèmes Distribués Autonomes (ACO²DA): application à une plate-forme multi-véhicules. PhD thesis, Clermont-Ferrand 2.
- [340] Mueller, R. P., Cox, R. E., Ebert, T., Smith, J. D., Schuler, J. M., et Nick, A. J. (2013). Regolith advanced surface systems operations robot (rassor). In 2013 IEEE Aerospace Conference, pages 1–12.

- [341] Muneer, T., Kolhe, M., et Doyle, A. (2017). Electric vehicles: prospects and challenges. Elsevier.
- [342] Munoz, J. C., et Daganzo, C. F. (2002). The bottleneck mechanism of a freeway diverge. *Transportation Research Part A: Policy and Practice*, 36(6):483–505.
- [343] Murphy, R. R. (2012). A decade of rescue robots. In 2012 IEEE/RSJ International Conference on Intelligent Robots and Systems, pages 5448–5449. IEEE.
- [344] Nagel, K., et Schreckenberg, M. (1992). A cellular automaton model for freeway traffic. Journal de physique I, 2(12):2221–2229.
- [345] Namazi, E., Li, J., et Lu, C. (2019). Intelligent intersection management systems considering autonomous vehicles: A systematic literature review. *IEEE Access*, 7:91946–91965.
- [346] Nasri, O., Ben Lakhal, N. M., Adouane, L., et Ben Hadj Slama, J. (2019). Automotive decentralized diagnosis based on can real-time analysis. *Journal of Systems Architecture*, 98:249–258.
- [347] Naumann, R., Rasche, R., et Tacken, J. (1998). Managing autonomous vehicles at intersections. *IEEE Intelligent Systems and their Applications*, 13(3):82–86.
- [348] Naumann, R., Rasche, R., Tacken, J., et Tahedi, C. (1997). Validation and simulation of a decentralized intersection collision avoidance algorithm. In *Proceedings* of Conference on Intelligent Transportation Systems, pages 818–823. IEEE.
- [349] Navy, U. (2004). The navy unmanned undersea vehicle master plan ().
- [350] Nayyar, A., Puri, V., Nguyen, N. G., et Le, D. N. (2018). Smart surveillance robot for real-time monitoring and control system in environment and industrial applications. In *Information Systems Design and Intelligent Applications*, pages 229–243. Springer.
- [351] Newell, G. F. (1993). A simplified theory of kinematic waves in highway traffic, part i: General theory. *Transportation Research Part B: Methodological*, 27(4):281–287.
- [352] NHTSA (2022). What is vehicle cybersecurity? available at https://www.nhtsa. gov/technology-innovation/vehicle-cybersecurity.
- [353] Niazi, M., et Hussain, A. (2011). Agent-based computing from multi-agent systems to agent-based models: a visual survey. *Scientometrics*, 89(2):479.
- [354] Nicolescu, M. N., et Matarić, M. J. (2002). A hierarchical architecture for behavior-based robots. In Proceedings of the first international joint conference on Autonomous agents and multiagent systems: part 1, pages 227–233.
- [355] Niittymäki, J., et Pursula, M. (2000). Signal control using fuzzy logic. Fuzzy sets and systems, 116(1):11–22.
- [356] Nilsson, N. J., et Nilsson, N. J. (1998). Artificial intelligence: a new synthesis. Morgan Kaufmann.

- [357] Nilsson, P., Haesaert, S., Thakker, R., Otsu, K., Vasile, C.-I., Agha-Mohammadi, A.-A., Murray, R. M., et Ames, A. D. (2018). Toward specification-guided active mars exploration for cooperative robot teams. *Robotics: Science and Systems Foundation*, pages 1–9.
- [358] Noh, S. (2018). Decision-making framework for autonomous driving at road intersections: Safeguarding against collision, overly conservative behavior, and violation vehicles. *IEEE Transactions on Industrial Electronics*, 66(4):3275–3286.
- [359] Noreils, F. R. (1993). Toward a robot architecture integrating cooperation between mobile robots: Application to indoor environment. *The International Journal* of *Robotics Research*, 12(1):79–98.
- [360] Ntousakis, I. A., Nikolos, I. K., et Papageorgiou, M. (2016). Optimal vehicle trajectory planning in the context of cooperative merging on highways. *Transportation research part C: emerging technologies*, 71:464–488.
- [361] NVIDIA (2021). Nvidia drive sdk. available at https://developer.nvidia.com/drive/ drive-sdk.
- [362] Olia, A., Abdelgawad, H., Abdulhai, B., et Razavi, S. N. (2016). Assessing the potential impacts of connected vehicles: mobility, environmental, and safety perspectives. *Journal of Intelligent Transportation Systems*, 20(3):229–243.
- [363] Papadoulis, A., Quddus, M., et Imprialou, M. (2019). Evaluating the safety impact of connected and autonomous vehicles on motorways. Accident Analysis & Prevention, 124:12–22.
- [364] Papageorgiou, M., Diakaki, C., Dinopoulou, V., Kotsialos, A., et Wang, Y. (2003). Review of road traffic control strategies. *Proceedings of the IEEE*, 91(12):2043–2067.
- [365] Papageorgiou, M., Kosmatopoulos, E., et Papamichail, I. (2008). Effects of variable speed limits on motorway traffic flow. *Transportation Research Record*, 2047(1):37–48.
- [366] Papageorgiou, M., et Kotsialos, A. (2002). Freeway ramp metering: An overview. *IEEE transactions on intelligent transportation systems*, 3(4):271–281.
- [367] Park, H., Su, S., Hayat, M. T., et Smith, B. L. (2014). A prototype freeway merging control algorithm under a connected vehicle environment. In *Transportation Research Board 93rd Annual Meeting*.
- [368] Parker, L. E. (2000). Lifelong adaptation in heterogeneous multi-robot teams: Response to continual variation in individual robot performance. *Autonomous Robots*, 8(3):239–267.
- [369] Parker, L. E., Rus, D., et Sukhatme, G. S. (2016). Multiple mobile robot systems. In *Springer Handbook of Robotics*, pages 1335–1384. Springer.
- [370] Paulius, D., et Sun, Y. (2019). A survey of knowledge representation in service robotics. *Robotics and Autonomous Systems*, 118:13–30.

- [371] Pendleton, S. D., Andersen, H., Du, X., Shen, X., Meghjani, M., Eng, Y. H., Rus, D., et Ang, M. H. (2017). Perception, planning, control, and coordination for autonomous vehicles. *Machines*, 5(1):6.
- [372] Perronnet, F., Abbas-Turki, A., et El Moudni, A. (2014). Vehicle routing through deadlock-free policy for cooperative traffic control in a network of intersections: Reservation and congestion. In 17th international IEEE conference on intelligent transportation systems (ITSC), pages 2233–2238.
- [373] Philippe, C., Adouane, L., Tsourdos, A., Shin, H.-S., et Thuilot, B. (2019). Probability collectives algorithm applied to decentralized intersection coordination for connected autonomous vehicles. In 2019 IEEE Intelligent Vehicles Symposium (IV), pages 1928–1934.
- [374] PIARC (2022). Traffic control centres. available at https://rno-its.piarc.org/en/ network-operations-rno-activities/traffic-control-centres.
- [375] Ploeg, J., Englund, C., Nijmeijer, H., Semsar-Kazerooni, E., Shladover, S. E., Voronov, A., et Van de Wouw, N. (2018). Guest editorial introduction to the special issue on the 2016 grand cooperative driving challenge. *IEEE Transactions on Intelligent Transportation Systems*, 19(4):1208–1212.
- [376] Ploeg, J., Shladover, S., Nijmeijer, H., et van de Wouw, N. (2012). Introduction to the special issue on the 2011 grand cooperative driving challenge. IEEE Transactions on Intelligent Transportation Systems, 13(3):989–993.
- [377] Poczter, S. L., Jankovic, L. M., et others (2014). The google car: driving toward a better future? *Journal of Business Case Studies (JBCS)*, 10(1):7–14.
- [378] Poudel, D. B. (2013). Coordinating hundreds of cooperative, autonomous robots in a warehouse. *Jan*, 27(1-13):26.
- [379] Pourmehrab, M., Elefteriadou, L., Ranka, S., et Martin-Gasulla, M. (2019). Optimizing signalized intersections performance under conventional and automated vehicles traffic. *IEEE Transactions on intelligent transportation systems*, 21(7):2864– 2873.
- [380] Pueboobpaphan, R., Liu, F., et van Arem, B. (2010). The impacts of a communication based merging assistant on traffic flows of manual and equipped vehicles at an on-ramp using traffic flow simulation. In 13th International IEEE Conference on Intelligent Transportation Systems, pages 1468–1473.
- [381] Qian, X., Altché, F., Grégoire, J., et de La Fortelle, A. (2017). Autonomous intersection management systems: criteria, implementation and evaluation. IET Intelligent Transport Systems, 11(3):182–189.
- [382] Qian, X., Gregoire, J., Moutarde, F., et De La Fortelle, A. (2014). Priority-based coordination of autonomous and legacy vehicles at intersection. In 17th international IEEE conference on intelligent transportation systems (ITSC), pages 1166–1171. IEEE.
- [383] Qu, X., Yu, Y., Zhou, M., Lin, C.-T., et Wang, X. (2020). Jointly dampening traffic oscillations and improving energy consumption with electric, connected and automated vehicles: a reinforcement learning based approach. *Applied Energy*, 257:114030.

- [384] Quaglia, G., Cavallone, P., et Visconte, C. (2018). Agri_q: agriculture ugv for monitoring and drone landing. In *IFToMM Symposium on Mechanism Design for Robotics*, pages 413–423. Springer.
- [385] Quaglia, G., Visconte, C., Scimmi, L. S., Melchiorre, M., Cavallone, P., et Pastorelli, S. (2020). Design of a ugv powered by solar energy for precision agriculture. *Robotics*, 9(1):13.
- [386] Quilliot, A., Sarbinowski, A., et Toussaint, H. (2021). Vehicle driven approaches for non preemptive vehicle relocation with integrated quality criterion in a vehicle sharing system. *Annals of Operations Research*, 298(1):445–468.
- [387] Radford, N. A., Strawser, P., Hambuchen, K., Mehling, J. S., Verdeyen, W. K., Donnan, A. S., Holley, J., Sanchez, J., Nguyen, V., Bridgwater, L., et others (2015).
 Valkyrie: Nasa's first bipedal humanoid robot. *Journal of Field Robotics*, 32(3):397–419.
- [388] Rahman, S. M., et Ratrout, N. T. (2009). Review of the fuzzy logic based approach in traffic signal control: prospects in saudi arabia. *Journal of transportation Systems engineering and information Technology*, 9(5):58–70.
- [389] Rajamani, R., Choi, S. B., Law, B., Hedrick, J., Prohaska, R., et Kretz, P. (2000). Design and experimental implementation of longitudinal control for a platoon of automated vehicles. J. Dyn. Sys., Meas., Control, 122(3):470–476.
- [390] Raju, A., Anusha, M., Sunil, K., Swathi, K., et Ahmed, K. (2017). Patrolling system using unmanned ground vehicle for war field and border security. *Asian Journal of Applied Science and Technology (AJAST)*, 1(5):17–21.
- [391] Ramezani, M., Haddad, J., et Geroliminis, N. (2015). Dynamics of heterogeneity in urban networks: aggregated traffic modeling and hierarchical control. *Transportation Research Part B: Methodological*, 74:1–19.
- [392] Rasheed, A., Zia, H., Hashmi, F., Hadi, U., Naim, W., et Ajmal, S. (2013). Fleet & convoy management using vanet. *Journal of Computer Networks*, 1(1):1–9.
- [393] Ren, W., et Sorensen, N. (2008). Distributed coordination architecture for multirobot formation control. *Robotics and Autonomous Systems*, 56(4):324–333.
- [394] Riener, A., et Ferscha, A. (2013). Enhancing future mass ict with social capabilities. In *Co-evolution of intelligent socio-technical systems*, pages 141–184. Springer.
- [395] Rios-Torres, J., et Malikopoulos, A. A. (2016a). Automated and cooperative vehicle merging at highway on-ramps. *IEEE Transactions on Intelligent Transportation Systems*, 18(4):780–789.
- [396] Rios-Torres, J., et Malikopoulos, A. A. (2016b). A survey on the coordination of connected and automated vehicles at intersections and merging at highway onramps. *IEEE Transactions on Intelligent Transportation Systems*, 18(5):1066–1077.
- [397] Ripon, K. S. N., Solaas, J., et Dissen, H. (2017). Multi-objective evolutionary optimization for autonomous intersection management. In *Asia-Pacific Conference* on Simulated Evolution and Learning, pages 297–308. Springer.

- [398] Rizk, Y., Awad, M., et Tunstel, E. W. (2019). Cooperative heterogeneous multirobot systems: A survey. ACM Computing Surveys (CSUR), 52(2):1–31.
- [399] Robertson, D. I. (1969). 'tansyt'method for area traffic control. Traffic Engineering & Control, 8(8).
- [400] Rubenstein, M., Ahler, C., et Nagpal, R. (2012). Kilobot: A low cost scalable robot system for collective behaviors. In 2012 IEEE international conference on robotics and automation, pages 3293–3298.
- [401] Ryan, C., Murphy, F., et Mullins, M. (2020). Spatial risk modelling of behavioural hotspots: Risk-aware path planning for autonomous vehicles. *Transportation re*search part A: policy and practice, 134:152–163.
- [402] Saeedi, S., Trentini, M., Seto, M., et Li, H. (2016). Multiple-robot simultaneous localization and mapping: A review. *Journal of Field Robotics*, 33(1):3–46.
- [403] Scarinci, R., Hegyi, A., et Heydecker, B. (2017). Definition of a merging assistant strategy using intelligent vehicles. *Transportation research part C: emerging technologies*, 82:161–179.
- [404] Scarinci, R., et Heydecker, B. (2014). Control concepts for facilitating motorway on-ramp merging using intelligent vehicles. *Transport reviews*, 34(6):775–797.
- [405] Schepperle, H., et Böhm, K. (2008). Auction-based traffic management: Towards effective concurrent utilization of road intersections. In 2008 10th IEEE Conference on E-Commerce Technology and the Fifth IEEE Conference on Enterprise Computing, E-Commerce and E-Services, pages 105–112.
- [406] Schmidt, G. K., et Posch, B. (1983). A two-layer control scheme for merging of automated vehicles. In *The 22nd IEEE Conference on Decision and Control*, pages 495–500.
- [407] Schneider, F. E., et Wildermuth, D. (2011). Results of the european land robot trial and their usability for benchmarking outdoor robot systems. In *Conference* towards autonomous robotic systems, pages 408–409. Springer.
- [408] Schrank, D., Eisele, B., et Lomax, T. (2012). Tti's 2012 urban mobility report. Texas A&M Transportation Institute. The Texas A&M University System, 4.
- [409] Schwarting, W., Alonso-Mora, J., et Rus, D. (2018). Planning and decisionmaking for autonomous vehicles. Annual Review of Control, Robotics, and Autonomous Systems, 1:187–210.
- [410] Schwarting, W., Pierson, A., Alonso-Mora, J., Karaman, S., et Rus, D. (2019). Social behavior for autonomous vehicles. *Proceedings of the National Academy of Sciences*, 116(50):24972–24978.
- [411] Seeni, A., Schäfer, B., et Hirzinger, G. (2010). Robot mobility systems for planetary surface exploration-state-of-the-art and future outlook: a literature survey. *Aerospace Technologies Advancements*, 492:189–208.
- [412] Shabestary, S. M. A., et Abdulhai, B. (2018). Deep learning vs. discrete reinforcement learning for adaptive traffic signal control. In 2018 21st International Conference on Intelligent Transportation Systems (ITSC), pages 286–293. IEEE.

- [413] Shan, Z., Zhao, D., et Xia, Y. (2013). Urban road traffic speed estimation for missing probe vehicle data based on multiple linear regression model. In 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), pages 118–123.
- [414] Sharon, G., et Stone, P. (2017). A protocol for mixed autonomous and humanoperated vehicles at intersections. In *International Conference on Autonomous Agents and Multiagent Systems*, pages 151–167. Springer.
- [415] Shi, M.-k., Jiang, H., et Li, S.-h. (2016). An intelligent traffic-flow-based real-time vehicles scheduling algorithm at intersection. In 2016 14th International Conference on Control, Automation, Robotics and Vision (ICARCV), pages 1–5. IEEE.
- [416] Shirazi, M. S., et Morris, B. T. (2016). Looking at intersections: a survey of intersection monitoring, behavior and safety analysis of recent studies. *IEEE Transactions on Intelligent Transportation Systems*, 18(1):4–24.
- [417] Shladover, S. E. (2018). Connected and automated vehicle systems: Introduction and overview. *Journal of Intelligent Transportation Systems*, 22(3):190–200.
- [418] Shrivastava, S. (2019). V2v vehicle safety communication. In *Connected Vehicles*, pages 117–155. Springer.
- [419] Shufeng, W., Dawei, Z., Junhui, A., et Junyou, Z. (2017). Lane level turning trajectory tracking of intelligent vehicle based on drivers' manipulate habits. In 2017 29th Chinese Control And Decision Conference (CCDC), pages 6873–6877. IEEE.
- [420] Siciliano, B., Khatib, O., et Kröger, T. (2008). Springer handbook of robotics, volume 200. Springer.
- [421] Siegel, J. E., Erb, D. C., et Sarma, S. E. (2017). A survey of the connected vehicle landscape—architectures, enabling technologies, applications, and development areas. *IEEE Transactions on Intelligent Transportation Systems*, 19(8):2391–2406.
- [422] Simmons, R., Singh, S., Hershberger, D., Ramos, J., et Smith, T. (2001). First results in the coordination of heterogeneous robots for large-scale assembly. In *Experimental Robotics VII*, pages 323–332. Springer.
- [423] Sislak, D., Volf, P., Pechoucek, M., et Suri, N. (2010). Automated conflict resolution utilizing probability collectives optimizer. *IEEE Transactions on Systems, Man, and Cybernetics, Part C (Applications and Reviews)*, 41(3):365–375.
- [424] Sjoberg, K., Andres, P., Buburuzan, T., et Brakemeier, A. (2017). Cooperative intelligent transport systems in europe: Current deployment status and outlook. *IEEE Vehicular Technology Magazine*, 12(2):89–97.
- [425] SounderBruce (2022). Ramp meter at i-5 & ne 45th. available at https://www. flickr.com/.
- [426] Spiliopoulou, A., Kontorinaki, M., Papageorgiou, M., et Kopelias, P. (2014). Macroscopic traffic flow model validation at congested freeway off-ramp areas. *Transportation Research Part C: Emerging Technologies*, 41:18–29.

- [427] Spiliopoulou, A., Kontorinaki, M., Papamichail, I., et Papageorgiou, M. (2018). Realtime route diversion control at congested freeway off-ramp areas. *Transportation research part A: policy and practice*, 107:90–105.
- [428] Stager, A., Bhan, L., Malikopoulos, A., et Zhao, L. (2018). A scaled smart city for experimental validation of connected and automated vehicles. *IFAC-PapersOnLine*, 51(9):130–135.
- [429] Stern, R. E., Cui, S., Delle Monache, M. L., Bhadani, R., Bunting, M., Churchill, M., Hamilton, N., Pohlmann, H., Wu, F., Piccoli, B., et others (2018). Dissipation of stop-and-go waves via control of autonomous vehicles: Field experiments. *Transportation Research Part C: Emerging Technologies*, 89:205–221.
- [430] Stoilova, K., et Stoilov, T. (2020). Bi-level optimization application for urban traffic management. In 2020 15th Conference on Computer Science and Information Systems (FedCSIS), pages 327–336. IEEE.
- [431] Sun, Q., Diao, M., Zhang, Y., et Li, Y. (2017). Cooperative localization algorithm for multiple mobile robot system in indoor environment based on variance component estimation. *Symmetry*, 9(6):94.
- [432] Sundaram, S., Koutsopoulos, H. N., Ben-Akiva, M., Antoniou, C., et Balakrishna, R. (2011). Simulation-based dynamic traffic assignment for short-term planning applications. *Simulation Modelling Practice and Theory*, 19(1):450–462.
- [433] Sussman, J. M. (1993). Intelligent vehicle highway systems: Challenge for the future. In 1993 IEEE MTT-S International Microwave Symposium Digest, pages 101–104.
- [434] Sutton, R. S., et Barto, A. G. (2018). Reinforcement Learning: An Introduction. The MIT Press, second edition.
- [435] Suzuki, H., et Marumo, Y. (2018). A new approach to green light optimal speed advisory (glosa) systems for high-density traffic flowe. In 2018 21st International Conference on Intelligent Transportation Systems (ITSC), pages 362–367. IEEE.
- [436] Takahashi, M., Suzuki, T., Shitamoto, H., Moriguchi, T., et Yoshida, K. (2010). Developing a mobile robot for transport applications in the hospital domain. *Robotics and Autonomous Systems*, 58(7):889–899.
- [437] Tan, H.-S., Rajamani, R., et Zhang, W.-B. (1998). Demonstration of an automated highway platoon system. In Proceedings of the 1998 American control conference. ACC (IEEE Cat. No. 98CH36207), volume 3, pages 1823–1827. IEEE.
- [438] Tan, M. (1993). Multi-agent reinforcement learning: Independent vs. cooperative agents. In Proceedings of the tenth international conference on machine learning, pages 330–337.
- [439] Teodorovic, D., et Janic, M. (2016). Transportation Engineering: Theory, Practice and Modeling. Butterworth-Heinemann.
- [440] Teodorovic, D., Lucic, P., Popovic, J., Kikuchi, S., et Stanic, B. (2001). Intelligent isolated intersection. In 10th IEEE International Conference on Fuzzy Systems.(Cat. No. 01CH37297), volume 1, pages 276–279.

- [441] Tesfamariam, S., et Goda, K. (2013). Handbook of seismic risk analysis and management of civil infrastructure systems. Elsevier.
- [442] The New NY Bridge Magazine (2021). Fast tolls on the horizon. available at https://www.newnybridge.com/fast-tolls-on-the-horizon.
- [443] Thomanek, F., et Dickmanns, E. D. (1995). Autonomous road vehicle guidance in normal traffic. In Asian Conference on Computer Vision, pages 499–507. Springer.
- [444] Tian, D., <u>Zhu, K.</u>, Zhou, J., Wang, Y., et Liu, H. (2015). Swarm model for cooperative multi-vehicle mobility with inter-vehicle communications. *IET Intelligent Transport Systems*, 9(10):887–896.
- [445] Tlig, M., Buffet, O., et Simonin, O. (2014a). Decentralized traffic management: A synchronization-based intersection control. In 2014 International Conference on Advanced Logistics and Transport (ICALT), pages 109–114.
- [446] Tlig, M., Buffet, O., et Simonin, O. (2014b). Stop-free strategies for traffic networks: Decentralized on-line optimization. In ECAI 2014, pages 1191–1196. IOS Press.
- [447] Tomas-Gabarron, J.-B., Egea-Lopez, E., et Garcia-Haro, J. (2013). Vehicular trajectory optimization for cooperative collision avoidance at high speeds. *IEEE Transactions on Intelligent Transportation Systems*, 14(4):1930–1941.
- [448] Tong, Y., Zhao, L., Li, L., et Zhang, Y. (2015). Stochastic programming model for oversaturated intersection signal timing. *Transportation Research Part C: Emerging Technologies*, 58:474–486.
- [449] Traffic Technology International (2021). Toyota and carmera camera to build for using data. available hd maps avs https://www.traffictechnologytoday.com/news/autonomous-vehicles/ at toyota-and-carmera-to-build-hd-maps-for-avs-using-camera-data.html.
- [450] Tsugawa, S. (2014). Results and issues of an automated truck platoon within the energy its project. In 2014 IEEE Intelligent Vehicles Symposium Proceedings, pages 642–647.
- [451] Tsugawa, S., Kato, S., Tokuda, K., Matsui, T., et Fujii, H. (2001). A cooperative driving system with automated vehicles and inter-vehicle communications in demo 2000. In *ITSC 2001. 2001 IEEE Intelligent Transportation Systems. Proceedings (Cat. No. 01TH8585)*, pages 918–923.
- [452] Tuci, E., Alkilabi, M. H., et Akanyeti, O. (2018). Cooperative object transport in multi-robot systems: A review of the state-of-the-art. *Frontiers in Robotics and AI*, 5:59.
- [453] Turner, S. M., Eisele, W. L., Benz, R. J., et Holdener, D. J. (1998). Travel time data collection handbook. Technical Report, United States. Federal Highway Administration.
- [454] Uhlemann, E. (2015a). Active safety vehicles evolving toward automated driving [connected vehicles]. *IEEE Vehicular Technology Magazine*, 10(4):20–23.

- [455] Uhlemann, E. (2015b). Introducing connected vehicles [connected vehicles]. *IEEE vehicular technology magazine*, 10(1):23–31.
- [456] Ulmer, B. (1994). Vita ii-active collision avoidance in real traffic. In Proceedings of the Intelligent Vehicles' 94 Symposium, pages 1–6. IEEE.
- [457] University, T. (2019). Intelligent vehicle test base at changshu, china. available at https://www.au.tsinghua.edu.cn/info/1062/1077.htm.
- [458] Uno, A., Sakaguchi, T., et Tsugawa, S. (1999). A merging control algorithm based on inter-vehicle communication. In Proceedings 199 IEEE/IEEJ/JSAI International Conference on Intelligent Transportation Systems (Cat. No. 99TH8383), pages 783–787.
- [459] U.S. Department of Transportation (2021a). Multi-modal intelligent traffic safety system (mmitss). available at https://www.its.dot.gov/research_archives/dma/bundle/ mmitss_plan.htm.
- [460] U.S. Department of Transportation (2021b). What are the benefits of connected vehicles? available at https://www.its.dot.gov/cv_basics/cv_basics_benefits.htm.
- [461] U.S. Department of Transportation (2022). Ngsim: Next generation simulation. available at https://ops.fhwa.dot.gov/trafficanalysistools/ngsim.htm.
- [462] Van der Pol, E., et Oliehoek, F. A. (2016). Coordinated deep reinforcement learners for traffic light control. Proceedings of Learning, Inference and Control of Multi-Agent Systems (at NIPS 2016).
- [463] Van Lint, J., et Van Hinsbergen, C. (2012). Short-term traffic and travel time prediction models. Artificial Intelligence Applications to Critical Transportation Issues, 22(1):22–41.
- [464] VanMiddlesworth, M., Dresner, K., et Stone, P. (2008). Replacing the stop sign: Unmanaged intersection control for autonomous vehicles. In Proceedings of the 7th international joint conference on Autonomous agents and multiagent systems-Volume 3, pages 1413–1416.
- [465] Varaiya, P. (2000). A question about hierarchical systems. In System Theory, pages 313–324. Springer.
- [466] Vasirani, M., et Ossowski, S. (2009). Evaluating policies for reservation-based intersection control. In Proceedings of the 14th Portuguese conference on artificial intelligence, volume 14, pages 39–50.
- [467] Vasirani, M., et Ossowski, S. (2012). A market-inspired approach for intersection management in urban road traffic networks. *Journal of Artificial Intelligence Research*, 43:621–659.
- [468] Velasco-Hernandez, G., Caicedo-Bravo, E., Barry, J., et Walsh, J. (2020). Intersection management systems and internet of things: A review. In 2020 IEEE 16th International Conference on Intelligent Computer Communication and Processing (ICCP), pages 439–445.
- [469] Verma, J. K., et Ranga, V. (2021). Multi-robot coordination analysis, taxonomy, challenges and future scope. *Journal of Intelligent & Robotic Systems*, 102(1):1–36.

- [470] Vilca, J., Adouane, L., et Mezouar, Y. (2015). A novel safe and flexible control strategy based on target reaching for the navigation of urban vehicles. *Robotics and Autonomous Systems*, 70:215–226.
- [471] Vilca, J., Adouane, L., et Mezouar, Y. (2016). Adaptive leader-follower formation in cluttered environment using dynamic target reconfiguration. In *Distributed Autonomous Robotic Systems*, pages 237–254. Springer.
- [472] Vilca, J., Adouane, L., et Mezouar, Y. (2018). Stable and flexible multi-vehicle navigation based on dynamic inter-target distance matrix. *IEEE Transactions on Intelligent Transportation Systems*, 20(4):1416–1431.
- [473] Vilca, J.-M., Adouane, L., et Mezouar, Y. (2013). Reactive navigation of mobile robot using elliptic trajectories and effective on-line obstacle detection. *In Gy- roscopy and Navigation Journal*, 4(1):14 25.
- [474] Vilca Ventura, J. M. (2015). Safe and flexible hybrid control architecture for the navigation in formation of a group of vehicles. PhD thesis, Clermont-Ferrand 2.
- [475] Virdi, N., Grzybowska, H., Waller, S. T., et Dixit, V. (2019). A safety assessment of mixed fleets with connected and autonomous vehicles using the surrogate safety assessment module. *Accident Analysis & Prevention*, 131:95–111.
- [476] Vis, I. F. (2006). Survey of research in the design and control of automated guided vehicle systems. *European Journal of Operational Research*, 170(3):677–709.
- [477] Vivaldini, K. C. T., Tamashiro, G., Martins Junior, J., et Becker, M. (2013). Communication infrastructure in the centralized management system for intelligent warehouses. In International Workshop on Robotics in Smart Manufacturing, pages 127–136. Springer.
- [478] Walter, W. G. (1953). The living brain. WW Norton.
- [479] Wang, M., Daamen, W., Hoogendoorn, S. P., et van Arem, B. (2016). Connected variable speed limits control and car-following control with vehicle-infrastructure communication to resolve stop-and-go waves. *Journal of Intelligent Transportation Systems*, 20(6):559–572.
- [480] Wang, Y., Cai, P., et Lu, G. (2020). Cooperative autonomous traffic organization method for connected automated vehicles in multi-intersection road networks. *Transportation research part C: emerging technologies*, 111:458–476.
- [481] Wang, Z., Bian, Y., Shladover, S. E., Wu, G., Li, S. E., et Barth, M. J. (2019a). A survey on cooperative longitudinal motion control of multiple connected and automated vehicles. *IEEE Intelligent Transportation Systems Magazine*, 12(1):4–24.
- [482] Wang, Z., Wu, G., et Barth, M. (2018). Distributed consensus-based cooperative highway on-ramp merging using v2x communications. Technical Report, SAE Technical Paper.
- [483] Wang, Z., Wu, G., et Barth, M. J. (2019b). Cooperative eco-driving at signalized intersections in a partially connected and automated vehicle environment. *IEEE Transactions on Intelligent Transportation Systems*, 21(5):2029–2038.

- [484] Ward, J., Agamennoni, G., Worrall, S., et Nebot, E. (2014). Vehicle collision probability calculation for general traffic scenarios under uncertainty. In 2014 IEEE Intelligent Vehicles Symposium Proceedings, pages 986–992.
- [485] Wei, L., Li, Z., Gong, J., Gong, C., et Li, J. (2021). Autonomous driving strategies at intersections: Scenarios, state-of-the-art, and future outlooks. In 2021 IEEE International Intelligent Transportation Systems Conference (ITSC), pages 44–51.
- [486] Wei, P., Cao, Y., et Sun, D. (2013). Total unimodularity and decomposition method for large-scale air traffic cell transmission model. *Transportation Research Part B: Methodological*, 53:1–16.
- [487] Werfel, J., Petersen, K., et Nagpal, R. (2014). Designing collective behavior in a termite-inspired robot construction team. *Science*, 343(6172):754–758.
- [488] Wikipedia (2022). Traffic light. available at https://en.wikipedia.org/wiki/Traffic_ light.
- [489] Williams, M. (1988). Prometheus-the european research programme for optimising the road transport system in europe. In IEE Colloquium on Driver Information, pages 1–1. IET.
- [490] Winner, H., Hakuli, S., Lotz, F., et Singer, C. (2014). Handbook of driver assistance systems. Springer International Publishing Amsterdam, The Netherlands:.
- [491] Wiquist, W. (2020). Fcc modernizes 5.9 ghz band for wi-fi and auto safety. FCC News.
- [492] Wirasingha, S. G., Schofield, N., et Emadi, A. (2008). Plug-in hybrid electric vehicle developments in the us: Trends, barriers, and economic feasibility. In 2008 IEEE vehicle power and propulsion conference, pages 1–8.
- [493] Wojtusiak, J., Warden, T., et Herzog, O. (2012). Machine learning in agent-based stochastic simulation: Inferential theory and evaluation in transportation logistics. Computers & Mathematics with Applications, 64(12):3658–3665.
- [494] Wolpert, D. H. (2006). Information theory—the bridge connecting bounded rational game theory and statistical physics. In *Complex Engineered Systems*, pages 262–290. Springer.
- [495] Wu, J., Abbas-Turki, A., et El Moudni, A. (2012). Cooperative driving: an ant colony system for autonomous intersection management. *Applied Intelligence*, 37(2):207–222.
- [496] Wu, J., Abbas-Turki, A., et Moudni, A. E. (2009). Intersection traffic control by a novel scheduling model. In 2009 IEEE/INFORMS International Conference on Service Operations, Logistics and Informatics, pages 329–334. IEEE.
- [497] Wu, W., Zhang, J., Luo, A., et Cao, J. (2014). Distributed mutual exclusion algorithms for intersection traffic control. IEEE Transactions on Parallel and Distributed Systems, 26(1):65–74.
- [498] Wuthishuwong, C., et Traechtler, A. (2013). Coordination of multiple autonomous intersections by using local neighborhood information. In 2013 international conference on connected vehicles and expo (ICCVE), pages 48–53. IEEE.

- [499] Wuthishuwong, C., et Traechtler, A. (2017). Consensus-based local information coordination for the networked control of the autonomous intersection management. Complex & Intelligent Systems, 3(1):17–32.
- [500] Wymann, B., Dimitrakakis, C., Sumner, A., Espié, E., et Guionneau, C. (2015). Torcs: The open racing car simulator. available at http://www.cse.chalmers.se/ ~chrdimi/papers/torcs.pdf.
- [501] Xie, X.-F., Barlow, G. J., Smith, S. F., et Rubinstein, Z. B. (2011). Platoon-based self-scheduling for real-time traffic signal control. In 2011 14th International IEEE Conference on Intelligent Transportation Systems (ITSC), pages 879–884.
- [502] Xie, Y., Zhang, H., Gartner, N. H., et Arsava, T. (2017). Collaborative merging strategy for freeway ramp operations in a connected and autonomous vehicles environment. *Journal of Intelligent Transportation Systems*, 21(2):136–147.
- [503] Xin, J., Wang, C., Zhang, Z., et Zheng, N. (2014). China future challenge: Beyond the intelligent vehicle. *IEEE Intell. Transp. Syst. Soc. Newslett*, 16(2):8–10.
- [504] Xu, B., Ban, X. J., Bian, Y., Wang, J., et Li, K. (2017). V2i based cooperation between traffic signal and approaching automated vehicles. In 2017 IEEE Intelligent Vehicles Symposium (IV), pages 1658–1664.
- [505] Xu, B., Li, S. E., Bian, Y., Li, S., Ban, X. J., Wang, J., et Li, K. (2018). Distributed conflict-free cooperation for multiple connected vehicles at unsignalized intersections. *Transportation Research Part C: Emerging Technologies*, 93:322–334.
- [506] Xu, H., Zhang, Y., Li, L., et Li, W. (2019). Cooperative driving at unsignalized intersections using tree search. *IEEE Transactions on Intelligent Transportation Systems*, 21(11):4563–4571.
- [507] Xu, X.-y., Liu, J., Li, H.-y., et Hu, J.-Q. (2014). Analysis of subway station capacity with the use of queueing theory. *Transportation research part C: emerging technologies*, 38:28–43.
- [508] Xu, Y., Zhou, H., Ma, T., Zhao, J., Qian, B., et Shen, S. (2021). Leveraging multi-agent learning for automated vehicles scheduling at non-signalized intersections. *IEEE Internet of Things Journal*, pages 1–1.
- [509] Yan, F., Dridi, M., et El Moudni, A. (2009). Autonomous vehicle sequencing algorithm at isolated intersections. In 2009 12th International IEEE conference on intelligent transportation systems, pages 1–6.
- [510] Yan, F., Dridi, M., et Moudni, A. (2011). A scheduling approach for autonomous vehicle sequencing problem at multi-intersections. *International Journal of Operations Research*, 9(1).
- [511] Yang, B., et Wu, R. (2016a). A modified probability collectives optimization algorithm based on trust region method and a new temperature annealing schedule. *Soft Computing*, 20(4):1581–1600.
- [512] Yang, D., Jiang, K., Zhao, D., Yu, C., Cao, Z., Xie, S., Xiao, Z., Jiao, X., Wang, S., et Zhang, K. (2018a). Intelligent and connected vehicles: Current status and future perspectives. Science China Technological Sciences, 61(10):1446–1471.

- [513] Yang, H., Rakha, H., et Ala, M. V. (2016b). Eco-cooperative adaptive cruise control at signalized intersections considering queue effects. *IEEE Transactions* on Intelligent Transportation Systems, 18(6):1575–1585.
- [514] Yang, J., Zhang, J., et Wang, H. (2020). Urban traffic control in software defined internet of things via a multi-agent deep reinforcement learning approach. IEEE Transactions on Intelligent Transportation Systems, 22(6):3742–3754.
- [515] Yang, K., Guler, S. I., et Menendez, M. (2016c). Isolated intersection control for various levels of vehicle technology: Conventional, connected, and automated vehicles. *Transportation Research Part C: Emerging Technologies*, 72:109–129.
- [516] Yang, K., Zheng, N., et Menendez, M. (2017). Multi-scale perimeter control approach in a connected-vehicle environment. *Transportation research procedia*, 23:101–120.
- [517] Yang, Z., Huang, H., Wang, G., Pei, X., et Yao, D.-y. (2018b). Cooperative driving model for non-signalized intersections with cooperative games. *Journal of Central South University*, 25(9):2164–2181.
- [518] Yao, B., Chen, C., Cao, Q., Jin, L., Zhang, M., <u>Zhu, H.</u>, et Yu, B. (2017). Short-term traffic speed prediction for an urban corridor. *Computer-Aided Civil and Infrastructure Engineering*, 32(2):154–169.
- [519] Yao, H., Cui, J., Li, X., Wang, Y., et An, S. (2018). A trajectory smoothing method at signalized intersection based on individualized variable speed limits with location optimization. *Transportation Research Part D: Transport and Environment*, 62:456–473.
- [520] Yildirimoglu, M., et Geroliminis, N. (2013). Experienced travel time prediction for congested freeways. *Transportation Research Part B: Methodological*, 53:45–63.
- [521] Yin, Y. (2008). Robust optimal traffic signal timing. *Transportation Research Part B: Methodological*, 42(10):911–924.
- [522] Yu, C., Feng, Y., Liu, H. X., Ma, W., et Yang, X. (2019). Corridor level cooperative trajectory optimization with connected and automated vehicles. *Transportation Research Part C: Emerging Technologies*, 105:405–421.
- [523] Yu, F., Xian, W., Chen, Y., Liu, F., Liao, M., Madhavan, V., et Darrell, T. (2018).
 Bdd100k: A diverse driving video database with scalable annotation tooling. arXiv preprint arXiv:1805.04687, 2(5):6.
- [524] Yuan, J., Abdel-Aty, M. A., Yue, L., et Cai, Q. (2020). Modeling real-time cyclelevel crash risk at signalized intersections based on high-resolution event-based data. IEEE Transactions on Intelligent Transportation Systems, pages 1–16.
- [525] Yun, I., et Park, B. (2012). Stochastic optimization for coordinated actuated traffic signal systems. *Journal of Transportation Engineering*, 138(7):819–829.
- [526] Yurtsever, E., Lambert, J., Carballo, A., et Takeda, K. (2020). A survey of autonomous driving: Common practices and emerging technologies. *IEEE access*, 8:58443–58469.

- [527] Yuta, S., et Premvuti, S. (1992). Coordinating autonomous and centralized decision making to achieve cooperative behaviors between multiple mobile robots. In Proceedings of the IEEE/RSJ international conference on intelligent robots and systems, volume 3, pages 1566–1574.
- [528] Zabbar, M. A. B., et Ahmed, C. N. (2016). Design & implementation of an unmanned ground vehicle (ugv) surveillance robot. International Journal of Electrical and Electronics Engineering (IJEEE) 5.6 (2016): 2278, 9944.
- [529] Zeadally, S., Javed, M. A., et Hamida, E. B. (2020). Vehicular communications for its: standardization and challenges. *IEEE Communications Standards Magazine*, 4(1):11–17.
- [530] Zhang, G., et Wang, Y. (2010). Optimizing minimum and maximum green time settings for traffic actuated control at isolated intersections. *IEEE Transactions* on Intelligent Transportation Systems, 12(1):164–173.
- [531] Zhang, J., Pei, H., Ban, X. J., et Li, L. (2022). Analysis of cooperative driving strategies at road network level with macroscopic fundamental diagram. *Transportation Research Part C: Emerging Technologies*, 135:103503.
- [532] Zhang, K., De La Fortelle, A., Zhang, D., et Wu, X. (2013). Analysis and modeled design of one state-driven autonomous passing-through algorithm for driverless vehicles at intersections. In 2013 IEEE 16th International Conference on Computational Science and Engineering, pages 751–757.
- [533] Zhang, M., Batta, R., et Nagi, R. (2009). Modeling of workflow congestion and optimization of flow routing in a manufacturing/warehouse facility. *Management Science*, 55(2):267–280.
- [534] Zhang, Y., Malikopoulos, A. A., et Cassandras, C. G. (2017). Decentralized optimal control for connected automated vehicles at intersections including left and right turns. In 2017 IEEE 56th Annual Conference on Decision and Control (CDC), pages 4428–4433.
- [535] Zhang, Y. J., Malikopoulos, A. A., et Cassandras, C. G. (2016). Optimal control and coordination of connected and automated vehicles at urban traffic intersections. In 2016 American Control Conference (ACC), pages 6227–6232. IEEE.
- [536] Zhao, C., Li, L., Pei, X., Li, Z., Wang, F.-Y., et Wu, X. (2021). A comparative study of state-of-the-art driving strategies for autonomous vehicles. *Accident Analysis & Prevention*, 150:105937.
- [537] Zhao, L., Malikopoulos, A., et Rios-Torres, J. (2018a). Optimal control of connected and automated vehicles at roundabouts: An investigation in a mixedtraffic environment. *IFAC-PapersOnLine*, 51(9):73–78.
- [538] Zhao, L., Song, Y., Zhang, C., Liu, Y., Wang, P., Lin, T., Deng, M., et Li, H. (2019a).
 T-gcn: A temporal graph convolutional network for traffic prediction. *IEEE Transactions on Intelligent Transportation Systems*, 21(9):3848–3858.
- [539] Zhao, L., et Sun, J. (2013). Simulation framework for vehicle platooning and car-following behaviors under connected-vehicle environment. *Procedia-Social and Behavioral Sciences*, 96:914–924.

- [540] Zhao, W., Ngoduy, D., Shepherd, S., Liu, R., et Papageorgiou, M. (2018b). A platoon based cooperative eco-driving model for mixed automated and humandriven vehicles at a signalised intersection. *Transportation Research Part C: Emerging Technologies*, 95:802–821.
- [541] Zhao, Z., Wang, Z., Wu, G., Ye, F., et Barth, M. J. (2019b). The state-of-the-art of coordinated ramp control with mixed traffic conditions. In 2019 IEEE Intelligent Transportation Systems Conference (ITSC), pages 1741–1748.
- [542] Zheng, J., Sun, W., Huang, S., Shen, S., Yu, C., <u>Zhu, J.</u>, Liu, B., et Liu, H. X. (2018). Traffic signal optimization using crowdsourced vehicle trajectory data. Technical Report, TRB committee AHB25 Standing Committee on Traffic Signal Systems.
- [543] Zheng, Y., Li, S. E., Wang, J., Cao, D., et Li, K. (2015). Stability and scalability of homogeneous vehicular platoon: Study on the influence of information flow topologies. *IEEE Transactions on intelligent transportation systems*, 17(1):14–26.
- [544] Zheng, Y., Ran, B., Qu, X., Zhang, J., et Lin, Y. (2019). Cooperative lane changing strategies to improve traffic operation and safety nearby freeway off-ramps in a connected and automated vehicles environment. *IEEE Transactions on Intelligent Transportation Systems*, 21(11):4605–4614.
- [545] Zhou, M., Qu, X., et Li, X. (2017). A recurrent neural network based microscopic car following model to predict traffic oscillation. Transportation research part C: emerging technologies, 84:245–264.
- [546] Zhou, M., Yu, Y., et Qu, X. (2019). Development of an efficient driving strategy for connected and automated vehicles at signalized intersections: A reinforcement learning approach. IEEE Transactions on Intelligent Transportation Systems, 21(1):433–443.
- [547] Zhu, F., Aziz, H. A., Qian, X., et Ukkusuri, S. V. (2015a). A junction-tree based learning algorithm to optimize network wide traffic control: A coordinated multiagent framework. *Transportation Research Part C: Emerging Technologies*, 58:487– 501.
- [548] Zhu, F., et Ukkusuri, S. V. (2015b). A linear programming formulation for autonomous intersection control within a dynamic traffic assignment and connected vehicle environment. *Transportation Research Part C: Emerging Technologies*, 55:363–378.
- [549] Zhu, H., Yuen, K.-V., Mihaylova, L., et Leung, H. (2017). Overview of environment perception for intelligent vehicles. IEEE Transactions on Intelligent Transportation Systems, 18(10):2584–2601.
- [550] <u>Zhu, Z.</u>, Adouane, L., et Quilliot, A. (2022). Decentralized cavs based on mimafc strategy for multi-intersection traffic network. Under review by IET Intelligent Transport Systems.
- [551] Zohdy, I. H., Kamalanathsharma, R. K., et Rakha, H. (2012a). Intersection management for autonomous vehicles using icacc. In 2012 15th international IEEE conference on intelligent transportation systems, pages 1109–1114.

[552] Zohdy, I. H., et Rakha, H. (2012b). Game theory algorithm for intersectionbased cooperative adaptive cruise control (cacc) systems. In *2012 15th International IEEE Conference on Intelligent Transportation Systems*, pages 1097–1102.

Abstract:

Cooperative navigation (CN) is a frequently used technique for ensuring the effective navigation of Intelligent Vehicles (IVs). In this background, efforts to establish a Connected Autonomous Vehicles (CAVs)-based traffic control system via vehicular communications technology have accelerated in recent decades. Safe and flexible multi-vehicle coordination (MVC) technology, in particular, has attracted considerable interest for its ability to deal with complicated environments/situations. Additionally, a feasible hierarchical architecture is critical for cooperative driving with numerous control goals in autonomous vehicles. Thus, the objective of this PhD thesis is to develop reliable MVC technology (e.g., trajectory planning and decision-making for motion planning) and CAVs-based frameworks for use in complex environments/situations. To achieve this goal, this thesis first presented a safe and flexible cooperative navigation technique with risk assessment for cramped local locations (defined as single intersection/roundabout). The *e*-constraint Probability Collectives (PC) algorithm, which is based on the distributed Collective Intelligence (CI) theory, is developed to offer proper solutions for cooperative driving. More precisely, IVs can compute their optimal/sub-optimal and risk-sensitive (i.e., invasive or conservative) cooperative navigation strategies base on the decentralized &-PC framework, enabling collision-free trajectories in the decision-making level. Next, it is suggested a global supervisor responsible for scheduling and improving vehicle navigation routes while also proposing well-suited trade-offs between speed and risk to achieve the targeted tasks. To better deal with the inhere complexity of CN system in a transportation network (e.g., intersection/roundabout and the expended intersection network), the second part of the thesis addresses the potentialities of adopting Multi-layer Hybrid Control Policy and Motion Planning (MHCP-MP) framework. Given the fluctuating road traffic, it was recommended that local supervisors be in control of the urban network's intersections (tricky regions). Specifically, the local supervisor works as a mediator between the global traffic management level and the CAVs decision level, sending instructions to regulate vehicles' trajectories and improve the mobility and safety of the overall transportation system. To accomplish the aim, a Macroscopic Fundamental Diagram (MFD)-based approach in the proposed MHCP-MP framework is designed with concise urban traffic data (e.g., vehicle position, speed, etc.). Further, the Micro-Macro Flow Control (MiMaFC) strategy is proposed to demonstrate the advantages of establishing a link between the suggested local collective optimization framework and macro traffic model for improving the fluidity of the overall transportation system. Following that, the suggested intersection navigation protocols in a deep relationship with our established intelligent intersection management system are designed to permit an uncertain traffic flow. Finally, the proposed CN management architecture in this thesis has been proven in a dedicated transportation network through intensive simulation.

Keywords: Cooperative navigation, Multi-vehicle coordination, Hierarchical architecture, Risk assessment, Probability collectives, Traffic management.

Résumé :

La Navigation Coopérative (NC) est une technique fréquemment utilisée pour assurer la navigation efficace des Véhicules Intelligents (VI). Dans ce contexte, les efforts visant à établir un système de contrôle du trafic basé sur les véhicules autonomes connectés (CAVs) par le biais des technologies de communication entre véhicules se sont accélérés au cours des dernières décennies. La technologie de coordination multi-véhicules (MVC) sûre et flexible a suscité, en particulier, un intérêt considérable grâce à sa capacité à gérer des environnements/situations complexes. En outre, une architecture hiérarchique faisable est essentielle pour la conduite coopérative avec de nombreux objectifs de contrôle des véhicules autonomes. Ainsi, l'objectif de cette thèse est de développer une technologie MVC fiable (e.g., la prise de décision pour la planification) et des cadres basés sur les CAVs pour une utilisation dans des environnements/situations complexes. Pour atteindre cet objectif, cette thèse présente, tout d'abord, une technique de NC sûre et flexible avec une évaluation des risques pour les emplacements locaux encombrés (définis comme une seule intersection et/ou un rond-point). L'algorithme ɛ-Probabilté Collective (PC) à contrainte, qui est basé sur la théorie de l'Intelligence Collective (IC) distribuée, est développé pour offrir des solutions appropriées pour la conduite coopérative. Plus précisément, les VI peuvent calculer leurs stratégies de navigation coopérative optimales/sous-optimales et sensibles au risque (invasives ou conservatrices) en se basant sur le cadre décentralisé ɛ-PC, ce qui garantit des trajectoires sans collision. Ensuite, nous suggérons d'utiliser un superviseur global responsable de l'ordonnancement et de l'amélioration des trajectoires de navigation des véhicules, tout en proposant des compromis adaptés entre la vitesse et le risque de la réalisation des tâches visées. Afin de mieux gérer la complexité inhérente aux systèmes NC dans un réseau de transport (e.g., les intersections/ronds-points et le réseau étendu d'intersections), la deuxième partie de la thèse aborde le potentiel de l'adoption d'une architecture de contrôle hybride multicouches et de planification du mouvement (CHM-PM). Compte tenu de la fluctuation du trafic routier, il a été recommandé que des superviseurs locaux contrôlent les intersections du réseau urbain (régions dangereuses). Plus précisément, le superviseur local joue le rôle intermédiaire entre le niveau de gestion global du trafic et le niveau de décision des CAVs, en envoyant des instructions pour réguler les trajectoires des véhicules et améliorer la mobilité et la sécurité du système de transport globale. Pour atteindre cet objectif, une approche basée sur le Diagramme Fondamental Macroscopique (DFM) dans l'architecture CHM-PM proposée est conçue avec des données de trafic urbain concises (par exemple, la position du véhicule, la vitesse, etc.). En outre, la stratégie de contrôle des flux micro-macro (MiMaFC) est proposée pour démontrer les avantages de l'établissement d'un lien entre l'architecture d'optimisation collective locale proposée et le macro modèle de trafic pour améliorer la fluidité du système de transport global. Ensuite, les protocoles suggérés de navigation aux intersections, en forte relation avec notre système de gestion intelligente des intersections, sont conçus pour permettre un flux de trafic incertain. Enfin, l'architecture de gestion des NC proposée dans cette thèse a été évaluée dans un réseau de transport par un travail de simulation intensive.

Mots-clés : Navigation coopérative, Coordination multi-véhicules, Architecture hiérarchique, Évaluation des risques, Probabilté Collective, Gestion du trafic.